

Carr, Katharine Elizabeth (2018) *Probing tissue surfaces*. D.Sc. in Medicine awarded by published work.

https://theses.gla.ac.uk/38922/

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Probing tissue surfaces

Katharine Elizabeth Carr

BSc PhD FRSB FAS FHEA

Submitted in fulfilment of the requirements for the degree of DSc in Medicine

School of Medicine, Dentistry and Nursing College of Medicine, Veterinary and Life Sciences University of Glasgow

Volume 1 of 3

© K.E Carr, October, 2018

Abstract

This thesis on 'Probing tissue surfaces' describes work done on *in vivo* and *in vitro* models between the 1960s and thesis submission in 2018. Output is by book excerpts, chapters and journal articles. Chapters 1 to 4 are linked through cognate output: Chapters 2, 3 and 4 each contain a review article summarising much of the content. Collaborative links are highlighted, as is the author's changing contribution from entirely 'hands-on' laboratory work to a principal investigator or supervisory role.

Chapter 1 provides a background to the other Chapters, highlighting the use of transmission electron microscopy and a continuing interest in the interpretation of data from sectioned material. Applications include the use of microscopy for quality control and for reporting tissue responses to environmental challenge. Chapter 2 sets out contributions to the development of image interpretation in the then new field of scanning electron microscopy, particularly of soft tissues, with relevance to research and wider applications, including the effects of environmental challenge. Key findings include: early images and reports of intestinal villi, skin or isolated cells; correlative techniques to interpret surface information; optimisation of techniques and descriptions of the surface responses to developmental change or ulcerogenic and other agents.

Chapter 3 deals with the impact of external irradiation, through the exploration by microscopy of its effects on tissue surfaces and deeper structures, mainly alimentary, with relevance to clinical side effects of radiotherapy and to space flight. Key findings include radiation-induced villous collapse differing in extent from changes in proliferative compartments; and variations in responses of structures from all four basic tissue types to different radiation schedules. Chapter 4 addresses questions on microparticle uptake, with relevance mainly to the environmental impact of the Chernobyl incident. Key findings include: most early particle uptake *in vivo in situ* occurring not at Peyer's patches but through villous epithelium, possibly at tight junctions; ethanol and cooling *in vitro* producing different changes to uptake, tight junctions and junctional proteins; and uptake being higher in late pregnancy and early lactation, affected by age not species and increased by irradiation.

Chapter 5 brings together the content of Chapters 1 to 4, summarising their contributions and relevance, both at publication and thereafter. The thesis title reflects the use of probing at different levels, not only during image formation but also by an environmental challenge such as radiation, a pharmacological agent or atmospheric pollution.

Table of Contents

Probing tissue surfaces	1
Abstract	2
List of accompanying material	6
Preface	17
Acknowledgements	19
Author's declaration	21
Abbreviations	22
Chapter 1 Ultrastructure	24
Preface - extract from Presubmission Abstract (Chapter 1)	24
1.1 Fore-runner to surface studies and links to Chapters 2 and 3	24
1.2 Ongoing studies of biological sections	26
1.3 Context and comment	27
Chapter 2 Scanning electron microscopy emerging as a biological probe	29
Preface - extract from Presubmission Abstract (Chapter 2)	29
2.1 Early scanning electron microscopy and links to Chapter 1	29
2.2. Early reviews of biological scanning electron microscopy	
2.3 Correlation of SEM images with those from established techniques	
2.4 Development of scanning electron microscopy techniques	35
2.5 Scanning around the alimentary system	
2.6 Later reviews of biological scanning electron microscopy	41
2.7 Context and comment	42
Chapter 3 Correlative morphology of irradiated small intestinal surfaces	45
Preface - extract from Presubmission Abstract (Chapter 3)	45
3.1 Villous radiation-induced damage and links to Chapter 2	46
3.2 Early reviews on 'topographical' and ultrastructural radiation effects	49
3.3 Radiation effects at different sites	50

3.4 Qu	alitative radiation effects on small intestinal surfaces	.52
3.5 Sco	oring of radiation-induced changes to intestinal surfaces	.53
3.6 Imj	portance of underlying structures to surface responses	.55
3.7 Me	chanisms of structural damage	. 59
3.8 Su	mmary of small intestinal responses	.61
3.9 Dif	ferent cell populations	.66
3.10 R	eviews of irradiated small intestine	.67
3.10	.1 Structure and function	.68
3.10	.2 Comprehensive review	.68
3.11 C	ontext and comment	.70
Chapter 4	4 Microparticles crossing intestinal surfaces	.73
Prefac	e - extract from Presubmission Abstract (Chapter 4)	.73
4.1 Qu	estions on uptake and links to Chapter 3	.74
4.2 Wh	nat model? Model 1, the standard in vivo in situ model	.76
4.3 Mc	odel 1, variation in particle type	.76
4.4 Mc	odel 1, developments in methodology	.77
4.5 Wh	nat model? Model 2, the standard in vitro model	.78
4.6 Mc	odel 2, developments in methodology	.79
4.7 Wh	nat model? Invited Review on microparticle uptake	.81
4.8 Q.	1 Do microparticles enter tissues by an intestinal route ?	. 82
4.8.1	In vivo in situ data (Model 1)	. 82
4.8.2	2 Uptake time-frame (Model 1)	. 82
4.8.3	B Uptake sites, proximo-distal and villous/Peyer's patches (Model 1)	.83
4.8.4	Uptake pathway, paracellular/apical (Model 1)	. 85
4.8.5	5 Onward movement (Model 1)	.86
4.8.6	5 Particle-associated factors (Model 1)	.86
4.8.7	Particle-free in vitro monolayers (Model 2)	. 87

4.8.8 Monolayer uptake routes (Model 2)	
4.8.9 Particle-associated factors (Model 2)	90
4.9 Q. 2 Are any groups 'at risk', such as pregnant or lactating females?	90
4.9.1 Pregnancy and lactation (Model 1)	90
4.9.2 Species, age and pathophysiology (Model 1)	91
4.9.3 Irradiation and related exposures (Models 1 and 2)	92
4.10 Overview and conclusions	94
4.11 Context and comment	95
Chapter 5 Conclusions – key findings and significance	97
Preface - extract from Presubmission Abstract (Chapter 5)	97
5.1 Background to output	97
5.2 Influence of publications in Chapters 1 to 4	98
5.2.1 Representative papers	98
5.2.2 Highly cited papers	100
5.2.3 Citations of interest	102
5.2.4 Summary of output analysis	
5.3 Context and comment - probes and surfaces	
Complete list of References in Harvard format	105

List of accompanying material

This list sets out the submissions in the order of their appearance in the thesis, not necessarily in exact chronological order, but instead chosen to develop the theme of the Chapter. Each submission is therefore identified by three numbers, according to Chapter, then Section, then order within its Section. An individual entry below begins with its number within the Section, then a note of how it appears in the text and then a record of its full reference in the Harvard format used in the thesis. The number in bold (in brackets) at the end of each reference signifies the position of the article in the Presubmission list, which is in chronological order.

The full references, in Harvard style, are also annotated to indicate the status of some of the authors, as follows, **only for an author in first or last position** –

* Student of thesis author, whether as sole or senior supervisor for either graduate or undergraduate student. In one case, the student continued in the group as a post-doctoral assistant: the asterisk sign is used here for all entries for her.

(*) Student co-supervised by the thesis author or supervised by another senior member of the collaborative group, but with the thesis author involved in guiding a microscopy aspect of the work. While Department Head or School Director, I formally appointed an Advisor for each research student: Advisors often became interested in the project and became co-authors, but this is not marked below.

† Technician in group.

Chapter 1 Ultrastructure

Chapter 1 - Section 1 Fore-runner to surface studies and links to Chapters 2 and 3

- 1. Carr et al, 1967- Carr, K.E., Arbuthnott, J.P., Toner, P.G. & Gemmell, C.G. 1967, "The fine structure of rat liver cells in suspension", *Zeitschrift für zellforschung und mikroskopische anatomie*, vol. 79, no. 2, pp. 265-271. (1)
- 2. Carr, 1967 Carr, K.E. 1967, "Fine structure of crystalline inclusions in the globule leucocyte of the mouse intestine", *Journal of anatomy*, vol. 101, no. 4, pp. 793-803. (2)
- 3. Carr and Whur, 1968 Carr, K.E. & Whur, P. 1968, "Ultrastructure of globule leucocyte inclusions in the rat and mouse", *Zeitschrift für zellforschung und mikroskopische anatomie*, vol. 86, no. 2, pp. 153-162. (4)
- Moore et al, 1975 Moore, M.R., Meredith, P.A., Goldberg, A., Carr, K.E., Toner, P.G. & Lawrie, T.D. 1975, "Cardiac effects of lead in drinking water of rats", *Clinical science and molecular medicine*, vol. 49, no. 4, pp. 337-341. (13)

Chapter 1 - Section 2 Ongoing studies of biological sections

- 1. Ettarh and Carr, 1996a *Ettarh, R.R. & Carr, K.E. 1996a, "Morphometric analysis of the small intestinal epithelium in the indomethacin-treated mouse", *Journal of anatomy*, vol. 189, no. 1, pp. 51-56. (**79**)
- Abbas et al, 1989 *Abbas, B., Hayes, T.L., Wilson, D.J. & Carr, K.E. 1989, "Internal structure of the intestinal villus: morphological and morphometric observations at different levels of the mouse villus", *Journal of anatomy*, vol. 162, pp. 263-273. (52)
- Sagher et al, 1991 Sagher, F.A., Dodge, J.A., Johnston, C.F., Shaw, C., Buchanan, K.D. & Carr, K.E. 1991, "Rat small intestinal morphology and tissue regulatory peptides: effects of high dietary fat", *British journal of nutrition*, vol. 65, no. 1, pp. 21-28. (62)
- McCullough, J.S. et al, 1998 McCullough, J.S., Ratcliffe, B., Mandir, N., Carr, K.E. & Goodlad, R.A. 1998, "Dietary fibre and intestinal microflora: effects on intestinal morphometry and crypt branching", *Gut*, vol. 42, no. 6, pp. 799-806. (89)

Chapter 2 Scanning electron microscopy emerging as a biological probe

Chapter 2 - Section 1 Early scanning electron microscopy and links to Chapter 1

- 1. Carr and Toner, 1968 Carr, K. E. & Toner, P. G. 1968, "Scanning electron microscopy of rat intestinal villi", *The lancet*, vol. 2, pp. 570-571. (3)
- 2. Ferguson et al, 1969 Ferguson, A., Maxwell, J.D. & Carr, K.E. 1969, "Progressive changes in the small-intestinal villous pattern with increasing length of gestation", *The journal of pathology*, vol. 99, no. 1, pp. 87-91. (5)
- 3. Carr et al, 1979a Carr, K.E., Joffe, S.N., Toner, P.G. & †Watt, C. 1979a, "A preliminary study of scanning electron microscopic changes of the duodenum during healing of duodenal ulcers", *Scandinavian journal of gastroenterology*, vol. 14, supplement 54, pp. 78-83. (**17**)

- 4. Carr, 1970c Carr, K.E. 1970c, "Scanning electron microscope studies of human skin", *British journal of plastic surgery*, vol. 23, pp. 66-72. (7)
- Carr and Carr, 1970 Carr, K.E. & Carr, I. 1970, "How cells settle on glass: a study by light and scanning electron microscopy of some properties of normal and stimulated macrophages", *Zeitschrift für zellforschung und mikroskopische anatomie*, vol. 105, no. 2, pp. 234-241. (6)
- Cook et al, 1979 Cook, G.T., Carr, K.E. & Duncan, H.J. 1979, "The influence of morphological differences in bracken pinnules on the foliar uptake of aminotriazole", *Annals of applied biology*, vol. 93, no. 3, pp. 311-317. (18)
- Al-Jaff et al, 1982 Al-Jaff, D.M.A., Cook, G.T., Carr K.E., & Duncan H.J. 1982, "Further studies on bracken morphology in relation to herbicide uptake", in *The plant cuticle. Linnean society symposium series 10*, Eds. Cutler, D. F., Alvin, K. L. & Price, C. E., Academic Press, London, pp. 293–301. (26)
- Chapter 2 Section 2 Early reviews of biological scanning electron microscopy
- 1. Carr, 1971 Carr, K.E. 1971. "Applications of scanning electron microscopy in biology", *International review of cytology*, vol. 30, pp. 183-255. Academic Press. (8)
- 2. Toner et al, 1971 Toner, P.G., Carr, K.E. & Wyburn, G.M. 1971. "The digestive system an ultrastructural atlas and review" (excerpt), Butterworths, London. (9)
- 3. Carr et al, 1974 Carr, K.E., Shaw Dunn, J. & Toner, P.G. 1974, "Scanning electron microscopy of the alimentary tract", *Scottish medical journal*, vol. 19, no. 5, pp. 211-220. (12)

Chapter 2 - Section 3 Correlation of SEM images with those from established techniques

- 1. Carr et al, 1981a Carr, K.E., Hamlet, R., Nias, A.H.W. & †Watt, C. 1981a, "Multinucleate giant enterocytes in small intestinal villi after irradiation", *Journal of microscopy*, vol. 123, no. 2, pp. 169-176. (20)
- Carr et al, 1981b Carr, K.E., Wong, A.L., Young, D.G., Toner, P.G. & †Watt, C. 1981b, "Scanning electron microscopy and resin histology of large bowel biopsies", *Scottish medical journal*, vol. 26, no. 2, pp. 103-114. (23)
- 3. Saleh et al, 1982 (*)Saleh, K.M., Toner, P.G., Carr, K.E. & Hughes, H.E. 1982, "An improved method for sequential light and scanning electron microscopy of the same cell using localising microcoverslips", *Journal of clinical pathology*, vol. 35, no. 5, pp. 576-580. (29)
- Carr et al, 1984a Carr, K.E., Kamel, H.M.H., Toner, P.G., McGadey, J. & Wong, A. 1984a, "Correlative scanning electron microscopy in gastrointestinal pathology", *Scanning electron microscopy*, vol. 5, pp. 761-772. (37)

Chapter 2 - Section 4 Development of scanning electron microscopy techniques

- 1. Carr and McGadey, 1974 Carr, K.E. & †McGadey, J. 1974, "Staining of biological material for the scanning electron microscope", *Journal of microscopy*, vol. 100, no. 3, pp. 323-330. (11)
- 2. Hodges et al, 1985 Hodges, G.M., Carr, K.E., Hume, S.P., Marigold, J.C., Southgate, J. & Marshall, J.F. 1985, "Changes in surface structure and concanavalin A-binding capacity of

urothelium in the mouse bladder after whole-body neutron irradiation", *Scanning electron microscopy*, vol. 4, pp. 1603-1614. (40)

- Carr et al, 1983a Carr, K.E., Hayes, T.L., McKoon, M., Sprague, M. & Bastacky, S.J. 1983a, "Low temperature scanning electron microscope studies of mouse small intestine", *Journal of microscopy*, vol. 132, no. 2, pp. 209-217. (32)
- 4. Carr et al, 1983b Carr K.E., Hayes T.L., Hume S.P. & (*)Kamel H.M.H. 1983b, "Qualitative and quantitative SEM of gastrointestinal tract", *Proceedings of the 41st annual meeting of the electron microscope society of America*, pp. 496-499. (**31**)
- 5. Carr et al, 1984b Carr, K.E., Hayes, T.L., McKoon, M., Bastacky, S.J. & (*)Kamel, H.M. 1984b, "Etched surfaces of plastic embedded and frozen hydrated gastrointestinal tissue", *Journal of submicroscopic cytology*, vol. 16, no. 2, pp. 219-226. (**36**)
- 6. Carr et al, 1986a Carr, K.E., Hayes, T.L., Watt, A., Bastacky, S.J., Klein, S. & †Fife, M. 1986a, "Specimen handling and data interpretation in X-ray microanalysis of frozen hydrated etched gastrointestinal tract", *Journal of electron microscopy technique*, vol. 4, no. 4, pp. 371-379. (43)
- 7. Carr et al, 1987a Carr, K.E., Hayes, T.L., Watt, A., Klein, S., McKoon, M., Bastacky, S.J. & †Ellis, S. 1987a, "Biological variation in cryo-microanalytical data from mouse small intestinal villi", *Journal of electron microscopy technique*, vol. 5, no. 1, pp. 65-74. (45)
- Dickson et al, 1989 Dickson, G.R., McKenna, S., McColl, K. & Carr, K.E. 1989, "Examination of mucosal surfaces by scanning electron microscopy before and after removal of debris", *Journal of microscopy*, vol. 153, no. 1, pp. 75-79. (54)
- 9. Carr et al, 1989 Carr, K.E., Ellis, S., Hayes, T.L. & Ainsworth, E.J. 1989, "Multiview technique for SEM study of intestinal villi", *Scanning*, vol. 11, no. 2, pp. 109-110. (53)
- Abbas et al, 1990a *Abbas, B., Boyle, F.C., Wilson, D.J., Nelson, A.C. & Carr, K.E. 1990a, "Radiation induced changes in the blood capillaries of rat duodenal villi: a corrosion cast, light and transmission electron microscopical study", *Journal of submicroscopic cytology and pathology*, vol. 22, no. 1, pp. 63-70. (55)
- Chapter 2 Section 5 Scanning around the alimentary system
- Nunn et al, 1990 *Nunn, S., Gilmore, R.S., Dodge, J.A. & Carr, K.E. 1990, "Exudate variation in the rabbit gastrointestinal tract: a scanning electron microscope study", *Journal of anatomy*, vol. 170, pp. 87-98. (58)
- Laferla et al, 1988 Laferla, G.A., Shasha'a, S., Crean, G.C., Gilmore, R.S. & Carr, K.E. 1988, "Topographical variation in the mucosal surface of oesophageal biopsies", *Journal of submicroscopic cytology and pathology*, vol. 20, no. 4, pp. 731-739. (51)
- 3. Shasha'a et al, 1993 *Shasha'a, S., Dickson, G.R., Gilmore, R.S., Crean, G.C., Butt, M.M. & Carr, K.E., 1993, "Rabbit and human non-keratinising stratified squamous oesophageal epithelium displays similar microridge structure by scanning electron microscopy", *Scanning microscopy*, vol. 7, no. 3, pp. 953-959. (67)

- 4. Johnston et al, 1996 Johnston, B.T., Nunn, S., Sloan, J.M., Collins, J.S.A., Mcfarland, R.J., Parkin, S., Carr, K.E. & Collins, B.J. 1996, "The application of microridge analysis in the diagnosis of gastro-oesophageal reflux disease", *Scandinavian journal of gastroenterology*, vol. 31, no. 2, pp. 97-102. (82)
- 5. Ferguson et al, 1978 Ferguson, A., Carr, K.E., Macdonald, T.T. & †Watt, C. 1978, "Hypersensitivity reactions in the small intestine, iv influence of allograft rejection on small intestinal mucosal architecture: a scanning and transmission electron microscope study", *Digestion*, vol. 18, no. 1-2, pp. 56-63. (15)
- 6. Ettarh and Carr, 1993 *Ettarh, R.R. & Carr, K.E. 1993, "Structural and morphometric analysis of murine small intestine after indomethacin administration", *Scandinavian journal of gastroenterology*, vol. 28, no. 9, pp. 795-802. (66)
- Ettarh and Carr, 1997 *Ettarh, R.R. & Carr, K.E. 1997, "A morphological study of the enteric mucosal epithelium in the streptozotocin-diabetic mouse", *Life Sciences*, vol. 61, no. 18, pp. 1851-1858. (87)
- McClean et al, 1996 McClean, P., Dodge, J.A., Nunn, S., Carr, K.E. & Sloan, J.M. 1996, "Surface features of small-intestinal mucosa in childhood diarrheal disorders", *Journal of pediatric gastroenterology and nutrition*, vol. 23, no. 5, pp. 538-546. (83)
- Galloway et al, 1987 Galloway, D.J., Jarrett, F., Boyle, P., Indran, M., Carr, K., Owen, R.W. & George, W.D. 1987, "Morphological and cell kinetic effects of dietary manipulation during colorectal carcinogenesis", *Gut*, vol. 28, no. 6, pp. 754-763. (46)
- 10. Ettarh and Carr 1996b *Ettarh, R.R. & Carr, K.E. 1996b, "Ultrastructural observations on the peritoneum in the mouse", *Journal of anatomy*, vol. 188, no. 1, pp. 211-215. (80)
- 11. Simpson et al, 1983 Simpson, C.J., Toner, P.G., Carr, K.E., Anderson, J.D. & Carter, D.C. 1983, "Effect of bile salt perfusion and intraduct pressure on ionic flux and mucosal ultrastructure in the pancreatic duct of the cat", *Virchows archiv abteilung B cell pathology*, vol. 42, no. 3, pp. 327-342. (**34**)

Chapter 2 – Section 6 Later reviews of biological scanning electron microscopy

- 1. Carr, 1980 Carr, K.E. 1980, "Scanning electron microscopy of macrophages", In '*The reticuloendothelial system*' Vol I. Eds. Carr, I. & Daems, W.T., Plenum Publishing Corporation, pp. 259-295. (**19**)
- 2. Carr et al, 1982a Carr, K.E., Toner, P.G. & (*)Saleh, K.M. 1982a, "Scanning electron microscopy", *Histopathology*, vol. 6, no. 1, pp. 3-24. (27)
- 3. Carr and Toner, 1982 Carr, K.E. & Toner, P.G. 1982. "Cell structure. An introduction to biomedical electron microscopy" (3rd edition, excerpt), Churchill Livingstone. (28)
- 4. Carr and Toner, 1984 Carr K.E., & Toner P.G. 1984, "Morphology of the intestinal mucosa", Handbook of experimental pharmacology, vol. 70, no. 1, pp. 1-50. (**38**)

Chapter 3 Correlative morphology of irradiated small intestinal surfaces

Chapter 3 - Section 1 Villous radiation-induced damage and links to Chapter 2

- 1. Carr and Toner, 1972 Carr, K.E. & Toner, P.G. 1972, "Surface studies of acute radiation injury in the mouse intestine", *Virchows archiv B zell-pathologie*, vol. 11, no. 1, pp. 201-210. (10)
- Hamlet et al, 1976 (*)Hamlet, R., Carr, K.E., Toner, P.G. & Nias, A.H.W. 1976, "Scanning electron microscopy of mouse intestinal mucosa after cobalt 60 and D-T neutron irradiation", *British journal of radiology*, vol. 49, no. 583, pp. 624-629. (14)
- 3. Hamlet et al, 1981 (*)Hamlet, R., Carr, K.E., Nias, A.H.W. & †Watt, C. 1981, "Surface damage in the small intestine of the mouse after X or neutron irradiation", *Scanning electron microscopy*, no. 4, pp. 73-78. (25)
- Carr et al, 1979b Carr, K.E., Hamlet, R., Nias, A.H.W. & †Watt, C. 1979b, "Lack of correlation between villus and crypt damage in irradiated mouse intestine", *British journal of radiology*, vol. 52, no. 618, pp. 485-493. (16)

Chapter 3 - Section 2 Early reviews on 'topographical' and ultrastructural radiation effects

- 1. Carr, 1981 Carr, K.E. 1981, "Scanning electron microscopy of tissue response to irradiation", *Scanning electron microscopy*, vol. 4, pp. 35-46. (24)
- 2. Carr and Seed, 1983 Carr K.E. & Seed T.M. 1983, "Ultrastructural effects of radiation on tissues and cells" (excerpt), soft-cover book, 17 papers, *Scanning electron microscopy*. (33)

Chapter 3 - Section 3 Radiation effects at different sites

- Robertson et al, 1987 Robertson, A.G., Wilson, P., Wilson, D.J., Carr, K.E. & Hunter, I. 1987, "Microplication patterns on human buccal epithelia following radiotherapy: a scanning electron microscopic analysis", *Journal of submicroscopic cytology and pathology*, vol. 19, no. 3, pp. 515-521. (47)
- 2. Wang et al, 1994 *Wang, Q., Dickson, G.R., Abram, W.P. & Carr, K.E. 1994, "Electron irradiation slows down wound repair in rat skin: a morphological investigation", *British journal of dermatology*, vol. 130, no. 5, pp. 551-560. (70)
- 3. Wang et al, 1996 *Wang, Q., Dickson, G.R. & Carr, K.E. 1996, "The effect of graft-bed irradiation on the healing of rat skin grafts", *Journal of investigative dermatology*, vol. 106, no. 5, pp. 1053-1057. (**85**)

Chapter 3 - Section 4 Qualitative radiation effects on small intestinal surfaces

- 1. Indran et al, 1988 *Indran, M., Carr, K.E. & Boyle, F.C. 1988, "Radiation-induced changes in mouse duodenal papilla", *British journal of radiology*, vol. 61, no. 731, pp. 1039-1042. (49)
- Carr et al, 1990 Carr, K.E., Hayes, T.L., Abbas, B. & Ainsworth, E.J. 1990, "Collared crypts in irradiated small intestine", *Journal of submicroscopic cytology and pathology*, vol. 22, no. 2, pp. 265-271. (57)

Chapter 3 - Section 5 Scoring of radiation-induced changes to intestinal surfaces

- Carr et al, 1983c Carr, K.E., Hamlet, R., Nias, A.H.W. & †Watt, C. 1983c, "Damage to the surface of the small intestinal villus: an objective scale of assessment of the effects of single and fractionated radiation doses", *British journal of radiology*, vol. 56, no. 667, pp. 467-475. (30)
- Carr et al, 1986b Carr, K.E., Ellis, S. & Michalowski, A. 1986b, "Surface studies of duodenal lesions induced by thoracic irradiation", *Scanning electron microscopy*, vol. 1, pp. 209-219. (42)

Chapter 3 - Section 6 Importance of underlying structures to surface responses

- 1. Carr et al, 1981c Carr, K.E., Hamlet, R. & †Watt, C. 1981c, "Scanning electron microscopy, autolysis, and irradiation as techniques for studying small intestinal morphology", *Journal of microscopy*, vol. 123, no. 2, pp. 161-168. (21)
- Carr et al, 1981d Carr, K.E., Toner, P.G., McLay, A.L.C. & (*)Hamlet, R. 1981d, "The ultrastructure of some gastrointestinal lesions in experimental animals and man", *Scandinavian journal of gastroenterology, supplement*, vol. 16, pp. 107-128. (22)
- 3. Carr et al, 1985 Carr, K.E., Hamlet, R., Nias, A.H.W., Boyle, F.C. & †Fife, M.G. 1985, "Stromal damage in the mouse small intestine after Cobalt 60 gamma or D-T neutron irradiation", *Scanning electron microscopy*, vol. 5, pp. 1615-1621. (**39**)
- 4. Carr et al, 1987b Carr, K.E., Hayes, T.L., Indran, M., Bastacky, S.J., McAlinden, G., Ainsworth, E.J. & †Ellis, S. 1987b, "Morphological criteria for comparing effects of X-rays and neon ions on mouse small intestine", *Scanning microscopy*, vol. 1, no. 2, pp. 799-809. (44)
- 5. Abbas et al, 1990b *Abbas, B., Hume, S.P., McCullough, J.S., Wilson, D.J., Stewart, P.C. & Carr, K.E. 1990b, "Early morphological changes in blood capillaries of mouse duodenal villi induced by X-irradiation", *Journal of submicroscopic cytology and pathology*, vol. 22, no. 4, pp. 609-614. (56)
- 6. Carr et al, 1984c Carr, K.E., Hamlet, R., Nias, A.H.W. & †Watt, C. 1984c, "Morphological differences in the response of mouse small intestine to radiobiologically equivalent doses of X and neutron irradiation", *Scanning electron microscopy*, vol. 5, no. 1, pp. 445-454. (**35**)
- 7. Wyatt et al, 1987 *Wyatt, M.G., Hume, S.P., Carr, K.E. & Marigold, J.C.L. 1987, "A preliminary study of the role of gastrointestinal endocrine cells in the maintenance of villous structure following X-irradiation", *Scanning microscopy*, vol. 1, no. 1, pp. 291-300. (**48**)

Chapter 3 - Section 7 Mechanisms of structural damage

- Kamel et al, 1988 (*)Kamel, H.M.H., Hume, S.P., Carr, K.E., Marigold, J.C.L. & Michalowski, A. 1988, "Development of villous damage in mouse small intestine after local hyperthermia or irradiation", *Journal of submicroscopic cytology and pathology*, vol. 20, no. 1, pp. 185-193. (50)
- Indran et al, 1985 *Indran, M., Boyle, F.C. & Carr, K.E. 1985, "Simulation with pharmacological agents of radiation damage to small intestinal villi", *Scanning electron microscopy*, vol. 3, pp. 1165-1175. (41)

- 3. Indran et al, 1991 *Indran, M., Carr, K.E., Gilmore, R.S. & Boyle, F.C. 1991, "Mucosal changes in mouse duodenum after gamma-irradiation or reserpine treatment", *Journal of submicroscopic cytology and pathology*, vol. 23, no. 2, pp. 267-278. (61)
- 4. Carr et al, 1991a Carr, K.E., Bullock, C., Ryan, S.S., McAlinden, M.G. & Boyle, F.C. 1991a, "Radioprotectant effects of atropine on small intestinal villous shape", *Journal of submicroscopic cytology and pathology*, vol. 23, no. 4, pp. 569-577. (59)

Chapter 3 - Section 8 Summary of small intestinal responses

- Carr et al, 1991c Carr, K.E., McCullough, J.S., Nunn, S., Hume, S.P. & Nelson, A.C. 1991c, "Neutron and X-ray effects on small intestine summarized by using a mathematical model or paradigm", *Proceedings of the royal society B: biological sciences*, vol. 243, no. 1308, pp. 187-194. (60)
- 2. Carr et al, 1992a Carr, K.E., Nelson, A.C., Hume, S.P. & McCullough, J.S. 1992a, "Characterization through a data display of the different cellular responses in X-irradiated small intestine", *Journal of radiation research*, vol. 33, no. 2, pp. 163-177. (64)
- Carr et al, 1996a Carr, K.E., Hume, S.P., Nelson, A.C., O'Shea, O., Hazzard, R.A. & McCullough, J.S. 1996a, "Morphological profiles of neutron and X-Irradiated small intestine", *Journal of radiation research*, vol. 37, no. 1, pp. 38-48. (77)
- 4. Carr et al, 1992b Carr, K.E., McCullough, J.S., Nelson, A.C., Hume, S.P., Nunn, S. & (*)Kamel, H.H.M. 1992b, "Relationship between villous shape and mural structure in neutron irradiated small intestine", *Scanning microscopy*, vol. 6, no. 2, pp. 561-572. (63)
- 5. Carr et al, 1993 Carr K.E., McCullough J.S., Gilmore, R. S., Abbas, B., Hume S.P., Nelson A.C., Hayes T.L. & Ainsworth E.J. 1993, "Effects of neon ions, neutrons and X rays on small intestine", in *Biological effects and physics of solar and galactic cosmic radiation. NATO ASI Series (Series A: life sciences)*, vol 243A. Eds. Swenberg, C.E., Horneck, G. & Stassinopoulos, E.G., Springer, Boston, MA, Plenum Press, New York, Part A, pp. 203-216. (65)
- Carr et al, 1994 Carr, K.E., McCullough, J.S., Brennan, P., Hayes, T.L., Ainsworth, E.J. & Nelson, A.C. 1994, "Heavy ion induced changes in small intestinal parameters", *Advances in space research*, vol. 14, no. 10, pp. 521-530. (69)
- Chapter 3 Section 9 Different cell populations
- 1. Brennan et al, 1998 (*)Brennan, P.C., Carr, K.E., Seed, T. & McCullough, J.S. 1998, "Acute and protracted radiation effects on small intestinal morphological parameters", *International journal of radiation biology*, vol. 73, no. 6, pp. 691-698. **(88)**
- 2. Ettarh et al, 2000 *Ettarh, R.R., Hodges, G.M. & Carr, K.E. 2000, "Radiation effects in the small bowel of the diabetic mouse", *International journal of radiation biology*, vol. 76, no. 2, pp. 241-248. (91)
- 3. Brennan et al, 1999 (*)Brennan, P.C., McCullough, J.S. & Carr, K.E. 1999, "Variations in cell and structure populations along the length of murine small intestine", *Cells tissues organs*, vol. 164, no. 4, pp. 221-226. (90)

Chapter 3 - Section 10 Reviews of irradiated small intestine

- 1. Carr et al, 1995 Carr K.E., Hume S.P., Ettarh R., Carr E.A. & McCullough J.S., 1995, "Radiation-induced changes to epithelial and non-epithelial tissue", in *Radiation and the gastrointestinal tract*, Eds. Dubois A., King, G.L. & Livengood. D.R., Taylor and Francis, pp. 113-128. (68)
- 2. Carr, 2001 Carr, K.E. 2001, "Effects of radiation damage on intestinal morphology", *International review of cytology*, vol. 208, pp.1-119. (93)

Chapter 4 Microparticles crossing intestinal surfaces

Chapter 4 - Section 1 Questions on uptake and links to Chapter 3

- Carr et al, 1997 Carr K.E., McCullough S.J., Hazzard R.A., Brennan P., McGuiness C., Seed, T. & Hodges G.M. 1997, "Structural and functional profiles of mammalian small intestine", in *Recent advances in microscopy of cells, tissues and organs*, Ed. Motta P.M., pp. 381-386. (86)
- Carr et al, 1996b Carr K.E., Dayan A.D., Florence A.T., Goodlad R.A., Hodges G.M. & Rowlands B.J. 1996b, "Symposium on particle translocation across epithelial membranes", *Journal of anatomy*, vol. 189, p. 475. (75)
- Hodges et al, 1995a Hodges, G.M., Carr, E.A., Hazzard, R.A., O'Reilly, C. & Carr, K.E. 1995a, "A commentary on morphological and quantitative aspects of microparticle translocation across the gastrointestinal mucosa", *Journal of drug targeting*, vol. 3, no. 1, pp. 57-60. (73)

Chapter 4 - Section 2 What model? Model 1, the standard in vivo in situ model

1. Hodges et al, 1995b - Hodges, G.M., Carr, E.A., Hazzard, R.A. & Carr, K.E. 1995b, "Uptake and translocation of microparticles in small intestine - morphology and quantification of particle distribution", *Digestive diseases and sciences*, vol. 40, no. 5, pp. 967-975. (72)

Chapter 4 - Section 3 Model 1, variation in particle type

- 1. McCullough et al, 1995 McCullough, J.S., Hodges, G.M., Dickson, G.R., Yarwood, A. & Carr, K.E. 1995, "A morphological and microanalytical investigation into the uptake of particulate iron across the gastrointestinal tract of rats", *Journal of submicroscopic cytology and pathology*, vol. 27, no. 1, pp. 119-124. (74)
- Cartwright-Shamoon et al, 1995 (*)Cartwright-Shamoon, J., Dickson, G.R., Dodge, J. & Carr, K.E. 1995, "Morphological aspects of particle translocation in vivo following ingestion of the yeast Saccharomyces boulardii", *Journal of drug targeting*, vol. 3, no. 1, pp. 61-63. (71)
- 3. Carr et al, 2001 Carr K.E., Youngman M.J., Barr M., Colhoun L. & Harrison J.D. 2001, "Visualisation of fused aluminoscilicate microparticles taken up by intestinal tissue", *Journal* of anatomy, vol. 199, pp. 225–226. (92)

Chapter 4 - Section 4 Model 1, developments in methodology

1. Hazzard et al, 1996 - *Hazzard, R.A., Hodges, G.M., Scott, J.D., McGuinness, C.B. & Carr, K.E. 1996, "Early intestinal microparticle uptake in the rat", *Journal of anatomy*, vol. 189, no. 2, pp. 265-271. (**81**)

- 2. McMinn et al, 1996 *McMinn, L.H., Hodges, G.M. & Carr, K.E. 1996, "Gastrointestinal uptake and translocation of microparticles in the streptozotocin-diabetic rat", *Journal of anatomy*, vol. 189, no. 3, pp. 553-559. **(84)**
- Carr et al, 1996c Carr, K.E., Hazzard, R.A., Reid, S. & Hodges, G.M. 1996c, "The effect of size on uptake of orally administered latex microparticles in the small intestine and transport to mesenteric lymph nodes", *Pharmaceutical research*, vol. 13, no. 8, pp. 1205-1209. (76)
- Cartwright-Shamoon et al, 1996 (*)Cartwright-Shamoon, J., Dickson, G.R., Dodge, J. & Carr, K.E. 1996, "Uptake of yeast (Saccharomyces boulardii) in normal and rotavirus treated intestine", *Gut*, vol. 39, no. 2, pp. 204-209. (78)
- 5. Smyth et al, 2005 *Smyth, S.H., Doyle-McCullough, M., Cox, O.T. & Carr, K.E. 2005, "Effect of reproductive status on uptake of latex microparticles in rat small intestine", *Life sciences*, vol. 77, no. 26, pp. 3287-3305. (94)
- Doyle-McCullough et al, 2007 *Doyle-McCullough, M., Smyth, S.H., Moyes, S.M. & Carr, K.E. 2007, "Factors influencing intestinal microparticle uptake in vivo", *International journal* of pharmaceutics, vol. 335, no. 1-2, pp. 79-89. (95)
- 7. Smyth et al, 2008 *Smyth, S.H., Feldhaus, S., Schumacher, U. & Carr, K.E. 2008, "Uptake of inert microparticles in normal and immune deficient mice", *International journal of pharmaceutics*, vol. 346, no. 1-2, pp. 109-118. (**98**)

Chapter 4 - Section 5 What model? Model 2, the standard in vitro model

1. Moyes et al, 2007 - *Moyes, S.M., Smyth, S.H., Shipman, A., Long, S., Morris, J.F. & Carr, K.E. 2007, "Parameters influencing intestinal epithelial permeability and microparticle uptake in vitro", *International journal of pharmaceutics*, vol. 337, no. 1-2, pp. 133-141. (96)

Chapter 4 - Section 6 Model 2, developments in methodology

- 1. Moyes et al, 2010a *Moyes, S.M., Morris, J.F. & Carr, K.E. 2010a, "Culture conditions and treatments affect Caco-2 characteristics and particle uptake", *International journal of pharmaceutics*, vol. 387, no. 1-2, pp. 7-18. (99)
- 2. Moyes et al, 2011 *Moyes, S.M., Morris, J.F. & Carr, K.E. 2011, "Roles of pre-treatment time and junctional proteins in Caco-2 cell microparticle uptake", *International journal of pharmaceutics*, vol. 407, no. 1-2, pp. 21-30. (**101**)
- Moyes et al, 2008 *Moyes, S.M., Killick, E.M., Morris, J.F., Kadhim, M.A., Hill, M.A. & Carr, K.E. 2008, "Changes produced by external radiation in parameters influencing intestinal permeability and microparticle uptake in vitro", *International journal of radiation biology*, vol. 84, no. 6, pp. 467-486. (97)
- Moyes et al, 2010b *Moyes, S.M., Morris, J.F. & Carr, K.E. 2010b, "Macrophages increase microparticle uptake by enterocyte-like Caco-2 cell monolayers", *Journal of anatomy*, vol. 217, no. 6, pp. 740-754. (100)
- 5. Carr, personal communication, 2018 (103)

Chapter 4 - Section 7 What model? Invited Review on microparticle uptake

1. Carr et al, 2012 - Carr, K.E., Smyth, S.H., McCullough, M.T., Morris, J.F. & *Moyes, S.M. 2012, "Morphological aspects of interactions between microparticles and mammalian cells: intestinal uptake and onward movement", *Progress in histochemistry and cytochemistry*, vol. 46, no. 4, pp. 185-252. (**102**)

Preface

The thesis is subdivided according to subject matter, all connected by the themes of probing surfaces and responses to environmental challenge, Chapter 1 on ultrastructure, Chapter 2 on emerging scanning electron microscopy, Chapter 3 on integrative responses to irradiation, Chapter 4 on microparticle uptake and Chapter 5 integrating the whole.

With respect to my contributions, in all submissions I was involved in some or all of the following: project initiation; protocol construction; data collection and analysis; and preparation and revision of the published manuscript. My role was, for Chapter 1, as electron microscopist and gut researcher, for Chapter 2 as correlative scanning electron microscopist, for Chapter 3 as morphologist, group leader and grant-holder in radiobiology topics and for Chapter 4 as group leader and grant-holder in a programme on microparticle uptake. All but 2 of the 103 submissions have been fully published, one an Abstract and one a personal communication on data currently being analysed.

With respect to author order, most of the 101 submissions were either sole (6), first (42) or last (33) author, signifying a major contribution to the work. Sole authorship was related to early work or review articles. First author signified: 'hands-on' involvement in data production, in this case image recording, analysis and interpretation; group leadership of a specific project; or prime responsibility for manuscript content and writing, due perhaps to a difficulty in allotting this position to one other author. Being in the last position often implied the first author place was a student or young colleague. The remaining submissions were middle position authorships (20), indicating: the order was chosen in the interests of another; a collaborator from another group was in the first or last position; or the paper arose from involvement in projects centred in another group. Further information on my role and on collaborative input is given in the Acknowledgements section and elsewhere.

In terms of organisation, the thesis is presented in three volumes, the first containing the front matter, linking text describing the work in context and a list of references. The other two volumes contain the accompanying papers and extracts from books, those for Chapters 1 and 2 in Volume 2 and those for Chapters 3 and 4 in Volume 3.

In terms of lay-out, in Volume 1, each Chapter has, at the beginning, an excerpt from the Presubmission Abstract. In Volumes 2 and 3, each Chapter has a 'Contents' page giving a tabular list of the papers, with Chapter number, Section number and the number of the paper in the order in which they are presented.

In terms of the content of the front matter, the list of accompanying material contains the Reference list in standard Harvard format as in the University recommendations, but also has information on the order of appearance of the submissions in the thesis, giving Chapter, Section and paper number. This does not necessarily present the publications chronologically as in the Presubmission papers, but matches the scientific development of the research theme. The chronological order is, however, also given.

With respect to content of Volume 1, the main text aims to place 'the whole work critically into perspective with the general state of knowledge in the field of investigation to which the candidate's researches are related' (Regulation 4). This has been done by putting each submission into context with the scientific knowledge at the time it was written and also with the ongoing literature. That was analysed through its citation history, using Scopus and Google Scholar, consulted in the months before thesis submission: only those citations available in full through Scopus or Google Scholar were used and mentioned in the text. Cross referencing within the text to other papers or parts of the thesis is done by providing information on which Chapter and Section are involved: this is explained in footnotes as the first examples appear. The Reference list at the end of the text is in Harvard format and contains in one alphabetically arranged document the list of accompanying material, the list of supporting references and the list of citations to the submissions.

With respect to Copyright, the thesis relies on the hard copy of approximately one hundred published submissions, on all of which I am an author. It has been assumed that these will not be made available on line, since this would be a large task. Obtaining Copyright permission would also be 'too onerous' (University guidelines) and in many cases impossible because of the passage of time and difficulty of identifying and contacting original authors and publishers or their representatives. In the text, citation of papers by others has relied on my interpretation of their content: direct quotation is obvious from the format and care has been taken to check accuracy. Permission has not been sought for citation of the papers or for quoting them: their use is little different from standard scientific practice in paper writing and some of the authors are long out of the scientific community, making it impossible to treat all in the same way. Apologies are sincerely offered to any authors omitted or misquoted and for any other errors in the referencing or citations.

In terms of binding, the text is bound in a separate volume from the accompanying material to allow for the reading of Volume 1, while the relevant paper is open in Volume 2 or 3.

Acknowledgements

The work would not have been possible without contributions from others, indicated here, in the Preface and 'Context and comment' subsections of the Chapters and, in the submissions, in the authorship order, academic addresses and Acknowledgements.

The research was carried out while on staff at the Universities of Glasgow¹ (1965 – 1967), Strathclyde (1968-1969), Glasgow² (1970 – 1985) and Queen's University Belfast (1985 to 1999). There were also attachments to other Institutions for laboratory visits, on sabbatical or after early retirement, as Guest Researcher/Academic Visitor at the Donner/Lawrence Berkeley Laboratory (LBL), California, USA (1982 to 1987), The MRC Radiation and Genome Stability Unit (RAGSU), Harwell (2001 – 2007) and The University of Oxford Department of Anatomy, now the Department of Physiology, Anatomy and Genetics (2001 to current as Academic Visitor in Anatomy). All dates are approximate.

The work was supported by Departmental or central microscope, laboratory, library and computing facilities at all the Institutions: senior staff involved included, among others: Professors George Wyburn, Robert Kenedi, Dame Kay Davies; Gary Love; Dudley Goodhead (Heads of Department, Faculty or Unit). For twelve years in Belfast, I was Head of the Department of Anatomy and then Director of the School of Biomedical Science. The University purchased a state-of-the-art scanning electron microscope, to add to the transmission microscope already there and complemented later by a confocal scanning laser microscope: many of the staff and students there contributed to the work. In research on imaging, excellent technical support is essential and this has been enthusiastically given in all Departments and Institutions, some technicians identified through authorship in the list of accompanying material, as are some research students, mostly doctoral.

Equipment, project grants, or travel support came, either Departmentally or personally, from, among others, the Scottish Home and Health Department, The Cancer Research Campaign, The Medical Research Council, The Imperial Cancer Research Fund, The Royal Society and particularly most recently the Department of Health Radiation Protection Research Programme. The initial research at LBL, when the international collaborations began, was additionally supported at first by the Lister Centenary Travelling Fellowship from the University of Glasgow and then by the Wellcome Trust.

There have been many collaborating scientists, both internal and external. These include senior colleagues who gave ideas and support in the beginning, academic and technical members of staff in the Department/School in Belfast and others, both staff and students,

who have been involved through the years in Glasgow, Strathclyde, Belfast, Harwell and Oxford. Many colleagues have contributed, but there is only space here to mention a few, such as those participating while the research was taking a new direction. These included, within the home institutions, Frances Boyle, Glenn Dickson, Roy Hamlet, Stephen McCullough, John Morris and Tony Nias: some of them moved on to other Institutions and continued to collaborate from there. Externally, they included John Ainsworth (LBL, then Armed Forces Radiobiology Research Institute, AFRRI), Tom Hayes (LBL), Gisele Hodges (Imperial Cancer Research Fund, ICRF, then Belfast), Sue Hume (Hammersmith Hospital, London), Tom Seed (Argonne National Laboratory, Illinois) and Udo Schumacher (University Medical Centre, Hamburg Eppendorf). There have been major contributions from students and research assistants. With them, as for colleagues, space limitations here required selection of some from this group. Concentrating on those whose authorships indicated continuing contributions to gut-centred papers, these included Babar Abbas, Patrick Brennan, Eúnan Carr (EC), Jalal Cartwright-Shamoon, Melissa Doyle-McCullough, Raj Ettarh, Rosemary Hazzard, Mallika Indran, Hassan Kamel, Sheila Nunn, Siobhan Moyes and Sharon Smyth. The collaborators have included the author's brother, Ian Carr, and husband, Peter Toner, but the former student EC is not a relative.

With respect to the writing of the thesis, the declaration makes clear that this was entirely my work: it is therefore considerably the poorer for being without the contributions and editing that come from others during the preparation of articles, book chapters or books. However, at the beginning of the project, the following three colleagues at Oxford read and commented on the first draft of the Presubmission Abstract: Maike Glitsch, Deborah Goberdhan and John Morris. At the end of the project, my husband Peter Toner assisted with the checking of the final reference list. Additionally, I gained substantially with respect to thesis construction from reference to DSc volumes in the Glasgow University Library.

To all these Institutions, funding bodies, individuals and colleagues or students, I am most grateful. In addition, since the work has extended over many years, being postponed for various academic reasons, I owe a debt of gratitude to family and to friends: some of these became scientific collaborators and in other cases collaborators became friends. Finally and in memoriam, two of the collaborating scientists mentioned above are now dead, Frances Boyle and Gisele Hodges: the writing of the thesis has been a reminder of their contributions and of the loss their passing has been to science and to friendship.

To them all, whether named collectively or individually, many thanks.

Author's declaration

The thesis 'Probing tissue surfaces' submitted to The University of Glasgow in fulfilment of the requirements of the degree of Doctor of Science in Medicine has been written entirely by myself.

None of the material included in the thesis has been submitted by me for any other Higher degree at any Institution.

Katharine E Carr

Abbreviations

Key techniques

Abbreviation	Term	in fu	ll
--------------	------	-------	----

BEI	Backscattered electron image/imaging
EM	Electron micrograph/microscope/microscopy
LM	Light micrograph/microscope/microscopy
SEM	Scanning electron micrograph/microscope/microscopy
TEM	Transmission electron micrograph/microscope/microscopy
XRMA	X-Ray microanalysis

Institutions, departments and grant bodies

Abbreviation Term in full

AEA	Atomic Energy Authority, AEA Technology
AFRRI	Armed Forces Radiobiology Research Institute
CRC	Cancer Research Campaign
ICRF	Imperial Cancer Research Fund
IRSN	Institut de Radioprotection et de Sûreté Nucléaire
LBL	Lawrence Berkeley Laboratory
MIT	Massachusetts Institute of Technology
NRPB	National Radiological Protection Board
RAGSU	Radiation and Genome Stability Unit

Other abbreviations and definitions

Abbreviation	Term in full/definition
D-T	Deuterium-tritium
E cadherin	Epithelial calcium-dependent adhesion
EAP	Epithelial/epithelium associated particle

Abbreviation	Term in full/definition
--------------	-------------------------

FAP	Fused aluminosilicate particle
GALT	Gut-associated lymphoid tissue
Gy	Gray, derived SI unit of ionising radiation dose
IORT	Intra-operative radiotherapy
IS	Injury score, see RIS
LET	Linear energy transfer
M cells	Microfold cells
MALT	Mucosa-associated lymphoid tissue
МСЈ	Multicellular junction
Microparticle	Particle with diameter measurable in micrometres
MLN	Mesenteric lymph node
MoI	Morphological Index
MODS	Multiple organ dysfunction syndrome
n	Sample number
Rad	Unit of absorbed radiation dose
RBE	Relative biological effectiveness, ratio of effectiveness of one type of ionising radiation relative to another
RIS	Radiation injury score
SCID	Severe compromised immunodeficiency
SD	Standard deviation
SMP	Submembranous particle
ΤΝΓα	Tumour necrosis factor alpha
UD	Ultrastructural Deviation, information on subcellular changes from controls
VIP	Vasoactive intestinal peptide/polypeptide
ZO-1	Zonula occludens 1

Chapter 1 Ultrastructure

Preface - extract from Presubmission Abstract (Chapter 1)

Since the work on scanning electron microscopy was facilitated by experience in the interpretation of light and transmission electron microscope images, some papers on these are presented in Chapter 1.

1.1 Fore-runner to surface studies and links to Chapters 2 and 3

The interpretation in the 1960s of the then 'new' images produced by scanning electron microscopy (SEM) and described in Chapter 2 was greatly assisted by a background in histology and in the images produced by transmission electron microscopy (TEM). Papers on tissue ultrastructure have therefore been included in this Chapter, to show the usefulness of images of sections in biological research, beginning with early papers but also including selected articles published through the years, not entirely in chronological order.

The early papers, centred in Anatomy in Glasgow University, applied transmission electron microscopy to several projects carried out in collaboration with other University Departments, beginning an ongoing interest in gastrointestinal questions, but also covering other topics. The papers contributed to the literature in various ways, including technical aspects, subcellular detail and information on environmental effects on human populations.

The first paper (¹Carr et al, 1967), written in collaboration with the Department of Microbiology, showed that TEM was necessary for the identification of the damage produced by dispersion of isolated liver cells for biochemical tests, adding to the information obtained by phase contrast light microscopy (Jacob and Bhargava, 1962). This application of TEM for quality control has remained useful, providing a baseline of morphological integrity. Of the available citations in the Scopus listings, one commented that it had used electron microscopy to demonstrate cellular damage (Walter et al, 1973) and a paper on nuclear parameters quoted it in a Table (Suciu, 1985).

Two other early papers (Carr, 1967, Carr and Whur, 1968) applied previous experience in high resolution TEM imaging, gained during earlier studies of non-biological material

¹ The submitted papers are described in the order given in the List of Accompanying Material. When each appears for the first time in this way, cross reference information is not given.

(Carr, 1970a, b), to the study of the substructure of inclusions in 'granules' of rodent globule leucocytes, cells known to be associated with parasitic infestation. This project was already a topic of interest in The University of Glasgow's Anatomy Department (Toner, 1965, Whur and Johnston, 1967). In Carr (1967) the cells were found in mouse intestine, where the inclusions were crystalline. With respect to citations, ²this paper and that by Carr and Whur (1968) was quoted by Vogel et al (2018) as one of the references relating to the presence of globule leucocytes in various organs and species. It was also cited as considering globule leucocytes to be derived from plasma cells, while other groups opted for mesenchymal or lymphocytic derivation (Heine und Schaeg, 1977). It was still being quoted more recently in the discussion of the development of ideas on the likely source of this cell type (Gorgani-Firouzjaee et al, 2015). The 'crystallite' appearance of the granules was also described as being 'of interest', possibly related to globulin (Hanker et al, 1977). Further ultrastructural evidence, in collaboration with the Veterinary School, (Carr and Whur, 1968) showed that the inclusions were different in rats, being fibrous rather than crystalline. In addition to the citation by Vogel et al (2018), mentioned above, Heine und Schaeg (1977) also quoted it with respect to the linking of increased numbers of globule leucocytes and parasitic infestation with anaphylactic responses. Both these papers on small intestinal mucosa would have benefited from scanning electron microscopy to record the presence of any surface parasites.

A fourth early paper illustrated the advantages of integrating ultrastructural studies with the TEM into established projects elsewhere in the University, in this case the Departments of Materia Medica and Medical Cardiology (Moore et al, 1975). The project related to the simultaneous collection of data on morphological and biochemical effects in rat cardiac tissues after exposure to the levels of lead then found in the soft drinking water in Glasgow: this was linked to earlier findings that there was a connection between cardiovascular problems and water hardness (Stitt et al, 1973). The parameters used were the ultrastructure of cardiac papillary muscle and the levels of the enzymes ferrochelatase and δ -aminolaevulinate dehydratase. The effects were greatest after treatment for 6 months. This paper was cited: as describing lead as damaging heart muscle (Harary and Masamitsu, 1978); for the time of maximum tissue lead levels (Quarterman and Morrison, 1978); and

² Cross reference information is not usually included when a paper is being discussed with respect to its citation history.

with reference to the risk that populations could be exposed to metals corroded from the water distribution system (Alam and Sadiq, 1989).

1.2 Ongoing studies of biological sections

Some more recent publications from the period 1989 to 1998 in Belfast have also been included in this Chapter, to show the continuation of the interest in the use of microscopy to study tissue sections, concentrating on intestine. The papers have been presented in two sets of pairs.

The first pair has been chosen to highlight the importance of plane of section, both stemming from initiatives within the group. The first paper, Ettarh and Carr (1996a), used standard preparations, producing longitudinal sections of the small intestinal villi, to complement the literature on the gastric effects of indomethacin (Robert, 1979). In this ulcerogenic two-dose study, full circumference resin histology sections of mouse duodenum, jejunum and ileum were used to investigate villi and crypts aligned in longitudinal section in non-ulcerated areas. Parameters measured included epithelial and crypt dimensions, cryptal mitotic activity and counts for a range of cryptal and villous cells, as used further in ³Chapter 3 (Sections 3.8, 3.9). The treated intestine showed some cryptal changes and increased mitotic activity. The citations for this paper were technical as well as data-based, including adaptation of its indomethacin protocol (Venkatesh et al, 2014), mention of its reporting of the side effects (Unmack et al, 2001) and of its findings on Paneth cells (Legraverend and Jay, 2013)

This approach contrasted with one where control data were collected for small intestinal villi cut in transverse section and not longitudinally as above, thereby allowing good analysis of axial structures that varied in the apico-basal direction (Abbas et al, 1989), as seen for functional parameters (McElligott et al, 1975, Pothier and Hugon, 1980). This provided baseline images for comparison with resin casting (Chapter 2, Section 2.4) and control information for studying responses to environmental challenge such as radiation (Chapter 3, Section 3.6). The images were from serial sections, alternating study of semithin sections for 'resin histology' light microscopy (LM) with examination of ultrathin sections in a TEM. The data collected were from mouse jejunum for apical, middle and basal thirds for a range of parameters from villous cross-sectional area to microvillous

³ Cross references to other Chapters include information on Chapter and Section.

length. There were several marked changes from base to apex, such as enterocyte widening and microvillous lengthening. Two citations from T. Mayhew's group for this paper included it as one of the references quoted for cellular changes as cells move from the crypt to villous tip (Zoubi et al, 1995a and b).

The next pair of papers illustrated continuing morphological contributions to other groups studying aspects of intestinal diet, including variations in fat or fibre components. In the dietary fat study (Sagher et al, 1991), measurements were made of rat jejunal and ileal villous height and crypt depth in standard wax histology sections. These were compared with the levels of regulatory peptides, to add to the literature on the effects of other interventions, such as starvation (Hopper et al, 1968), and feeding (McManus and Isselbacher, 1970). Maize or olive oil increased the ratios of villous height/crypt depth and also altered the tissue concentrations of somatostatin, neurotensin or substance P. The outcome varied with intestinal site, a topic returned to later (Chapter 3, Section 3.9). The citations for this paper included comments that it had linked relative saturation of fat with villous morphology (Tamura and Suzuki, 1997, Moussavi et al, 2008) and commented on the importance of the ratio of villous length/crypt depth (Peuranen et al, 2004). The second paper in this pair (McCullough, J.S. et al, 1998) took further the interest in the effect of dietary manipulation on intestinal function, this time with a different collaborative group, including R. Goodlad as one of the authors. Exposure to fibre supplementation was one of the investigative tools, another being variation in intestinal microflora in germ-free as well as conventional rats. The paper used resin histology for the study of rat small and large intestine: measurements of several parameters were made, as set out in reports of postirradiation effects (Chapter 3, Section 3.8): data were also collected on crypt branching (Levi et al, 1992), with discussion of how this related to crypt fission. The presence of fibre increased crypt number and branching and also affected goblet cells. Citations for this paper included references to technical aspects, with the advantages pointed out of being able to carry out additional analyses on fixed tissues stored for later use (Goodlad, 2017). Citations based on data reporting included reference to various aspects of gut biology: examples were Montagne et al (2003), Olivera et al (2003), Fastinger et al (2008) and Brownleee (2011).

1.3 Context and comment

The theme of this introductory Chapter was centred on the interpretation of information from sectioned material, paving the way for the addition of detailed study of surfaces. The work submitted in this Chapter took place initially in the University of Glasgow and then in Belfast. It covered thirty of the fifty years, the dates of the submissions running from 1967 to 1998.

Output destination was on the whole similar for the two subsets of articles, a mixture of anatomical journals, such as the Journal of Anatomy, and clinical journal or specialist journals, such as Gut.

Collaboration was also important. The addresses of the publications charted its continuing significance, demonstrating contributions to research in University clinical departments and also to output with gastrointestinal and imaging groups elsewhere. Departments or Laboratories involved included: in Glasgow University, Materia Medica, Medical Cardiology, Microbiology, Veterinary Histology and Embryology and also Pathology; in Queen's University Belfast, Child Health, Dental Surgery and Medicine; and elsewhere, Lawrence Berkeley Laboratory (LBL), California, USA, Imperial Cancer Research Fund (ICRF), London, Robert Gordon University, Aberdeen and University College, Dublin, the last-named being the next address of a student after graduation.

The thesis author's roles and contributions were seen from the order of the names on the two sets of four articles. The first four papers in this Chapter, published in the period 1967 to 1975, were the earliest output, taking projects on from the work of colleagues or joining collaborative projects already established by groups in other specialities. These papers were produced as a 'hands-on' electron microscopist, based in the Department of Anatomy of the University of Glasgow, contributing to the maintenance of standards in the TEM Unit there and to the use of TEM images to update collections of illustrations for the study of cells and tissues. The second group of papers showed that the research continued, in the period 1989 to 1998, to produce data from sectioned material. This illustrated the change of roles to supervisor and leader of a group involved in gastrointestinal research and therefore benefiting from collaborations with scientists working in cognate fields.

Links to the next Chapter were clear from the continuing interest in the use of morphology to explore soft tissue structures, especially those of the alimentary system. The title of the thesis, 'Probing tissue surfaces' was selected to reflect the content of the four main Chapters and has been discussed in Chapter 5. Chapter 1's concentration on the study of sectioned material has provided a baseline for the later Chapters.

Preface - extract from Presubmission Abstract (Chapter 2)

Early papers on scanning electron microscopy covered small intestinal villous surfaces, among others, and early reviews described the technique and its applications. Methodological developments included: correlation of surface and sectioned images having investigative and clinical relevance; histochemical or immuno- 'staining'; low temperature X-ray microanalysis, surface cleaning, 360° detail and microvascular casting. Papers on scanning electron microscopy of the alimentary system often used it as one of several tools, as did some reviews and chapters that included such images, referred to as 'topographical histology'.

2.1 Early scanning electron microscopy and links to Chapter 1

The work presented here, not necessarily in chronological order, was carried out in the early days of the application of scanning electron microscopy (SEM) to biomedical sciences.

The instrument used initially, producing a publication in 1968, was a Cambridge Scientific Instrument's 'Stereoscan' in the BioEngineering Unit at the University of Strathclyde, sited at Canniesburn Hospital. Other groups were publishing in related fields, including Sikorski (1960), Jaques et al (1965), Hayes et al (1966), Clarke and Salisbury (1967), Boyde and Lester (1968) and Echlin (1968). However, the SEMs in Glasgow, in the Universities of Strathclyde and then Glasgow, were early in concentrating on soft tissues. On the establishment of the first instrument, advice was sought on best practice through visits to the Sikorski and Boyde laboratories by the author and a technician from the group respectively.

The early interest in the ultrastructure of cells of the small intestine seen in Chapter 1 (Carr, 1967, Carr and Whur, 1968) was extended to take advantage of the new SEM approach by applying it to the study of villous surfaces. Although there had been earlier illustrations of the intestinal surface (Jaques et al, 1965), the first paper in this Chapter (Carr and Toner, 1968) was one of the earliest publications concentrating on it, in a letter to Lancet. This demonstrated the use of SEM to give more detail of 3 dimensional structure and epithelial surface detail than had previously been possible with dissecting light microscopy, often in

the study of specimens of clinical interest (Rubin et al, 1960). Technical aspects of interest included dehydration by a form of air drying using a vacuum desiccator and coating the specimen surface at 45° with carbon-platinum, an adaptation of the 'shadowing' of surface replicas prior to study in a transmission electron microscopy (TEM). The letter commented on the clear imaging of villous shape and cell surfaces and on the possibility that greater detail could be made available during the study of human biopsies. This paper was followed six weeks later by a another letter to Lancet from a different group, using an alternative preparation technique, extending the detail seen to microvilli by vigorous syringing of the mucosal surface and the use of a higher microscope kilovoltage (Swift and Marsh, 1968): this second letter, however, confirmed the general morphology and possible relevance to clinical studies. Although short, the paper by Carr and Toner (1968) was cited by other early publications in key journals (Gastroenterology, Journal of Ultrastructure Research) as one of the sources for images of the surface structure of the small intestine (Balcerzak et al, 1970, Takeuchi and Zeller, 1972, respectively). It was also quoted over a substantial period, for example for the description of villous shape (Sherman et al, 1987).

The Lancet letter was followed quickly by collaborations with colleagues already working with other techniques on alimentary topics that could be explored further by detailed examination of surfaces using the 'new' technique of SEM. One example (Ferguson et al, 1969) was provision of a clear visual record of the three dimensional structure of the connective tissue cores of autolysed villous surfaces from human newborns, confirming the overall appearance seen earlier (Loehry and Creamer, 1966). This illustrated well the development of villi from finger-like to a mixture at birth of finger, tongue and leaf shapes. There were no available citations of interest to the SEM aspects of this paper. Another use of the technique was its application to experimental situations at natural surfaces, such as secretagogue-induced duodenal ulceration (Carr et al, 1979a). Here the SEM was used for the kind of investigation previously carried out by stereomicroscopy (Giampaolo et al, 1978), charting the details of healing from one to ten days after 24 hours exposure to pentagastrin and carbachol. Technical features of note included the use of: large samples, making full use of the SEM's capacious specimen stage, situated below all the demagnifying lenses and therefore lacking the size constraints of the TEM; and montage techniques to record information about the surface of these samples, combining survey and detail in the final composite images. Eleven years on from the initial descriptions of villous surfaces in 1968, critical point drying and gold sputter coating were by then available as standard preparative techniques. SEM showed: variation in the extent of healing; sequential mucosal damage and villous abnormalities at the ulcer margin; the honeycomb surface appearance of the immature re-epithelialised surface; and irregular villi. Some signs of gastric metaplasia were also seen. This paper also had no citations relevant to the use of SEM.

The next four papers illustrated the use of the SEM for a range of other applications. The first paper (Carr, 1970c), published two years after the first SEM paper mentioned above and using similar early preparative techniques, explored the three dimensional structure of human skin. This built on work already done on rodent wound healing by Forrester et al (1969), where the authors included T. Hayes, a future collaborator (Chapters 1, 2, 3). Rather than studying naturally occurring surfaces, as in small intestine, this study described the exposed surfaces of dewaxed sections 25 to 35μ m thick. This revealed the three dimensional arrangement of dermal connective tissue fibres: faint cross-banding identified the larger collagen bundles, with elastin as the thinner coiled structures. The advantage of having previous knowledge of how such structures appeared when studied by light microscopy and TEM gave rise to an interest in correlative approaches, dealt with later (⁴Section 2.3). The citation record included reference to the paper in the description of skin (Wei et al, 1989), of dermis as a whole, with its collagen fibres (Tsuji et al, 1979) or of bundles of these fibres with intervening fibrils (Li, 2015): the continuing relevance of the paper was illustrated by the fact that the citations ran to 2015.

The paper by Carr and Carr (1970) also moved away from the study of natural, easily accessed surfaces, this time by examining isolated mouse peritoneal macrophages on a substrate such as glass, following up a suggestion that stimulation could alter their speed of movement (Dannenberg et al, 1963). This article was, like the skin paper, published only two years after the Lancet letter and was still using unsophisticated techniques for drying the samples, leading to loss of surface detail in the images. The stages of settling on glass, were, however, reported, as was the quicker full extension of cells from mice exposed to peritoneal stimulation. Citations included references to cells spreading or their cellular surface features (Witkowski and Brighton, 1971, Mantovani et al, 1972, Rajaraman et al, 1974). The most interesting citations were to the use of the observations as a pointer to the Kupffer cells of the liver being of the macrophage line, in reports from Niigata, Japan and Rome, Italy respectively, (Muto, 1975, Motta, 1975).

⁴ Cross references within a Chapter only include the Section number.

The next two papers (Cook et al, 1979, Al-Jaff et al, 1982) moved on the time-frame of the work by ten years and illustrated further the way in which early scanning electron microscopists contributed to collaborative networks, in this case between the Departments of Agricultural Chemistry and Anatomy at the University of Glasgow, where an instrument had by now been installed. The research produced two publications on the effects of the herbicide aminotriazole on bracken: its ability to thrive on open terrain had been linked to a decrease in the number of stomata (Bright, 1928). Both papers have been included here, to illustrate the continuing use of SEM across such a collaborative grouping before there was a more widespread availability of such instruments. In Cook et al (1979), bracken fronds growing in three different shade conditions were studied by light microscopy of plastic sections as well as by SEM of upper and lower surfaces. Herbicide uptake was also assessed in the laboratory. Shaded fronds took up less herbicide and their lower surfaces had different patterns of distribution of hairs and stomata. Further work (Al-Jaff et al, 1982) extended the range of growing sites to four, at each comparing shaded and exposed bracken. In addition to the differences observed earlier, the shaded frond surfaces were also more corrugated, restricting herbicide spreading. Citations included references to environmental conditions and other factors affecting penetration (McPhail and Duncan, 1981, Alonso-Amelot and Rodulfo-Baechler, 1996, Chachalis et al, 2001).

Several of these early papers showed the importance of correlating SEM results with other data collected in parallel: this is expanded on below (Section 2.3).

2.2. Early reviews of biological scanning electron microscopy

Reviews published between 1971 and 1974 put the work in context with other early SEM reports (Section 2.1). The first (Carr, 1971), published in a mainstream biological review series, began with correlative images of biological surfaces, using light and transmission microscopy of sectioned material: surfaces were illustrated by dissecting microscopy, SEM and TEM of both replicas and freeze-dried samples. Thereafter, the review covered image formation and instruments; preservation and preparation of specimens; stereopairs and different scanning modes; and images from a range of biological specialties, some from other microscopists, showing the connections developing world-wide. The conclusion that this versatile instrument would have a useful future in biological research has been borne out by events. With respect to citations, there were several mentions of this review as one of a small number describing the emergence of SEM as a biological tool (Taylor and Anderson, 1972, Sanders and Singal, 1973). An important citation came in a Special Invited

Review (Hayes, 1974), published in the Journal of Microscopy by T. Hayes, himself one of the early leaders in the SEM field (Section 2.1, above), working at the University of California, Berkeley and a future collaborator. He introduced the subject by noting the developments at Cambridge (Oatley et al, 1965) and describing the SEM as useful in physical (Thornton, 1968) and biological sciences (Carr, 1971). Another group (Martin et al, 1975) advised that interpretation of SEM images would be helped by understanding how they were formed, citing two authors specialising in instrumentation (Thornton, 1965 and Crewe, 1971) and also the Carr review (1971). This indicated that the review was seen as a source for technique and instrumentation as well as biological applications.

In the same year, SEM images of intestinal surfaces also contributed, with text, to an ultrastructural atlas and review of the digestive system (Toner et al, 1971), including comments on the possible variations of human villous shape from finger-shaped through leaf to ridges and convoluted structures. Non-standard images were included, such as intervillous crypt openings, villous creases, sites of enterocyte junctional complexes, the surface pockmarks of goblet cells and the autolysed villous connective tissue framework. The mucosa of coeliac disease was also illustrated, showing the lack of villi and the raised mucosal rings around some crypt mouths. Scopus did not list this book, but it was Google Scholar's top listing, with 135 citations, ranging from 1974 to 2015 (see also Chapter 5, Section 5.2.2).

The last of the three early reviews, (Carr et al, 1974) concentrated entirely on SEM surface images, briefly describing not only small intestine, but also the flat mucosal surfaces of oesophagus, the gastric pits of the stomach and the opening of colonic crypts. There was a mention of intestinal surfaces in coeliac disease and the effects of irradiation, dealt with again in Chapter 3. The only citation of interest was in a paper on oesophageal surface morphology at the junction of squamous and Barrett's epithelium (Shields et al, 1993), where the cells were said to resemble some features reported in the literature, including those in this short review.

2.3 Correlation of SEM images with those from established techniques

It became important to identify surface features more accurately, to 'validate' SEM image interpretation (Brummer et al, 1975, Carter, 1980). Correlation began by simply using different microscopy techniques on other, similar specimens, but progressed to postembedding screened SEM samples for light microscopy resin histology and for TEM. This technique has been illustrated here through the study of one intestinal villus in these three ways (Carr et al, 1981a). A swollen 'bleb' on an irradiated villous surface was confirmed not to be an artefactual deposit of the glue used to affix the sample to the metal stub: it was instead a multinucleate giant epithelial cell (Chapter 3, Sections 3.1, 3.6). Post-SEM embedding was also used to illustrate the surface and internal features of large bowel biopsies (Carr et al, 1981b). Orientation identifiers were used and plastic sections depolymerised to permit the use of stains normally restricted to wax histology sections, including azure blue, haematoxylin and eosin, periodic acid-Schiff (PAS) or Mallory (Laschi and Govoni, 1978). The often-used secondary fixation with osmium tetroxide was avoided, since it interfered with these staining procedures. It was concluded that in a few cases, here Hirschsprung's disease and post-irradiation proctitis, the preliminary SEM screening provided unexpected information that might be useful. There were no available citations of interest to the correlative technique used in either of these papers.

Another twist to the correlative studies involved light microscopy as the first technique, with SEM thereafter. Two different approaches were developed. The first was used for cells from body cavity fluids (Saleh et al, 1982), identified in light microscopy with localising micro-coverslips. These then acted as finder grids to produce SEM images with greater surface detail, giving selected areas of a more useful size than previously (Domagala et al, 1979). The few citations were technical and from papers specialising in microscopy or optics (Wouters, 1987, Sandoz and Jacquot, 2011), an appropriate readership. The second approach (Carr et al, 1984a) also changed the order of the techniques used from that of post-scanning light microscopy (Fenger and Knoth, 1981). Features of interest were identified by light microscopy in wax histology sections from gastrointestinal Pathology files: the whole block was then dewaxed for SEM surface examination. While the quality of the surface image was good, no case was made for the addition of SEM to the routine diagnostic procedures. One further feature was the paper's publication in Scanning Electron Microscopy after oral presentation at the relevant annual international meeting in the USA. As in the standard journal reviewing procedure, changes to text or figures were requested before publication: an example in this paper was the addition of images from SEM backscattered (primary) or transmission detectors. An additional aspect, however, of this Journal's publication method was the inclusion at the end of each paper of the authors' responses to points raised by the reviewers, whose names were usually given, such as, in this case, Shirley Siew, an authority in the field of gastroenterology. Questions covered possible techniques for mucus removal, details of preparative techniques for SEM, nonstandard detectors in the SEM and the use of the terms 'investigative' or 'diagnostic' for the application to pathology of SEM or other tools. This paper was cited as part of an interesting discussion of techniques, where Roessler et al (1991) included it in a list of papers using what they called 'intermicroscopic correlation', defined as 'the sequential examination of a specimen by different microscopic techniques'.

2.4 Development of scanning electron microscopy techniques

Following on from the paper above, where an image was shown from a backscattered (primary) electron detector (⁵Carr et al, 1984a, Section 2.3) and from a report of the use of silver nitrate to produce contrast in SEM images (Geissinger, 1972), a paper by Carr and McGadey (1974) adapted enzyme histochemistry techniques. This produced localised precipitates of substances that gave a high SEM signal, in this case of lead in sections of pancreas, epididymis and cerebellum. The results marked the development of SEM towards the asking of more precise morphological questions. There were two citations of interest, one from P. Nakane, well known for his work on electron microscope (EM) immunocytochemistry, quoting the paper as a source for the use of enzymes as smaller markers (Nakane and Hartman, 1980). A second citation, from a paper on atomic number contrast, quoted it as having used variable accelerating voltages and primary or secondary electron signals (Bowen et al, 1983). Another labelling protocol (Hodges et al, 1985) used surface conconavalin-A and gold on control and irradiated mouse bladder: this was done in collaboration with G. Hodges, then at Imperial Cancer Research Fund (ICRF) laboratories in London and involved in the development of biomedical SEM and cell surface markers (Hodges et al, 1982). The gold/con-A labelling in control samples was present at the regions of the aysymmetric unit cell membrane then known as interplaque and later as hinge (Liang et al, 1999), as opposed to the intervening plaque sites. Data on changes in the epithelium and binding patterns after irradiation have been included in Chapter 3 (Section 3.3). The few available citations to the immunolabelling paper made no particular reference to the technique of SEM labelling: references to the effects of irradiation have been given in Chapter 3 (Section 3.3).

The next group of five papers reported on the production of chemical information from Xray microanalysis (XRMA) of gastrointestinal samples prepared by freeze-hydration: this

⁵ Cross reference to specific papers in the accompanying material give the reference first, then Chapter and/or Section information.
protected them from chemical exposure and left tissue fluid *in situ* (Echlin, 1978). It also minimally disrupted the relationship between luminal contents and the apical surfaces of villous epithelium, a theme addressed again later in this Section through reports on specimen cleaning and surface exudate (this Section, below and Section 2.5 respectively). The five papers came from work done during a series of laboratory visits to T. Hayes at the Donner and Lawrence Berkeley Laboratories, this being one of only a small number of groups world-wide with a low temperature X-ray SEM microanalysis capability. This collaboration stemmed from earlier citation of each other's work, such as to the review above (Carr, 1971, Section 2.2) and from meeting at the annual SEM conferences. It led to the five publications between 1983 and 1987 on low temperature SEM, described here, and to another on related material (Abbas et al, 1989, Chapter 1, Section 1.2). It also produced other papers between 1987 and 1994, on the effects on small intestine of radiation, including Berkeley heavy ion beams, (Chapter 3, Sections 3.4, 3.6, 3.8).

The first of these five papers, Carr et al (1983a) described specimen preparation in high vacuum and at low temperatures in an AMray Biochamber (Pawley and Norton, 1978), followed by screening in an AMray SEM with a cold stage. Tissue compartments, cell types and some internal detail were identifiable. Specimens were subsequently freeze-dried in the microscope and then prepared for routine surface SEM, resin histology and TEM. The second and third papers (Carr et al, 1983b, Carr et al, 1984b) compared frozen-hydrated etched images with those from plasma etching of resin sections (Humphreys and Henk, 1979), using either secondary or backscatter (primary) mode and allowing a range of magnifications from survey work to imaging of organelles such as mitochondria. The third paper (Carr et al, 1984b) also contained pilot X-ray microanalysis data of chlorine levels in stomach contents and adjacent surface cells and had input from Glasgow University Departments of Medicine and Chemistry and from Hitachi Limited, for high resolution field emission SEM images. The fourth and fifth papers (Carr et al, 1986a, Carr et al, 1987a), had contributions from a scientist with an interest in the handling of large data sets, from the Glasgow University Department of Physics: these papers described the X-ray microanalysis procedures (Fuchs and Fuchs, 1980, Echlin et al, 1981) and also the resulting data. The levels of sodium, sulphur, chlorine and calcium increased from cell apex to central luminal contents, while the differences in sulphur in particular were greater between distal (upper) and proximal (lower) villous regions than between apical, middle and basal parts of a cell. The first paper (Carr et al, 1983a) was the only one of the five with available citations in the field: it was quoted as a source for low temperature SEM in biology (Inoué and Koike,

1989), during a comparison of techniques (Jongebloed et al, 1999) and in the study of mucus (Lumsden et al, 1994).

The next paper, Dickson et al (1989) also reported on the interface between the gastrointestinal mucosal interface and the luminal contents, concentrating on removal of exudate and debris (Goldstein et al, 1981). This allowed comparison of rat stomach mucosal surfaces before and after cleaning. The low citation rate was linked to the fact that most groups removed the exudate before SEM examination. The subsequent 'multiview' paper, also relating to mucosal surfaces, used the boundary with the stomach as a reference line and recorded the four main 'faces' of villi (Carr et al, 1989), taking further the concept of stereo-pair imaging (Wells, 1960, Boyde, 1974). Different types of surface creases were described, showing that, in control samples, they were mostly short and ran around the villous diameter. The lack of citations here was probably because stereopairs were less used than they should have been and most groups were therefore unlikely to use a 360° analysis.

The final paper in this Section, Abbas et al (1990a) changed the focus from the mucosal/luminal interface to the internal vascular trees, exposed by microvascular casting (Murakami, 1971): this involved perfusion of resin monomer and then digestion of non-vascular tissues from the polymerised cast. The control villous arteriolar supply divided apically into two to form an inverted U-shape from which came a capillary plexus and then venules, running to the submucosal veins: these observations were used as a baseline for the changes produced by irradiation (Chapter 3, Section 3.6). This paper was better cited (from 1998 to 2012) than the other papers on technical developments in SEM. Jahnson et al (1998) commented on the 'excellent perfusion quality' and Burrell et al, (2012) cited it as one of their sources for the microcasting technique.

2.5 Scanning around the alimentary system

An early article, (Carr et al, 1974, Section 2.2) had briefly reviewed the information that SEM could provide on alimentary tract mucosal surfaces. More detail, almost all on mucosal surfaces, has been included in this Section, often including correlative information from other techniques.

The first paper (Nunn et al, 1990) analysed the SEM appearance of luminal 'exudate', giving different information from that provided by low temperature analysis (Carr et al, 1983a, 1986a, 1987a, Section 2.4). There had already been some reports on exudate in the upper

tract (Siew and Goldstein, 1981), stomach (⁶Carr et al, 1974) and children's small intestine (Poley, 1983). The paper by Nunn et al (1990) scored mucus, micro-organisms/bacteria and cells/debris in rabbit exudate on uncleaned surfaces. It reported that there was more variation down the tract than across animals. All three components were found on the stomach surface, which also had the greatest amount. Features elsewhere included debris at the oesophageal/cardiac junction and caecum, but micro-organisms at upper oesophagus and transverse colon. Although the paper was not highly cited, no doubt, as for Dickson et al (1989), because exudate was usually removed, a review of mucus during Helicobacter pylori infection reproduced one of its stomach images (Bansil et al, 2013).

The next group of three papers described the microridges of the epithelium of the upper part of the tract (Laferla et al, 1988, Shasha'a et al, 1993, Johnston et al, 1996). This stratified squamous non-keratinising epithelium had already been described (Ackerman et al, 1976, Siew and Goldstein, 1981), but there was not a large literature on microridges. The first paper (Laferla et al, 1988) used the surface characteristics of human oesophagus to categorise it as either a form of squamous epithelium, with typical or atypical microridge patterns, or alternatively as non-squamous, more like gastric mucosa. The squamous and non-squamous groups were from patients respectively without or with oesophageal symptoms. The second paper on microridges (Shasha'a et al, 1993) compared the surface features of human lower oesophageal biopsies with those from rabbits of different ages. Surfaces were cleaned as above if necessary (Dickson et al, 1989, Section 2.4), stereopairs (Boyde, 1974) confirmed that microridges were indeed raised above the cell surface and the percentage taken up by microridges was calculated. During development, surfaces of rabbit oesophagus had an increasing percentage area of microridges. All oesophageal regions of older rabbits resembled the typical squamous group of human biopsies. Points raised by reviewers (G. M. Roomans, W.H. Wilborn), included the distinction between microvilli and microridges, rabbit oesophagus as a model and possible identification of gland openings. The third paper on microridges (Johnston et al, 1996) looked for a link between them and the features of gastro-oesophageal reflux disease (GORD, Branicki et al, 1984, Johnsson et al, 1987). No correlation was found and it was concluded that the role of the SEM was more qualitative. All three publications were poorly cited, reflecting the small literature, although

⁶ Chapter and/or Section information is only given for the first mention of a paper in any Section.

the 1988 paper was quoted for the presence of microridges, usually in parallel rows (Raymond et al, 1991).

Some work had also been done on stomach (Carr et al, 1984b, Carr et al, 1989, Dickson et al, 1989, all Section 2.4, Nunn et al, 1990, this Section), but the only systematic study led to no publications. The next four papers described the mucosal surface of small intestine, three after experimental intervention and one on childhood diarrhoea.

The first experimental procedure (Ferguson et al, 1978) involved allograft rejection of mouse fetal small intestine from its transplant site under adult kidney capsule: this was regarded as a model of delayed hypersensitivity (Ferguson and Parrott, 1973). The image quality in this early paper was limited by the air drying technique. The SEM images showed that isografts contained well-formed villi, while allografts showed great surface variation, including stunted villi or a flattened mucosa. The allograft damage profile was more like coeliac disease or parasitic infestation than a post-irradiation surface. With respect to citations, none of those available mentioned SEM.

The second experimental paper on small intestine (Ettarh and Carr, 1993) was the group's first on indomethacin-induced changes (Kent et al, 1969, Satoh et al, 1981). It followed on from an earlier paper on ulcerogenesis, (Carr et al, 1979a, Section 2.1) and from work on radiation-induced ulcers (Chapter 3, Section 3.5). It predated the paper included in Chapter 1 to illustrate the importance of plane of section (Ettarh and Carr, 1996a, Section 1.2) and showed early 'butterfly' lesions at the gastroduodenal junction, with adjacent villous distortion. Oval or elliptical jejunal and ileal ulcers were also reported, with villous separation and irregular creasing. Re-epithelialisation occurred by 24hours after treatment. This was the highest cited paper in Scopus in Chapter 2. It was used as a protocol source by several groups, including Playford et al (1996, writing with R. Goodlad and N. Wright), Croci et al (2003) and Xue et al (2009). It was cited for indomethacin having produced gastric ulceration (Langenbach et al, 1995). Faucheron (1999) quoted it, in a 'review in depth', for greatest damage being at 20hrs and for being 'of special interest' in a list of annotated references, where it was noted as 'demonstrating a systemic effect of NSAIDs on gastrointestinal tract'.

The third experimental paper, on streptozotocin-induced diabetes in mice (Ettarh and Carr, 1997), was a fore-runner to the study of radiation responses in subjects suffering from an underlying pathology (Chapter 3, Section 3.9): changes had previously been described in several morphological parameters (Mayhew and Carson, 1989, Pinto et al, 1995). Jejunal

and ileal villi were distorted, with attached filamentous microorganisms. Several histologybased quantitative data sets were given, using an approach developed to report multiparameter radiation-induced changes (Chapter 3, Section 3.8). This paper, with SEM as one of the techniques, was cited for weight increase (Osborne et al, 2000), small intestinal elongation (Domènech et al, 2011), structural and functional changes to the small intestine (Durmus-Altun et al, 2011) and responses in Paneth cells (Porter et al, 2002, Min et al, 2014).

The fourth paper on small intestine (McClean et al, 1996) described the clinical features of jejunal surfaces from otherwise healthy children with symptoms of diarrhoea (Davidson and Wasserman, 1966, Thomas et al, 1992). In addition to recording villous shape, an estimate was made of mucus and debris coverage. Overall, the SEM data were regarded as useful. Coeliac disease and giardiasis were identifiable and microorganisms were sometimes associated with glycocalyx loss and microvillous clumping. For citations, it was interesting that a paper on villi in childhood diarrhoea, with authors from Departments of Child Health and Pathology as well as Anatomy, attracted interest less in clinical than in modelling papers (Bhuiyan and Poston Sr, 2005, Willman et al, 2009), albeit quoting biological findings, on mucus or villous numbers respectively.

A paper on rat large intestine, (Galloway et al, 1987) highlighted environmental factors (Wynder and Shigematsu, 1967, Berg and Howell, 1974). It included SEM images of mucosa following carcinogenesis and variations in diet, a topic that continued to be of interest (Sagher et al, 1991, McCullough, J.S. et al, 1998, both Chapter 1, Section 1.2). Histology showed no crypt hyperplasia. The most marked changes in crypt unit integrity were for the high fat low fibre diet, using an SEM scoring method constructed for irradiated small intestine (Chapter 3, Section 3.5). With respect to citations, it formed part of a debate on the details of how cell proliferation was affected by the type of dietary fibre ingested (Zhang et al, 1998) and was quoted for the relationship between proliferation and colon cancer risk (Robblee et al, 1989). It was also cited for the similarity between human and animal findings on the effects of fat and fibre (Freudenhieim et al, 1990) and for its report on surface responses in the high risk diet containing high fat/low fibre (Ma et al, 1996).

The next paper, by Ettarh and Carr (1996b), although a study of small intestine, described serosal and not mucosal surfaces, concentrating on mesothelium, important during peritoneal dialysis (Raftery et al, 1989). It had previously been studied with electron microscopy (Odor, 1954, Baradi and Hope, 1964, Abu-Hijleh et al. 1994). The presence of

straggly serosal microvilli was confirmed. Two other features were reported, one a serosal exudate, the other mesenteric stomata joining the two mesothelial aspects, but without evidence of connection to the lymphatic system, as seen at the diaphragmatic peritoneum (Tsilibary and Wissig, 1977, Fukuo et al, 1990). This paper was cited for the report of mesenteric peritoneal stomata (Wassilev et al (1998). It was also quoted for the thickness of the layer covering the mesothelial surface, examinable through peritoneal 'washings', and also for its report that this was only observed if osmication were omitted (Flessner et al, 2001). Reference was also made to it for the biosynthetic nature of mesothelial cells (Bird, 2004, in a review), as shown by its description of micropinocytotic vesicles. Gilloteaux et al (2013) cited it for the similarity of the mesothelial surface in the nude mice they used and that found in other animals, with reference to the spread of cancer.

The last paper in this Section (Simpson et al, 1983) reported on cat pancreatic duct after bile salt perfusion, exploring further the mucosal barrier (Reber et al, 1979): an interesting feature of the surface was the presence of scattered single cilia. It was quoted for its reporting on the duct epithelium (Arendt, 1991). It was of interest that the citing author commented on the relevance of the epithelial tight junctions, given their possible plasticity during microparticle uptake (Chapter 4, especially Section 4.8.4).

2.6 Later reviews of biological scanning electron microscopy

The first review (Carr, 1980) was a description of macrophage surfaces, in a book on the reticuloendothelial system. It covered: preparative techniques; phagocytic cells *per se* and in organ systems; phagocytosis; and pathophysiology. The macrophage family members, found in a wide range of body sites, were described as rounder when resting, flatter when more mobile; and heavily ruffled when busy with phagocytosis, but larger and smoother after engulfing extraneous material. Stimulated cells spread faster than resting cells and surface receptors were found more often at peripheral veils and were less common during spreading. This chapter on macrophages was included to show involvement in the cell culture field and the range of scientists contributing illustrations.

The second paper in this Section (Carr et al, 1982a), was an invited review of scanning electron microscopy in the journal Histopathology. It provided an outline of different aspects of SEM, such as instrumentation, sample preparation and examples of the types of images produced with standard detectors: all the illustrations were in-house. It used the term 'topographical histology' for the study of cell and tissue surfaces, to push towards more

rigorous image descriptions. It concluded that SEM was part of the range of available investigative approaches, rather than a diagnostic tool.

Further integration of SEM imaging was found in the descriptive text and illustrations of a range of surfaces in the 3rd edition of a book on cell structure (Carr and Toner, 1982): such images had not been available for the first two editions (Toner and Carr, 1968, 1971). The material from it presented in the thesis has concentrated on SEM, including the preface, instrumentation, critical examination of electron micrographs and a number of figures including a few correlative images. Some of the pictures were drawn from the wider community of scanning electron microscopists, but many were in-house. It has been included to point to the work done in the days of expanding TEM (Chapter 1) and developing SEM (this Chapter). This has provided contributions to the body of knowledge underpinning much modern biomedical research and education, despite predictions in the early 1960s that electron microscopy would never have anything to contribute to the teaching of histology.

Finally, the integration of surface and internal structure of tissues was also seen in an account of the morphology of the intestinal mucosa (Carr and Toner, 1984), the first chapter in a volume of the Handbook of Experimental Pharmacology. SEM Figures illustrated in 'three dimensional' format the structures specialised for the main intestinal function of absorption or reabsorption at the luminal/mucosal interface. The chapter also dealt with other functions, including defence, mechanical propulsion and neural or endocrine control.

Citation ranking for these four submissions did not reflect the influence of such publications. However, both the Pharmacology chapter (Carr and Toner, 1984) and the book 'Cell Structure' (Carr and Toner, 1982), listed only in Google Scholar, were placed at similar levels of interest to some of the articles in Chapters 2 to 4. In particular, the book had citations up to 2015, covering a range of topics, most published in journals rather than chapters or books.

2.7 Context and comment

The theme of Chapter 2 built on the interest in learning about tissue and cell relationships through the study of sectioned material, described in Chapter 1, by exploring surface structure and its underlying tissues. This was done through the development of SEM to take its place as one of a battery of morphological approaches used in the scientific study of the interface between environmental challenge and biological cells and tissues. The

submissions were produced from work carried out originally in the Universities of Strathclyde and Glasgow. The move to Belfast brought access to a more sophisticated instrument which allowed, for example, the use of non-standard detectors and image analysis: the facilities of LBL were also used during laboratory visits. The work covered slightly under thirty years, similar to Chapter 1, with article publication dates spanning the period from 1968 to 1997.

With respect to destination of articles, there was continuing use of anatomical or more applied journals and the first papers using SEM told their story in Lancet, the Journal of Pathology and the Scandinavian Journal of Gastroenterology, rather than in the 'microscope' journals. Reviews, books and book chapters showed the broad interest in the developing applications of SEM.

Collaborations with other Institutions, Departments or individuals, were, as for the work reported in Chapter 1, important in maximising the quality and breadth of the output: some of these have already been named above. In general, collaborating groups while at Strathclyde and Glasgow Universities were found in other Departments and Centres, including: Agricultural Chemistry, Bacteriology and Immunology, Chemistry, Cytology, Gastrointestinal Centre (Southern General Hospital), Medicine, Pathology; Pharmacology, Radiobiology Research Group, Glasgow Institute of Radiotherapeutics and Oncology (Belvidere Hospital), Royal Hospital for Sick Children (Yorkhill) and Surgery. Groups in Queen's University Belfast included Child Health, Dental Surgery, Medicine and Pathology.

Collaborations external to the 'home' Institution included the Centre for Bioengineering, University of Washington, Seattle, the Centre for Applied Microbiology and Research, PHLS, Porton Down, the Department of Anatomy, University of Sheffield, the Imperial Cancer Research Fund (ICRF), London, the Lawrence Berkeley Laboratory (LBL), California, USA and the MRC Cyclotron Unit, Hammersmith Hospital, London. Some colleagues or graduate students moved elsewhere during the work and continued to contribute, particularly to the writing of the papers: centres involved included Calgary, Alberta, Canada, Anatomy Department, the University College, Dublin, the Gastrointestinal Unit (Western General Hospital) University of Edinburgh, King Saud University, Riyadh, Saudia Arabia and the University of Basrah, Iraq.

The changing roles and contributions were again obvious from the authorship order, beginning with 'hands-on' scanning electron microscopy, addressing early technical and

interpretative questions, in laboratories in the UK and the USA, but moving later into involvement as supervisor, group leader, Head of Department or Director of School, the latter two applying during the last ten years of the work in this Chapter. This still involved a central role in the work, from protocol design to publication, but with students or colleagues often taking the lead authorship position.

Links between chapters were highlighted by the last submission (Carr and Toner, 1984), which set the scene for the next Chapter, where radiation effects have been described more for intestine than for other organs. The increasing use of correlative techniques with the SEM in this Chapter led on to its application to explore radiation-induced changes, at surfaces and tissues deep to them, dealt with in Chapter 3.

Preface - extract from Presubmission Abstract (Chapter 3)

Early papers on scanning electron microscopy of irradiated rodent small intestine revealed villous collapse. This was dependent on radiation modality and dose, but was not always to the same extent as the changes in counts per circumference of proliferative crypts, the method then used for estimating intestinal damage. Early reviews came from annual international workshops on radiation-induced 'topographical' and subcellular changes, relevant to radiotherapy side effects and space travel. Further papers recorded morphological effects in other organs, villous changes over large samples and links between cryptal abnormalities and radiation conditions. Modalities included γ - and X- rays, electrons, neutrons and heavy ions. The importance of non-proliferative responses was highlighted by the shorter time-frame of damage and recovery after hyperthermia and by comparison of the effects of irradiation and pharmacological agents. The progressive collapse of villous surface shape was charted using a scale linked to published histometric measurements of section area: this recorded 'direct' intestinal radiation effects and 'distant' abscopal 'ulceration' after thoracic irradiation.

Correlative work showed that surface responses were accompanied by subsurface changes, in stromal, epithelial and neuromuscular components. Numerical data for changes in histological parameters per intestinal circumference produced a single number for morphological effects: subcellular information could also be included. The outcomes were plotted after fixed doses or times and were dependent on radiation dose, time and modality. Individual changes were also tabulated, revealing: possible adaptive responses after protracted irradiation; the lack of additive effects for irradiation and experimentally induced diabetes; and variations in control parameter numbers at different sites proximodistally. Reviews highlighted the individual responses linked to radiation-induced physiological symptoms and topics needing further attention.

3.1 Villous radiation-induced damage and links to Chapter 2

Leading on directly from Chapter 2, the work in this Chapter has been arranged to describe the surface changes in small intestinal villous structure after external irradiation and how they related to the underlying tissue responses. The results of other radiation or mechanistic studies have also been included. As before, the accompanying publications have not necessarily been presented in exact chronological order, but rather setting out optimally the developing theme.

The introductory paper on scanning electron microscopy (SEM) of radiation-induced villous shape changes (Carr and Toner, 1972) appeared only four years after the first paper on the SEM of their basic shape (Carr and Toner, 1968). Following this, links were made with radiation physicists, radiobiologists or oncologists, to ensure good protocol construction of the treatments and clinical relevance to patient care. Thereafter, as space radiobiology became important, it was fortunate that one of the collaborating groups had access to heavy ion facilities, important for the work on the possible effects of galactic cosmic rays.

The addresses used for this work included Anatomy, University of Glasgow, Anatomy (latterly the School of Biomedical Science), Queens, Belfast and The MRC Radiation and Genome Stability Unit, Harwell.

As set out above, the first paper in this Chapter, Carr and Toner (1972) followed on from Chapter 2, listing the flurry of papers from 1968 onwards on the insight given by SEM into control rodent small intestinal villous surface structure and normal and coeliac human samples. This was done either by initial short reports (Carr and Toner, 1968, Chapter 2, Section 2.1, Swift and Marsh, 1968) or by more lengthy descriptions (Demling et al, 1969, Marsh and Swift, 1969, Balcerzak et al, 1970, also Toner et al, 1971, Chapter 2, Section 2.2, Carr and Toner, 1984, Chapter 2, Section 2.6). Although there had been earlier descriptions of intestinal radiation damage (Wierneik, 1966a, b), the paper by Carr and Toner (1972) was possibly the first account using SEM to provide three dimensional information on the collapse of villous shape after external irradiation. It described the effects on mouse small intestine at up to 100 hours after exposure to whole body cobalt gamma rays at supra-lethal doses, 1500 to 2500rads. Contrast enhancers such as ruthenium red were applied to improve the coating (Luft, 1966), limiting surface charging and optimising image quality. There was little radiation-induced change within the first 48 hours, but thereafter the villi appeared sequentially clumped and less erect, then shorter and more conical with disorganised creasing: finally, after 90 to 100 hours, the villi were grossly abnormal, appearing as stunted projections. The irradiated surfaces were different from the flat mucosa and prominent crypt mouths seen in human coeliac disease (Toner et al, 1971). At the latter stages, the surfaces had flagellate parasitic infestations. This paper (Carr and Toner, 1972) was followed shortly afterwards by an account, from one of the USA radiobiology laboratories specialising in gut damage, of radiation effects on rat small intestine up to 12 days after 1000rads of gamma radiation (Anderson and Withers, 1973). With respect to available citations, it was quoted for abnormally sized and shaped villi, obvious structural radiation-induced changes and intestinal damage (Brady and Hayton, 1977a, b, c). Z. Somosy, himself a productive author in the field, cited it as one of a dozen of the group's papers on irradiation-induced changes in villous structure and related changes, in this case commenting on the paper's report of 3 to 7 days as the time of maximum damage post-irradiation (Somosy et al, 2002).

The next three papers, in collaboration with R. Hamlet and T. Nias, then at the Glasgow Institute of Radiotherapeutics and Oncology at Belvidere Hospital, compared the effects on villous structure: of two radiation qualities with fixed time/variable dose or variable time/fixed dose; and of one radiation quality with single or fractionated doses. The radiation qualities had different Linear Energy Transfer (LET) values, low for gamma or X-rays and higher for neutrons, where the clinical interest was to understand the side effects of this comparatively new therapeutic tool. Villous shape and the number of microcolonies (crypts)/circumference were compared (Withers and Elkind, 1970).

In the first of these papers, on dose variation (Hamlet et al, 1976), published in a UK radiology journal, mouse small intestine was collected 3.5 days after whole body irradiation with either ⁶⁰Co gamma rays or D-T neutrons, at doses producing equivalent crypt count damage (Withers et al, 1970), also related to cryptal changes (Hendry and Potten, 1974). Partly in response to the need for standardisation of technique (Anderson and Withers, 1973), it was confirmed that intestinal distention affected the outcome, but did not mimic radiation damage. The usually sequential stages in radiation-induced surface villous damage seen with SEM were defined qualitatively as lateral, vertical, conical and rudimentary collapse. In general, gamma irradiation produced bent, laterally collapsed villi, whereas neutron irradiated villi were more conical. Such changes were seen before the shoulder of the crypt counting curve and they increased with dose. SEM, therefore,

provided a more sensitive reflection of radiation damage, which differed with the type of radiation. This paper had a similar citation history to the original radiation paper (Carr and Toner, 1972).

The second paper in the series, Hamlet et al (1981), adapted the protocol to give variable times after fixed doses of 10Gy (1000rads) for X-rays and 5Gy (500rads) for neutrons. As above, gamma or neutron irradiation produced early lateral or vertical villous collapse respectively, both with further progressive changes. Other features included early changes at villous tips, possibly associated with the extrusion zone or epithelial/stromal disruption, later fused villi (Anderson and Withers, 1973), pericryptal cell mounds, later called collared crypts (Carr et al, 1990, Section 3.4) and giant cells (Carr et al, 1981a, Chapter 2, Section 2.3, Carr et al, 1981d, Section 3.6). The report of possible stromal involvement highlighted the need for correlative studies of villous surface changes, as recommended in Chapter 2 (Section 2. 3). An addition to the literature in 1980 had come from a doctoral thesis on radiation-induced changes to rat small intestine by L-G Friberg, one of the reviewers of the Hamlet et al (1981) paper, another being J. Anderson (Anderson and Withers, 1973). The questions touched on sample preparation, evidence for the cellular nature of giant cells, other radiation-induced changes and correlative studies. This paper was cited for radiationinduced intestinal epithelial damage, permitting bacterial translocation, an important aspect of sepsis (Elliott et al, 1995).

The third paper in the series (Carr et al, 1979b) related villous radiation damage after three fractionation schedules of low LET Cobalt⁶⁰ irradiation (Withers et al, 1975). It noted that damage after high total doses sometimes included heaped circles of cells around crypt mouths (Hamlet et al, 1981), calling for comparison with coeliac disease (Toner et al, 1971, Wellwood and Jackson, 1973). A technical development was the use of a pilot assessment system, adding subjective scores for seven villous features. This showed that the imbalance continued between villous and crypt count changes (Hamlet et al, 1976), this time to a different extent across the three fractionation schedules. With respect to citations, the core point of the paper was identified by Zook et al in 1983, one of the authors being G. Casarett: they reported, in their study of the effects of neutrons or photons on dog alimentary organs, that it had 'argued that crypt cell counts do not always correlate with villous cell death'.

3.2 Early reviews on 'topographical' and ultrastructural radiation effects

The early publications on radiation damage (Carr and Toner, 1972, Section 3.1, Anderson and Withers, 1973, Heinzmann et al, 1978, Porvaznik, 1979, Friberg, 1980) included several published through the annual Scanning Electron Microscopy conferences, one being an early review (Carr, 1981). Regular meetings were therefore established on 'ultrastructural radiobiology' at this forum, leading to the publication of a booklet (Carr and Seed, 1983) compiled by the session organisers: T. Seed was a radiobiologist at the Argonne National Laboratory, with electron microscope (EM) interests. These papers included the review mentioned above (Carr, 1981), in which intestinal radiation damage was compared to that produced by other challenges such as methotrexate (Altmann, 1974), ulcerogenic treatment (Carr et al, 1979a, Chapter 2, Section 2.1) and abnormalities such as human coeliac disease (Toner et al, 1971, Chapter 2, Section 2.2, Wellwood and Jackson, 1973). Intestinal radiation effects included ulceration (Friberg, 1980) vascular changes (Egawa and Ishioka, 1978) and responses in microorganism populations (Porvaznik, 1979, Friberg, 1980). The surface changes began with damage, through stages of villous change from lateral collapse to a flat mucosa. Damage was followed by repair. SEM scoring systems for radiation damage included one entirely villous (Carr et al, 1979b, Section 3.1) and another scoring on three levels (Friberg, 1980). The reaching of any of the villous stages could be used as an 'end-point' to produce radiobiological equivalent (RBE) doses to allow comparison across radiation qualities. The question was raised, however, of RBEs being non-standard for different component tissues or compartments, a disadvantage in their use for such a multiparameter organ, a subject dealt with later (Section 3.8). Surface damage was again described as probably related to underlying stromal effects and the review also gave brief descriptions of radiation injury to other tissues, such as vascular changes or neural effects, as well as to other epithelial responses. With respect to citations, this review was quoted, like the group's other papers, by Somosy et al (2002), again for the time of maximum damage and also for the formation of radiation-induced giant cells. Two papers on radioprotection (Kanter and Akpolat, 2008, Akpolat et al, 2009), cited it as showing damage at subcellular level, such as lateral and basal epithelial projections at intercellular spaces.

Extracts from the booklet (Carr and Seed, 1983) covering the 1981 and 1982 'ultrastructural radiobiology' sessions have been selected for inclusion in the thesis, to show the range of topics and information on input from reviewers, who, with the authors, included many of those working in the field. The concluding remarks by Seed and Carr touched on the

variation in radiation responses and the range of 'target' cells, radiation conditions or analytical tools used. *In vivo* topics included heavy ion induced retinal lesions; the role of type II pneumocytes; the importance of LET in intestinal damage; and structural stages in the onset of myeloproliferatove disorders. *In vitro* topics included: membrane damage; modifications of lectin or hormone binding; chromatin anchoring sites and cell shape. The 1983 booklet had almost no citation record.

3.3 Radiation effects at different sites

Most of Chapter 3, as for Chapter 2, has so far involved the small intestine, but other organs were also studied, all lined or covered with complex epithelial layers

With respect to stratified squamous non-keratinising epithelium, the surface structure of oral mucosa had been studied previously by SEM (Matravers and Tyldesley, 1978, Dourov, 1984) and microridges have been described above (Chapter 2, Section 2.5). In Robertson et al (1987), non-invasive sample collection was made of buccal smears from radiotherapy patients after tumour excision, allowing SEM assessment of microridges before, during and after fractionated radiotherapy: groups included control and pre-irradiated smears, those irradiated in four dosage ranges from 3 to 60Gy and finally post-irradiation samples. Control microridges were usually in parallel, equally spaced arrangements, with variable mucus and low bacterial colonisation. The pre-irradiation cells were similar, but with slightly more ruffled microridges, abnormal microvilli and also mucus and bacteria. A possible relationship was described between patient radiotherapy dose and microridge patterns, either directly or through exposure of cells at different stages of differentiation, implying that the surface details of this cell type (Chapter 2, Section 2.5) could have reflected the state of the epithelium. The citations for this paper (Robertson et al, 1987) have been difficult to interpret. On the one hand, further work by one of the radiotherapists/oncologists involved (A. Robertson), working with a different group of morphologists, concluded strongly that SEM was not likely to be useful in this way (Badran et al, 1994, title). On the other hand, Asikainen et al (2014), in their experimental study of irradiated dog oral mucosa simply quoted the original Robertson et al (1987) paper along with others and themselves concluded that 'radiation disrupts superficial cells of the oral mucosa'. The same group (Asikainen et al 2015) in a review article on microridges (or microplicae as they called them) again merely quoted Robertson et al (1987) for radiation changes in the epithelium or its surface features and concluded that 'The three-dimensional findings with TEM and SEM offer the opportunity to re-evaluate the classic histopathology

of the oral mucosa and oral mucosal diseases using more refined microtopographical parameters'. They also recommended further study of the tips of 'microplicae' and the use of immuno-localisation at electron microscope level.

With respect to urothelium, the paper by Hodges et al (1985), already mentioned for SEM methodology (Section 2.4), described immunogold-labelled mouse bladder after 5Gy whole body neutron irradiation. Surface atypia included blebbing and pleomorphic microvilli. There was parallel redistribution of some conconavalin A receptor sites from inter-plaque (later termed hinge, Liang et al, 1999) to plaque regions, possibly because of alterations in the cytoskeleton (Oliver and Berlin, 1982) or stroma (Hodges, 1982), another sign that surface responses reflected deeper problems. Discussion with reviewers covered internal cellular changes, surface atypia and lectin binding. The citations used this paper for radiation-induced changes in bladder epithelium, specifically after neutron treatment and greatest at 1 to 5 days after treatment (Dörr, 1995) and for membrane lectin-binding (Somosy, 2000, review).

With respect to epidermis and wound repair, the last two papers in this Section described rat skin irradiated with an electron beam (Mould, 1986), of interest because of its depthdose distribution. This involved either incisional healing, already recorded for control and X-irradiated situations (Messerschmidt, 1986) or intraoperative radiotherapy (IORT, Swanson et al, 1988) of a graft-bed. The project was initiated by a Chinese plastic surgeon, who arrived with ideas already formulated for his post-graduate project. Wang et al (1994) reported on wound healing, 1 to 14 days after an incision, itself 7 days after local, 9.6Gy electron irradiation. The SEM images charted the healing wound and measured granulation tissue via image analysis taken directly from the microscope. The delay in healing reflected reductions in granulation tissue formation, epithelial cell migration, fibroblast proliferation and the inflammatory response: collagen bundle formation was also slower. Wang et al (1996) studied a graft bed rotated 180° from its original site 1 to 21 days after irradiation with 10Gy 6MeV electrons delivered as IORT. Irradiation again delayed but did not stop wound healing, reducing deposition of fibrinogen, fibrin and fibronectin, affecting reepithelialisation and granulation tissue formation and reducing collagen fibril diameter. This paper recorded that the procedures were approved in Beijing, reflecting the disruption of the Belfast group's work by anti-vivisection pressure: graft formation and irradiation were carried out in China, with specimens brought to Belfast for analysis. With respect to citations, Wang et al (1994) was used with respect to wound healing (Kurul et al, 1997, Bernatchez et al, 1998). Citations for the second skin paper (Wang et al, 1996) included some to technique, such as grafting methods (Eckhaus et al, 2008). Others cited it as reporting that there was delay in wound healing and less fibrin and fibronectin (Gentilhomme et al, 1998), the latter also quoted by Currie et al (2001). Ferguson et al (1999) cited both papers, the first for reporting reductions in fibroblast proliferation and migration and also in collagen volume and the second for the collagen being structurally altered.

3.4 Qualitative radiation effects on small intestinal surfaces

Returning to irradiated small intestinal surfaces, the next three topics have been chosen to illustrate features of interest separate from the developing theme of scoring the changes: they involved irradiated duodenal papilla, collared crypts and giant cells.

With respect to the papilla, Indran et al (1988) used the capacious SEM stage to record large samples, with the gastroduodenal junction as a reference line, (Carr et al, 1989, Chapter 2, Section 2.4). SEM produced more information than was previously available on the papilla (Edemskiĭ, 1983, Edemskiĭ and Svischev, 1986). Its opening was expanded by radiation and became easily visible. This could have been produced by radiation-induced effects on neuromuscular components, taking forward the proposal (Hamlet et al, 1981, Section 3.1) that non-epithelial tissues were important, highlighted again below (Sections 3.6, 3.7, 3.8). The relaxation of the papillary opening could in turn have functional significance. The only available citation to this paper (Gorea et al, 2010) was as a source reporting the obvious alterations to the papilla.

With respect to collared crypts, Carr et al (1990) explored further these enlarged regions resembling 'heaped up mounds of cells' (Hamlet et al, 1981): they were found around crypt mouths in the intervillous basins or clefts (Cocco et al, 1966) and resembled somewhat those seen in human coeliac disease (Toner et al, 1971, Chapter 2, Section 2.2). The collared crypt paper explored the effect of increased linear energy transfer (LET) on their appearance and incidence in mice 3 to 7 days after irradiation at Berkeley. This extended the collaboration with T. Hayes to include J. Ainsworth. Control crypt mouths were unremarkable, with no marked collars and openings approximately 5µm in diameter, but the entire collared crypt measured 87µm across, with an opening of 10µm. They were seen in some circumstances after neon and iron but not silicon ion irradiation, adding to the earlier sightings after a single dose of neutron irradiation (Hamlet et al, 1981). The collared crypts after iron ion irradiation were so tall that they could have alternatively been described

as villi with a central orifice: this suggested that crypts and villi were definable not only by shape, but also by vascular arrangements, a crypt having a peripheral network and a villus a central core. After X irradiation, they were only found after a high, fractionated dose (Carr et al, 1979b, Section 3.1). They seemed, therefore, to be mainly associated with some high LET irradiation conditions. Resin histology and TEM revealed possibly equivalent cryptal regions, with dark cells and prominent shoulders. This paper (Carr et al, 1990) produced no available citations of interest.

With respect to giant cells, correlative techniques had shown these to be epithelial swellings (Carr et al, 1981a, Chapter 2, Section 2.3). Five days after irradiation with either X-rays or neutrons, they were seen with SEM to be outgrowths 25 - 60µm in diameter, similar in size to the collars described above. Resin histology and TEM showed that giant cells had several nuclear profiles, lipid inclusions, vacuoles, microvilli, boundaries with neighbouring enterocytes and attachment over lengths of 40 - 80 µm to the underlying basement membrane and stroma. They were possibly formed after radiation damage to cryptal progenitor cells, allowing continuing division but incomplete separation of daughter cells. They had not been reported in vivo by other groups but had been seen in in vitro Hela cells (Tolmach and Marcus, 1960) and fibroblasts (Nias and Paul, 1961). With respect to citations to this paper, Syljuåsen et al (1997, with R. Withers included as one of the authors) listing it as one of only two reports on these cells in vivo, stated that their formation 'has not received much attention compared to other pathways to cell death such as apoptosis' and that 'they are an important pathway to cell death for first generation single cells after irradiation'. Another citation (Bhattathiri et al, 1998) quoted Carr et al (1981) as linking giant cells to problems in cytokinesis.

3.5 Scoring of radiation-induced changes to intestinal surfaces

The two papers featured here took forward the concept of quantifying radiation-induced villous shape changes to a more objective scoring system. Carr et al (1983c), again in collaboration with R. Hamlet and T. Nias, set out the basic concepts, important for understanding patients' radiation injury (Berthrong and Fajardo, 1981). Previous small intestinal SEM scoring systems (Carr et al, 1979b, Section 3.1, Carr, 1981, Section 3.2) were dependent on the three dimensional appearance, whereas this development related these changes in mice directly to those recorded histometrically for rat villous area in sections (Altmann, 1974): the ratios of these 'two-dimensional' changes with time, 1/2.5/2.5, were used to quantify the villous shape changes. This scoring system was then

used in two ways. Firstly, mean values were plotted for different radiation schedules, using scores from 0 for control erect villi to 6 for rudimentary structures and to 8.5 by extrapolation for a flat mucosal surface. Alternatively, a ratio was calculated of the doses of two radiation schedules required to produce one stage of collapse, such as conical villi, using this as an 'end-point', to give RBE values. Application of either of these more quantifiable methods confirmed that neutron damaged villi were more abnormal than those produced by X-irradiation, in conditions calculated to be equivalent for crypt loss. The paper (Carr et al, 1983c) also related villous shape changes to alterations in two different stromal compartments, suggesting involvement of villous stromal pegs and the stromal pericryptal plate. Control mucosa had competence in one, the other or both of these. This paper (Carr et al, 1983c) produced citations to villous surface effects (Köst et al, 1998) epithelial responses (Höckerfelt et al, 2000) and 'well documented' structural changes (Freeman et al, 2001, with W.K MacNaughton, a notable expert in the field, as one of the authors).

The other paper, (Carr et al, 1986b), in collaboration with A. Michalowski in the Hammersmith Hospital, adapted the scoring system to analyse larger intestinal areas, from the gastroduodenal junction to the duodenal papilla. More detail was produced through 'villous mapping', a detailed but time-consuming technique used in only one other study (Chapter 3, Section 3.7). After irradiation of a non-abdominal region, such as the lower mediastinum, there was localised intestinal 'ulceration' at 'abscopal' i.e. distant sites (Michalowski et al, 1983). An acetate sheet with grid squares was placed over standard prints and each square allotted a villous score, from which was calculated the total for that sample. Three days after irradiation, there was little change, but by five days, a range of sizes of sharply defined lesions containing conical villi began at the stomach junction: at 28 days, variable lesions were still present. Proximal regions were more damaged than distal. Possible causes were damage to sphincteric, glandular, vascular or nervous structures. The lesions differed from those produced by secretagogue exposure (Carr et al, 1979a, Chapter 2, Section 2.1), in that the abscopal 'ulcerated areas' usually had an epithelial covering and were therefore not strictly speaking 'ulcers', despite appearances in light microscopy (Michalowski et al, 1983). The stages of lesion development and 'recovery' were also different in the two forms of 'ulcer'. The only citation of note to this paper used it as a source for variation in the damage seen (Höckerfelt et al, 2000).

3.6 Importance of underlying structures to surface responses

The next four papers explored the internal responses deep to the surface changes already described (Sections 3.1 and 3.5).

With respect to stroma and villous/cryptal relationships, the first paper (Carr et al, 1981c), used autolysis (Ferguson et al, 1969, Chapter 2, Section 2.1) to reveal sub-epithelial stromal detail. Control samples treated in this way had double rows of crypts, each opening from a tube which was sometimes divided by a partition. A test frame (Gunderson, 1977) over micrographs of mouse small intestine was used to calculate the ratio of crypts to villi (Hagemann et al, 1970): this was 5.01/1 SD 0.73, n 32, close to previous findings of 4.53±0.99 (Smith and Jarvis, 1980). After irradiation, it was not easy to identify either the crypt mouths or their double rows, as reported earlier for unautolysed rat intestinal samples (Anderson and Withers, 1973). Many were occluded by stromal material and the crypt/villus ratio could not be established. Other radiation-induced abnormalities included a villus formed of skeins of cells twisted helically around each other, suggesting a spiral epithelial path from crypt mouth on to and up an adjacent villus. Radiation was used here as a 'probe' to distort the normal situation and facilitate consideration of the control results. Citations for this paper included an editorial in the Journal of Paediatric Gastroenterology and Nutrition, where it was referenced for the ratio of crypts to villi being 5/1, for possible radiation-induced spiral migration of cells from crypts to villi and for the use of SEM to study healing coeliac sprue (Lev, 1983). It was also quoted as reporting on how postmortem changes have been helpful in the study of the structure of small intestine (Pietzonka et al, 2002a). This 1981c paper was also mentioned by Ross and Mayhew, (1984) in a paper on the advantages of stereological methods, citing it to point to the problems of quantifying villous shape changes after irradiation, but possibly just pre-dating the publication of the scoring system (Carr et al, 1983c, Section 3.5) linking them to histometric data. The possibility of spiral cell migration pathways was again noted with respect to cell migration patterns in chimeras, where it was shown that the pathway of epithelial cells was straight from crypt to villus, in a series of papers all of which included B. Ponder and one of which had N. Wright as an author (Schmidt et al, 1985, Wilson et al, 1985, Ponder et al, 1986).

With respect to the epithelial-stromal boundary, Carr et al (1981d) gave more information on giant cells (Carr et al, 1981a, Chapter 2, Section 2.3, Hamlet et al, 1981, Section 3.3), one containing an intracytoplasmic desmosome, possibly indicating that cell fusion or incomplete cell division could have had a role in its formation. The stromal changes seen by TEM also included contact between epithelial and stromal cells through basal lamina herniation, the basal lamina looping away from the base of the epithelial sheet and complete epithelial/stromal stripping, comparable with the swollen villous tips seen with SEM. These interface abnormalities were reported as possibly being associated with weakened enterocyte function. This paper (Carr et al, 1981d) was cited by Eigenbrodt et al (1995), as one of the sources for radiation-induced structural responses, including giant cell formation, epithelial/stromal abnormalities and villous shape changes.

With respect to stromal components, the next paper (Carr et al, 1985) used high doses (18-20Gy Cobalt⁶⁰ gamma or 10Gy neutrons) to reveal where such damage might be seen 3.5 days after irradiation. Villous shape changes were recorded (Carr, 1981, Section 3.2) and, since the neutron irradiation completely destroyed villous structure in one of the animals, more attention was paid to the pericryptal plate (Carr et al, 1983c, Section 3.5). TEM showed that deep stromal fibroblasts in both irradiated groups were irregular with more cytoplasmic inclusions: these were pale after gamma irradiation and dark after neutron treatment, implying lipid and lysosomal responses respectively. Gamma and neutron irradiation also produced responses in muscle and nerve respectively and there were changes in endothelial flaps, particularly after gamma irradiation. Some of these responses were similar to previous reports, including radiation-induced damage in other gut models (Lieb et al, 1977), in nerve (Mastaglia et al, 1976), at the epithelial/stromal interface (Chomette et al, 1977, Lieb et al, 1977) or in vessels (Egawa and Ishioka, 1978, Eriksson et al, 1983). One conclusion was that more data were needed, from a range of doses, radiation qualities and time-points. With TEM, this would have been very time-consuming and, as a result, light microscopy was preferred (Section 3.8). This paper, like Hamlet et al (1981), was cited by Elliott et al (1995), who used it as a source for the different changes caused by neutrons and gamma photons to villous shape and the stromal pericryptal plate.

With respect to the basement membrane at the epithelial-stromal interface, Carr et al (1987b) used the heavy ion facility at Berkeley to explore further how internal and surface changes might relate to each other, in another comparison of low and high LET treatments (1600 or 1720 cGy X-rays and 1000 or 1240cGy neon ions). The interest in heavy ion irradiation was by 1987 related to space research as well as to cancer treatment (Steinberg et al, 1990), the previous trigger for research (Leith et al, 1982). The paper by Carr et al (1987b), built on the data available from crypt survival (Alpen et al, 1980) and late ultrastructural changes, where basement membranes responded differently after heavy ion and gamma irradiation (Fatemi et al, 1985). Villous scoring was carried out and parameters

were analysed per circumference in resin histology sections stained with periodic acid Schiff (PAS) or silver, as well as the routine toluidine blue: the structures assessed included crypt counts, villous profiles, goblet cells and basement membrane integrity. All changes were related to their own control values rather than producing ratios of effects for the two schedules. X-irradiation had a greater effect on crypt counting than predicted (Alpen et al, 1980), making it more notable when neon ion irradiation produced a larger response, such as at basement membranes: this reinforced the importance of the epithelial/stromal boundary. The reviewers included S. Siew (gastroenterologist), R Laschi (editor of the Journal of Submicroscopic Cytology and Pathology) and T. Seed (ultrastructural radiobiologist at the Argonne National Laboratory). Questions or comments related to doses, sample orientation, staining of resin sections, non-removal of mucus, collared crypts and comparison of crypt counts with those of Alpen et al (1980). Dörr et al (1999), in a report on pig lung response to accelerated ¹²C ions, cited this paper as one of only a few with data from *in vivo* experiments.

Concentrating on stromal vessels, two papers addressed different aspects of the response. The first has already been discussed for the description of control microvascular casts of small intestinal villi (Abbas et al, 1990a, Chapter 2, Section 2.4). Three days after 10Gy X-rays whole body irradiation, the vessels were more tortuous and compressed apico-basally than in the controls. Destruction of the apical parts of the cast (Egawa and Ishioka, 1978) was not seen, possibly because of the glutaraldehyde fixation and critical point drying (Lametschwandtner et al, 1984). Resin histology and TEM revealed capillary dilatation and luminal endothelial extensions. The paper concluded that more data were needed on how separate the vascular response was from the epithelial changes. The technical citation record for this paper was set out in Chapter 2 (Section 2.4) on SEM methods. Data-based citations were also made to its reports of changes in capillary tortuosity and density (Jahnson et al, 1998) and to vascular constriction (Chikui et al, 2012).

Continuing the stromal vascular theme, Abbas et al (1990b) added the earlier time-points of 6hrs and 1 day after 10Gy partial body X-irradiation: the villous transverse sections for light microscopy and TEM allowed good morphometric comparison between irradiated capillaries and those in control villi (Abbas et al, 1989, Chapter 1, Section 1.2). There were consecutive capillary dilatation, constriction and then mixed results. Changes were also seen in endothelial thickness, fenestrae and processes and in the perivascular space. Similarities were noted between these intestinal vascular responses and those in skin (Hopewell, 1980) or cultured endothelial cells (Friedman et al, 1986). The changes

corresponded to functional alterations rather than cell death and could have contributed to the radiation-induced response of this multicellular organ. The citations to this paper included references to it for radiation-induced endothelial pseudopods (Lee and Schmid-Schonbein, 1995) and, as to the previous paper (Abbas et al, 1990a), for changes in vascular diameter (Chikui et al, 2012).

Two other papers dealt with changes to other parameters in villous and cryptal compartments. The first (Carr et al, 1984c) compared numbers for villous scores, intraepithelial lymphocytes, mitotic figures, parasitised crypts and giant cell incidence 5 days after 10Gy X-ray and 5Gy neutron irradiation. Villous tips showed oedema and epithelial/stromal stripping after X and neutron irradiation respectively. The reviewers included WE Dewey, WR Hanson and T Seed, all with a radiobiology background, and they asked questions on sample preparation, the parameters used, the villous scoring method, giant cells, dose rates, RBE calculations and the use of 'end-points'. The authors commented that RBE values for neutrons vs X-rays probably could not be extrapolated from intestine to other organs, such as skin (Hornsey, 1970, Hornsey and Field, 1974): the paper proposed further that RBE calculations for 'any one manifestation' of damage to a functional unit will apply only to that 'manifestation' and not to other cell populations (Chamberlain et al, 1980). The difficulty of the end-point approach was more obvious as the range of responses for different parameters increased, making it a challenge to use them to comment on total damage. Reviewers also requested statistical information, previously avoided because of small samples and 'n' values: when this was provided, it showed that, in data from sections, only the mitotic figure values were significantly different across the two radiation qualities. This point was borne in mind when a system was developed to report changes in all four tissue types in the small intestinal wall (Section 3.8). Elliott et al (1995) cited this paper (Carr et al, 1984c), like Carr et al (1985) and Hamlet et al (1981) for neutrons causing more villous responses than X-rays, while Dublineau et al (2002) concentrated instead on its report of approximately parallel time-related responses.

Also with respect to parameters and the possibility of different response 'manifestations', the next paper asked whether constituent cell types in a compartment all responded similarly to each other and to the compartment as a whole. Information was already available on many of the cryptal cell types (Potten et al, 1983) and the Wyatt et al (1987) paper investigated one of these, the endocrine population. The influence of gut hormones was already of interest (Besterman et al, 1982): since the neural component responded (Carr et al, 1985), it was important to explore this other player in alimentary tract control. Groups

of mice studied included fed, fasted, sham-irradiated and ether sedated shams, but only at one time-point. Irradiated samples were collected 1 or 3 days after partial body Xirradiation with 6, 10 or 18Gy, these being on the plateau, shoulder and exponential parts respectively of the crypt count dose-response curve. Most endocrine cells were cryptal (Ferreira and LeBlond, 1971) and control villous shapes were as previously reported (Carr and Toner, 1968, Chapter 2, Section 2.1), although the fasted group had slender, 'cleaner' villi. After some irradiation schedules, epithelial/stromal stripping was inferred from SEM images (Carr et al, 1981d), confirming that this region was at risk. Endocrine numbers decreased. Both this and villous shape were more sensitive than crypt counting, showing that a response in this compartment did not reflect the changes in all its constituent cells. Reviewers included CP Sigdestad, F Bonvicini, MS Al-Tikriti and T Seed. There were questions or comments on: the functional significance of the endocrine changes; the sham/ether effect; crypt size; sampling of other parts of the small intestine and identifying specific types of endocrine cells; why the animal pool was female only; the choice of doses; and epithelial/stromal stripping. With respect to citations, Panozzo et al (1991) cited the paper for the endocrine cell number decreasing after irradiation.

3.7 Mechanisms of structural damage

The papers in this section compared radiation effects with those after other procedures, either those acting as useful comparators, such as hyperthermia, or those with known mechanisms, such as pharmacological agents.

With respect to hyperthermia, when the Glasgow neutron facility closed, a collaboration was established to continue that work with the radiobiologists at the Hammersmith Hospital, London and to contribute further morphological data to their hyperthermia programme: this aimed to optimise its possible role in cancer therapy (Hume et al, 1979a, b, Tsubouchi et al, 1984). The paper included here (Kamel et al, 1988) was only one of several from the group (Carr et al, 1982b, Kamel et al, 1985, Dickson et al, 1992). It used light microscopy and SEM villous scoring (Carr et al, 1983c, Section 3.5) to compare the effects on small intestine of hyperthermia at 43° and 10Gy whole body X-irradiation. These two treatments produced maximum damage by 2hrs and 3.5 days respectively: the corresponding times for substantial recovery were 24hrs and 7 to 9 days. This timescale for hyperthermia was too short for villous shape to be reconstructed entirely through epithelial proliferation. Some of the responses after both treatments could therefore have been related to changes in non-epithelial components such as smooth muscle. This paper was cited in technical terms by

Oliver et al (2012), who adapted the scoring system to assess *in vitro* segments. It was also quoted, mainly with respect to data relevant to irradiation, along with Carr et al (1983c) and other papers, for the documentation of the morphological changes produced in small intestine (Höckerfelt et al, 2000). Forsgren et al (2000, co-authored with Höckerfelt), cited it for villous damage, as did Somosy et al (2002), particularly with respect to responses after different types of treatment.

The next three papers tested the hypothesis that radiation-induced villous changes were related to neuromuscular effects. This was done, in collaboration with Frances Boyle at Glasgow University's Pharmacology Department, by comparing them with those produced by pharmacological agents that either mimic radiation or reduce its effects by acting on this part of the intestinal wall.

With respect to radiomimetic effects (Indran et al, 1985, 1991), the 'radiation' aspect related to radiation responses in epithelum (Quastler and Hampton, 1962, Friburg, 1980, Carr et al, 1984c, Section 3.6) or muscle (Wirtz et al, 1982). The 'mimetic' effects were represented by reserpine, which increased gut motility through loss of inhibitory control, producing changes to smooth muscle (Casteels and Login, 1983) or nervous tissue (Patel et al, 1979). The 1985 paper (Indran et al, 1985) confirmed that ascorbic acid, the reserpine vehicle, produced clean mucosal surfaces (Maugel et al, 1980). Both irradiated and reserpinised treatments were shown by SEM scoring and light microscopy to produce villous shape changes and smooth muscle abnormalities respectively. The reviewers included C.S Potten from Manchester, UK, with a reputation in analysing different intestinal epithelial cell types after irradiation and J.R. Poley from Ulm, Germany, with a background of paediatrics and SEM in clinical applications. Reviewers' questions or comments included interpretation of villous scores; brush border cytoskeleton; colchicine effects; crypt morphology; links between villous and muscle changes; fixation techniques; the use of duodenum; cell position in crypts; and epithelial and vascular damage. Indran et al (1985) was cited by one of the authorities in the field, Hauer-Jensen (1990), who, while stating in a review that the crypts were the main target, quoted it as reporting that other parts of the wall could respond, thus taking note of the key point of the thrust of the work. The 1991 paper reported villous scores as 'enhancement factors', since shams were available for all groups. Damage occurred later after irradiation than after reserpine. Resin histology and TEM showed smooth muscle changes, deep to sites of villous collapse: it was concluded that this tissue could have contributed to the changes up to villous conical collapse, but that the more advanced rudimentary villous shape after irradiation required additional factors such as epithelial

lethality. This underlined the need to record responses in as many components as possible of multicellular organs. For citations to this paper, Amer et al (2005), noted its reference to villous changes and neuromuscular effects.

With respect to 'radioprotective' effects, Carr et al (1991a) reported on responses to atropine, which produced opposite effects to those of reserpine, decreasing gut motility. It was also known to affect the radiation response (Larkin, 1949, Burnett et al, 1952). Mice were irradiated with and without atropine. The radiation-alone schedule produced more villous damage than the combined treatment. MacNaughton (2000) cited this paper in a review on the pathogenesis of intestinal responses to irradiation: he noted the 'detailed morphometric studies by Carr and colleagues'. Somosy et al (2002) included it in a list of publications describing a range of effects. The main point of the paper was quoted for atropine blocking radiation-induced villous shortening by MacNaughton (2000) and Basson (2003, also a review paper). Özyurt et al (2014) referred to it while reporting a non-significant trend in the radioprotectant effect of this cholinergic antagonist. Work was also under way on conventional radioprotectants misoprostol and WR-2721 in collaboration with a group in the USA specialising in this area, led by W. Hanson. This was only published in Abstract form (Carr et al, 1991b) and has not been included as a submitted paper in the thesis.

Overall, the mechanistic arm of the programme had delivered an outcome, in supporting the need to study all tissues of the organ: the work on mechanisms of radiation damage and radioprotection was not pursued further.

3.8 Summary of small intestinal responses

The material here followed on from the findings on radiation-induced changes to different components of the intestinal wall (Section 3.6) and on the time-frames of responses to hyperthermia and pharmacological agents (Section 3.7): this approach also reflected the links between morphological and biochemical changes (Nikjoo and Goodhead, 1989). It focussed on producing, with collaborators in the USA and the UK, detailed below, a standardised approach for collection of quantitative information from well-preserved tissue sections and on controlling the time taken for analyses, using respectively data/circumference and resin histology rather than TEM. The responses of the key cells and structures in the intestinal wall were collected and brought together in one figure (the Morphological Index, or M°I). Six papers have been included here. The first, using neutron-

irradiated samples, set out the method for two data levels, carrying information on tissues and cells. The second and third covered X irradiation after different doses or times and a comparison of neutrons and X-rays respectively. The fourth added ultrastructural data, extending the Index to three levels, while the last two presented data relating to heavy ion irradiation.

The first paper (Carr et al, 1991c) collected information on different tissues (Fajardo, 1989, Rubin, 1989), using mouse duodenum samples at four time points after whole body neutron treatment. Control tissue was the comparator and the parameters were also assessed at one time after X-ray treatment. Irradiation was carried out by S. Hume and colleagues at the Hammersmith Hospital and the data handling method was devised by the thesis author and A. Nelson, sequentially at Lawrence Berkeley Laboratory (LBL), the Massuchusetts Institute of Technology (MIT) and the University of Washington, Seattle: he also set out the procedures mathematically in an equation. Data collection, performed in Belfast, included areas for the four main tissues (epithelium, connective tissue, muscle and nerve) and counts for ten parameters, each assigned to one of the tissues. All measurements were per circumference, as for the microcolony assay (Withers and Elkind, 1970). The data were set out in tables, one row per tissue, and showed the statistical significance, direction of any change and ratios of the data, treated as absolute numbers, for treated/control values, categorising this as a Ratio Score system (Section 3.10). Data for each tissue row, weighted for area, were multiplied to give one figure, which could then be added for the four tissues to give a single number, called the Morphological Index: this was calculated per group for mean values or per animal to provide error bars for graphs. Information on villous numbers, crypts and mitotic figures were often included in the table, but not in the Index equation.

Interpretation of the changes contributing to the final Index number were read from the tables, as follows for neutron irradiation. At 6hrs, there was little change, but at 1 day the decrease in the Index related to changes in endocrine cells and Auerbach's plexus. The further Index drop at 3 days was due to significant epithelial, stromal and nerve responses. By 7 days, the return towards a control level Index was reflected in the lack of individual changes, found only in the stromal cells. After both treatments, the epithelial Index was affected more than others, but the two sets of individual responses were different. The conclusion above (Sections 3.1, 3.5, 3.6, 3.7) that non-epithelial cells contributed to the overall changes was confirmed by these findings. This paper (Carr et al, 1991c), with its range of data, was quoted as one of a number describing changes in different individual

epithelial cell types (Somosy et al, 2002), including goblet cells (Kanter and Akpolat, 2008) and stromal cells (Ch'ang et al, 2012).

With respect to the Index and variations of dose and time after partial body X-irradiation, the multi-tissue approach continued to be used (Carr et al, 1992a), while others reported radiation-induced effects on small intestinal epithelium (Potten et al, 1983). The Index displays included additional multiple graphs, showing the complete lack of uniformity in the parameter responses. For the dose experiment 3d after irradiation, the total Index decreased with dose, although that for 10Gy Index overlapped with the others. Reading the tables showed that, after 6Gy, endocrine cells were affected, at 10Gy there were several epithelial responses, but after 20Gy only two such changes appeared. The 10Gy time experiment data showed that the 6hrs total Index was within the control range, with only submucosal arterioles responding. The 1d Index was somewhat lower, with fewer endocrine cells, but larger error bars: the 3day data were as for the dose experiment. Carr et al (1992a) was the earliest of three cited by Scanff et al (2004) for the 'well-known' diminished absorptive surface, associating this with impaired active transport. It was also quoted, like the paper above, for changes to goblet cells (Kanter and Akpolat, 2008).

With respect to the Index for two radiation qualities, whole body irradiation was used, 5Gy neutrons and 10Gy X-rays, sampled at four time-points (Carr et al, 1996a). The displays included Index graphs and a parameter Table. The literature had continued to expand, with respect to intestinal irradiation (Busch, 1990, Cobb and Galland, 1990, Flickinger et al, 1990) and multi-parameter or multi-level approaches following experimental exposure (Argenzio et al, 1990, Goodlad and Wright, 1990, Mayhew et al, 1990, Vigneulle et al, 1990). The 6hr Indices showed greater damage by neutrons than X-rays, possibly related to the number of parameters responding. After 1 day, the continuing greater neutron effect reflected changes in endocrine cells and Auerbach's plexus profiles, while the similar 3d Indices reflected the individual parameter situation. At 7d, the neutron Index was at control level, while that for X-irradiation was still lowered, possibly due to greater data variation. In conclusion, crypt depletion was not predictive for changes to other parameters or compartments: both schedules led to epithelial changes and while X-irradiation produced early vascular changes, neutron treatment gave neuromuscular effects. Citations to Carr et al (1996a) included protocol topics, such as parameter assessment (Patel et al, 2012) or choice of time-points (Lehy et al, 1998) and also general radiation-induced intestinal damage (Yan et al, 2009). It was also mentioned for the greater effect of neutrons than Xrays (Mizukami-Murata et al, 2010) and for data for one or more of the parameters, such as

64

endocrine cells (Lehy et al, 1998) or myenteric plexus profiles (François et al, 1999, Ropenga et al, 2004): authors of papers from the French radiobiology groupings included I. Dublineau and/or N. Griffiths. As further papers were published on the Morphological Index, more groups appeared to obtain from its data display results for different parameters of interest to them.

While work was being done on adding extra ultrastructural or topographical information (Carr et al, 1992b), there had continued to be interest elsewhere in responses in different compartments (Fajardo, 1989, Rubin, 1989, Wright et al 1989). Villous scoring was carried out as described in an earlier paper (Carr et al, 1983c, Section 3.5), which had also recorded some of the responses without constructing an Index display. The additional factor for ultrastructural deviation (UD) was calculated as for the resin histology data (Carr et al, 1991c), where the two-level Index was analytical, using statistically testable numbers. However, the three-level Index, although more informative, was empirical, since it also contained subjective micrographs assessment. Ultrastructural responses were seen for epithelium in nuclei, organelles and membranes, for connective tissue in cell separation and endothelial structure, for muscle in vacuolisation and myo-filamentous arrangements and for nerve in dense bodies. Study of the villous scores and other data showed again that villous shape could be related to responses in muscle tissue (Section 3.7). Reviewers included LG Friburg and M Albertson, both with an interest in ultrastructural radiobiology, and J R Poley, with experience in SEM of gastrointestinal material. Topics raised included: the choice of neutrons, dose and proximal small intestine; extrapolation to late effects in human tumour tissue; vascular changes; cell separation; reasons for the changes in Auerbach's plexus profiles; and the selection of the time-points. Carr et al (1992b) was quoted (like Carr et al, 1986b) for having reported radiation-induced morphological changes (Höckerfelt et al, 2000). Other groups already mentioned above as citing the group's work continued to do so (Kanter and Akpolat, 2008, Akpolat et al, 2009, 2011). Rubio and Jalnäs (1996) commented on the presence of SEM and TEM data, but reported that there was 'no conventional histology', with no apparent reference to the resin histology data displayed in the tables.

With respect to Indices for heavy ion effects, (Carr et al, 1993 and 1994), this related to interest in space research after the Berlin Wall came down in 1989. Open discussion was then possible between American and Soviet or Russian scientists on space flight data and risks (NRCP Report, 1989), on further topics for research and on what advice should be given to Governments on whether to plan missions to Mars or to the moon. The two papers

were presented at 'space' conferences, since the heavy ion results from the LBL collaboration were possibly important for space travel risks, although originally collected because of relevance to clinical topics, such as targeted treatment of intra-cranial lesions (Steinberg et al, 1990). The heavy ion results were also one of the few *in vivo* data sets then available. The two papers used the Index approach, building on an earlier, more qualitative description (Carr et al, 1987b, Section 3.6).

The first paper (Carr et al, 1993) compared data from samples exposed to different LET irradiation schedules, involving contributions for neutrons from the Hammersmith Hospital, London and for neon ions and various aspects of data interpretation from three scientists originally from LBL. T. Hayes was still on staff there, A. Nelson by then worked in his own scientific company and J. Ainsworth had moved to the Armed Forces Radiobiology Research Institute, Bethesda, where he was Scientific Director from 1989–1998. The data were compared directly and also entered into a three-level Morphological Index display, building on results published earlier before the MoI approach was developed (Carr et al, 1987b, Section 3.6). Responses included neon ion-induced collared crypts (Carr et al, 1990, Section 3.4). Ultrastructural analysis revealed more severe changes after neon ion irradiation in epithelium, fibroblasts, endothelium and at the epithelial/stromal interface. Nerve tissue responded only after neon ions, this time in Meissner's plexus profiles. Bringing in the neutron and X irradiation, it seemed that higher LET treatment produced more changes. This paper also compared structural responses with radiation-induced functional pathophysiologies, linking malabsorption and ulcer risk with enterocyte and goblet cell problems and changes in muscle tone with damage to Auerbach's plexus. This paper produced no available external citations of interest, although it was cited as part of an extensive literature in a chapter on radiation biology of space (Kiefer et al, 1996). The second paper (Carr et al, 1994) compared the effects of neon ions with those of heavier ion types, niobium of interest because of its LET and iron ions more relevant to space flight, like neon (NRCP Report, 1989). Review of the data showed that collared crypts were easily identified after neon and iron but not after niobium or, as reported earlier, silicon (Carr et al, 1990). Neon ions produced most damage to several cell types and niobium ions led to least damage to mitotic figures. These findings pointed to a break in the previously perceived link between rising LET and increased damage. This was confirmed by animal by animal Morphological Indices, where three out of the four niobium ion values were lower than all iron data. The results were relevant to variation across animal strains (Hanson et al, 1987) and to work on anatomical models (Atwell, 1994). The citation profile was similar to that for the 1993 paper.

The Index data display had confirmed that non-epithelial changes were important in the villous response to irradiation and it would have been interesting to collect more data on the effects of varying LET, dose and time. However, the balance of the work was moving towards the importance, post-Chernobyl, of understanding the uptake route of large, potentially radioactive particles through the gut wall (Chapter 4). The approach was still applied to irradiated intestinal tissues, but with more emphasis on individual responses (Section 3.9).

3.9 Different cell populations

In the next three papers, the Morphological Index protocol was adapted for specific projects. Other groups were also providing a comprehensive picture of the multi-factorial intestinal responses (Rubio and Jalnäs, 1996).

Brennan et al (1998), in collaboration with T. Seed of the Argonne National Laboratory, as a result of links made through the annual SEM conferences (Section 3.2), compared small intestinal responses to acute and protracted schedules 6hrs, 1d and 3d after 5Gy whole body gamma irradiation. Qualitative observations on apoptosis (Ijiri and Potten, 1990) were added to the parameters and two tables were used, one for epithelial tissue and one for the others. 'Damage' responses (Kwan and Norman, 1977) 6hrs after both schedules included mitotic arrest and fewer lamina propria cells. By 1d, the reduction in enterocytes was regarded as 'damaging', but the return of mitotic activity during the protracted schedule reflected some recovery. The 3d data showed continued lamina propria depletion for the acute schedule, but a possible move towards normality for the protracted schedule. It was concluded that 'adaptive' processes (Specian and Oliver, 1991) were present during continuous irradiation with low doses. Citations to this paper referred to general responses, such as epithelial damage, after genotoxic stress including irradiation (Morgan et al, 2014), decreased numbers of enterocytes and lamina propria cells (MacNaughton, 2000) or changes in the Paneth cell population (Porter et al, 2002).

The next paper (Ettarh et al, 2000) applied in an experimental setting the relevance of underlying pathology when planning radiotherapy (Gerard et al, 1995, Mayr et al, 1997, Morris and Powell, 1997). Mice with established small intestinal changes due to streptozotocin-induced diabetes, already studied in rodents (Rakieten et al, 1963, Mayhew

and Carson, 1989, Ettarh and Carr, 1997, Chapter 2, Section 2.5) were exposed to Xirradiation and samples collected three days later. Resin histology was used for qualitative observations and collection of Morphological Index data (Carr et al, 1991c, Section 3.8). Statistical analysis to differentiate between the effects of the two treatments showed that diabetes did not enhance radiation-induced damage and perhaps counteracted it through improved repair. The fact that the experimental procedures were carried out in London, courtesy of Imperial Cancer Research Find (ICRF), probably related to the anti-vivisection pressure that caused problems for the group about that time (Wang et al, 1996, Section 3.3). For citations, Salin et al (2001) quoted this paper (Ettarh et al, 2000) for a definition of how to identify cryptal mitotic cells, while Epperly et al (2004) cited it with respect to general radiation effects in villi and cryptal cells.

The above report on diabetic irradiated intestine (Ettarh et al, 2000) included data on three parts of the small intestine, in line with interest on proximo-distal variation (Mayhew and Carson, 1989). The approach dovetailed with that of the next paper (Brennan et al, 1999), where the control small intestine was divided into seven equal parts. Parameters were counted as before (Carr et al, 1991c, Section 3.8), but additional structures were also assessed, including intraepithelial lymphocytes, Paneth cells, apoptotic bodies and intestinal circumference. Some parameters had no inter-site differences; some decreased caudally, some increased caudally and others peaked at the middle of the intestinal length. Correlations were noted between some of the parameter numbers and villous height or crypt number. The data, although more detailed especially for non-epithelial parameters, had a similar overall pattern to earlier results from studies of three sites (Cheng and Leblond, 1974). With respect to citations for this paper, its report that proximal villi were taller than those distally was linked to the greater proximal villous fragility in Ussing chamber experiments (Inagaki et al, 2005). Foureau et al (2010), in a paper on intestinal defensins, cited it for the greater numbers of ileal Paneth cells. Randall et al (2011), in a review of gastrointestinal explant culture, used it as a source for the caudal decrease in crypts and therefore in crypt/villous ratio.

3.10 Reviews of irradiated small intestine

The descriptions of collection and interpretation of data from several components of the small intestine, highlighting its multi-parameter make-up (Sections 3.8, 3.9), featured in two reviews produced seven years apart, both summarising much of the material in this chapter of the thesis.

3.10.1 Structure and function

The first review was published after a meeting on 'Radiation and the Gastrointestinal Tract' held in Washington DC (Carr et al, 1995). This paper stemmed from the opening talk in the session on 'Diarrhoea', giving it a clinical relevance (Busch, 1990, Flickinger et al, 1990, Levi and Hodgson, 1990). It compared radiation-induced epithelial and non-epithelial effects, included an Index table for two notional radiation schedules and used backscattered SEM images (Chapter 2, Sections 2.3, 2.4) to illustrate epithelial/stromal stripping, crypt disorganisation and villous tip enterocyte discontinuities. Two-level Morphological Index displays reviewed the changes with time, dose and radiation quality (Section 3.8) and took further the linking of experimental data with functional or clinical questions (Carr et al, 1993, Section 3.8), in that response rankings were given for each parameter. For those related to diarrhoea, three were important, high ranking endocrine cells, giving humoral input and moderate ranking goblet cells or myenteric nerve plexus profiles providing lubrication and input to peristalsis. This paper was listed only by Google Scholar. It was cited as the introductory reference for intestinal cellular injury in a paper on selenium compounds as radioprotectants (Kunwar et al, 2011) and also for its comparison of morphological intestinal changes produced by whole and partial body irradiation, by Freeman et al (2001), a paper which included W.K. MacNaughton in the author list. Lebrun et al (1998, including Nina Griffiths in the authorship), noticed its interest in linking radiation-induced diarrhoea with alterations in regulatory systems, while Griffiths (1999) commented on its reporting of changes in nerve or endocrine cell populations. Picard et al (2002, also including Griffiths), in their introductory comments on the diarrhoea or constipation occurring as side-effects of abdominal radiotherapy or treatment with 5-OHtryptamine antagonists, noted the epithelial effects described.

3.10.2 Comprehensive review

The linking of structural changes and functional consequences continued in this Chapter's main review (Carr, 2001), which drew attention to topics of interest with illustrations, some from other groups. The addresses for this paper were Queen's University of Belfast *and* the MRC Radiation and Genome Stability Unit, Harwell, due to a two year sabbatical there. The account given here of the contents of this review has been selective, due to space constraints.

Part I reviewed the literature back to an early report of gut radiation injury (Walsh, 1897). Part II described the intestinal wall and epithelial responses to challenge, including multiple organ dysfunction syndrome, possibly associated with bacterial translocation (Niewenhuizen et al, 1996) and thus looking ahead to Chapter 4. Part II continued with a brief account of radiation, including mention of a 'new golden age' (Alpen, 1998) and the need for more 'translational research' (Coleman and Harris, 1998).

Part III on the effects of a single dose of partial body, low LET irradiation took most information from early time-points. Animal-based observations (Vigneulle et al, 1990) included a report of abnormalities associated with malfunctions of the alimentary system (Summers et al, 1991). Multiparameter results often led to Scoring Systems. One type used Injury Scores or Radiation Injury Scores (RIS), which increased with time, as damage became more obvious (Hauer-Jensen 1983a, Rubio and Jalnas, 1996): other factors were also considered by Hauer-Jensen's group (Richter et al, 1997, Wang et al, 1999). The second type calculated Ratio Scores from irradiated and control samples, as in the Morphological Index (Carr et al, 1991c, Section 3.8): these also increased initially with time, given the treatment of the data as absolute numbers. Another group compared SEM scores with data from contrast medium in serum and urine (Solheim et al 1991). Part III then considered changes in intestinal compartments (Becciolini et al, 1982) and epithelial cells (Wiernik and Plant, 1971 onwards). Stromal involvement (Rubio and Jalnäs, 1996, Wang et al, 1998, Section 3.6) included changes in fibres and fibrogenesis, the inflammatory and immune systems and vascular components (Chomette et al, 1977). The next set of topics involved changes in endocrine, neural and muscle components (Sprügel et al, 1977, Wyatt et al, 1987, Section 3.6, Summers et al, 1989, Geisinger et al, 1990, Carr et al, 1992a, Section 3.8, Höckerfelt et al, 2000).

Part IV on the effects of a single dose of unshielded low LET irradiation described functional effects (MacNaughton et al, 1997) and linked them to structural responses (Gunter-Smith, 1989). There were more data on gut associated lymphoid tissue (GALT, Chen et al, 1995) and endocrine cell responses (Linard et al, 1997, Dublineau et al, 1998, Lehy et al, 1998). Part V on the effects of a fractionated dose of low LET radiation noted that much of the information related to human radiotherapy schedules, often with little detailed treatment information provided. Side effects included almost immediate radiation enteritis (Classen et al, 1998), early diarrhoea, later bleeding and obstruction or fistula (Spitzer, 1995), confirmed in animal studies (Allgood et al, 1996, including M. Hauer-Jensen). Cellular changes included some early severe epithelial atypia (Wartiovaara and

Tarpila, 1977), continuing later, when there was also ulceration (Hauer-Jensen et al, 1983b). There were vascular (Kallfass et al, 1996) and neuroendocrine changes (Höckerfelt et al, 1999). Tumours also occurred (Ming and Goldman, 1998).

Part VI, on factors affecting the outcome, covered several topics which had not fitted comfortably in other sections. These included radiation volume (Trott et al, 1995) or type (Alpen et al, 1980) and ingested radioactive material (Harrison and Stather, 1996, Hunt 1998), dealt with in Chapter 4: also touched on were pathophysiology (Gerard et al, 1995) and the treatment of radiation-induced symptoms (Classen et al, 1998). The section finished with a summary of the radiation-induced changes.

The conclusions section of the review highlighted the need for more data: on different irradiation schedules; on interfaces, composite structures and inter-compartment relationships; on cross-talk, signalling molecules and 'bystander' input; and on underlying factors such as stress, diet, aging or pathophysiology.

Citations to Carr (2001) included one from Hauer-Jensen's group, linking it with radiationinduced changes in 'most intestinal wall compartments' (Wang and Hauer-Jensen, 2003). These and further aspects of the responses were cited by other groups (Parihar et al, 2006, Prabhakar et al, 2007, Williams et al, 2010, including Hauer-Jensen, Kanimozhi et al, 2012, Shanthakumar et al, 2012). It was also used as a reference for the time after treatment (Ropenga et al, 2004, with N. Griffiths) or for the involvement of reactive oxygen species (Senesse et al, 2009). There were also clinical papers, relating to colorectal cancer emergencies (Barnett et al (2013) or to intestinal effects after liver irradiation (Cameron et al, 2012). It was also used by Ishihara et al (2011) as a first general reference in the Introduction of their paper on the effect of anabolic steroids in post-irradiation regeneration, while Veerginadis et al (2017) used it as a citation to fibrosis. More recently, Hauer-Jensen et al (2014) referred the reader to it as a 'very comprehensive review' in their section on clinico-pathological aspects of radiation damage to the gut.

3.11 Context and comment

The theme of this Chapter was the application of SEM and associated correlative approaches from Chapter 2 to a specific topic, as tissues, mainly intestinal, responded to external challenge such as irradiation. The work was carried out in the Universities of Glasgow and Belfast, while visiting the Lawrence Berkeley Laboratory (LBL) and on sabbatical at the MRC Radiation and Genome Stability Unit, Harwell. The period involved covered slightly under thirty years, about the same span as Chapters 1 and 2, with publication dates from 1972 to 2001.

With respect to article destination, there were fewer articles in anatomical journals, apart from Scanning Electron Microscopy and the Journal of Submicroscopic Cytology and Pathology. There were several publications in radiation and radiobiology journals and a few in general journals, review series or conference proceedings.

Collaborations continued to be important: some begun as internal to Glasgow University remained in place. Collaborating scientists and groups in other Departments and various branches of radiation science included, in Glasgow, Pharmacology, Pathology, The Glasgow Institute of Radiotherapeutics and Oncology/Radiology Research Group, Belvidere Hospital, The Glasgow Institute of Radiotherapy and Oncology, The Western Infirmary. Groups in Belfast included the School of Clinical Dentistry/Dental Surgery, the School of Clinical Medicine/Oncology. Collaborations external to the 'home' Institution at the time included in London, the MRC Cyclotron Unit, The Hammersmith Hospital and the Imperial Cancer Research Fund (ICRF); and in the USA, the Centre for Bioengineering, University of Washington, Seattle, the Radiation Haematology Group, the Argonne National Laboratory, Argonne, Illinois, USA, and LBL, California, USA.

Some colleagues or graduate students continued to contribute after moving to other centres. These included Anatomy, University College, Dublin (UCD); Anatomy, University of Sri Lanka Peradeniya Campus; The Armed Forces Radiobiology Research Institute (AFRRI), Bethesda, Maryland, US; the CRC Normal Tissue Radiobiology Research Group, the Research Institute, The Churchill Hospital, Oxford; Diagnostic Imaging, UCD; Life Sciences, Kingston; NeoPath, Bellevue, Washington, USA; the Richard DimblebyDepartment of Cancer Research, St Thomas's Hospital Medical School, London.

A special factor needing mentioned was anti-vivisection pressure in Belfast. This was distracting for both staff and students. However, radiation and associated experimental facilities were kindly provided by other laboratories, namely Imperial Cancer Research Fund (ICRF) in London and Beijing Medical University in China.

The roles and contributions involved were still obvious from the order of the authors in the submissions, again reflecting the change from 'hands-on' scanning and transmission electron microscopist, to one of encouraging others to use the SEM, especially students. As before, as the years went on, more of the data were collected by them, but there was
continuing involvement in thesis supervision, protocol design, the provision of technical training, data quality control and paper writing.

With respect to links to the next Chapter, the above output in the overlapping fields of radiation, morphology and the intestine led to invitations after the Chernobyl accident to attend workshops on its likely impact and to funding for research on large particle uptake, with a resulting gradual switch of effort, in the period from 1986 to 2001, away from radiation effects.

The work by then had come full circle, beginning with the study in Chapter 1 of some aspects of the small intestine, following on with the need to produce correlative images to relate to the additional detail of mucosal surfaces produced by SEM, then to the need to explain villous collapse of these surfaces after radiation and tie it in to internal changes. The full circle returned interest to the exploration of the small intestine after a new, but cognate, environmental challenge, that of exposure to large particles (Chapter 4), but with the knowledge that as a multi-parameter organ, all of its compartments and their constituent cells or structures were of possible relevance.

Chapter 4 Microparticles crossing intestinal surfaces

Preface - extract from Presubmission Abstract (Chapter 4)

Relating surface effects with morphological counts per circumference was useful in work on microparticle uptake, begun soon after the Chernobyl accident in 1986, the continuity with Chapter 3 being highlighted in a conference chapter. The pharmaceutical and medical relevance of uptake was also noted and the timing of the writing of a review of the morphological aspects underlined its environmental importance, given the 2011 Fukushima incident.

The post-Chernobyl work addressed two questions. Firstly, could particles the size of those released into the radioactive cloud enter the human body through an intestinal route? Secondly, were any population groups particularly at risk, specifically pregnant or lactating women? Two models were used. In vivo, human exposure was simulated by exposing rodent intestine in situ rather than exteriorised: particles were usually $2-20 \mu m$ fluorescent latex, identified after maceration or by confocal scanning laser and electron microscopy. The in vitro model comprised Caco-2 monolayers, allowing recording of transepithelial resistance and particle uptake, which was in the same range as that in the other model. M cell monolayers have also been used.

The first answer was that, after in vivo in situ exposures as short as 5 minutes, particles did cross the small intestinal epithelium, moving on to secondary organs such as lymph nodes: particle dose, size and volume were important. Entry was, unexpectedly, mostly through villous epithelium and not M cells of lymphocyte-rich Peyer's patches, with uptake not greatly reduced by severe compromised immunodeficiency (SCID). Particles were observed at intercellular junctions or spaces, highlighting enterocyte tight junctions as possibly involved. In vitro work addressed this, showing that ethanol loosened junctions and increased uptake, while cooling produced opposite effects and different changes in immunolabelled junctional proteins. Particle retention at intercellular junctions affected the protein disturbances.

The second answer was that in vivo in situ uptake: was not increased by diabetes; was more affected by age than species and sex; and could be higher in late pregnancy and early lactation. In vitro uptake could be increased by irradiation. The Chapter finishes with further reference to the 2012 review.

4.1 Questions on uptake and links to Chapter 3

As indicated above, involvement in this area of research stemmed from previous work on radiation effects on small intestine (Chapter 3). This was relevant to the topic at the centre of this Chapter, namely the exploration of the impact of the 1986 Chernobyl incident on humans and their food chain, since the radioactive microparticles produced could have used an intestinal route to enter body tissues of exposed populations, according to earlier publications (Volkheimer et al, 1965 and later, e.g. Volkheimer, 1977, 2001). Although no written record has been retained of this, memory has indicated that the grant body initiating and supporting the programme of work on microparticle uptake, the UK Department of Health, London, set two questions.

Question 1 'Do microparticles enter tissues by an intestinal route ?'

Question 2 'Are any groups 'at risk', such as pregnant and lactating females ?'

Initial consideration of these two questions led to the grant-holder and thesis author posing a **supplementary methodological question** – 'What model, what microparticles, how locate them, what animal or cells exposed ?' The answer to this question was the construction of two models, one with rodent small intestine *in vivo in situ* (Model 1, Sections 4.2 - 4.4, 4.7) and the other with an *in vitro* culture of a monolayer resembling a sheet of absorptive intestinal epithelial cells (Beaulieu, 1999, Model 2, Sections 4.5 - 4.7). The first was used mainly by what could be called a 'Belfast particle uptake group', while the second, later one was centred in Oxford. The models have been described first, while data from them have mainly been reported in relationship to Questions 1 and 2.

The subject of microparticle uptake has been introduced by relating it to three meetings after Chernobyl.

The first publication (Carr et al, 1997) stemmed from a meeting in Rome on recent advances in microscopy of cells, tissues and organ, organised by P. Motta. The paper related surface responses after an environmental challenge to changes in the underlying tissues, through morphological analysis, as used above (Chapter 3, Section 3.8). It dealt with radiation effects (Buell and Harding, 1989, Potten, 1990, Carr et al, 1991c, Chapter 3, Section 3.8), taking further the approach (Chapter 3, Section 3.10) of matching changes in function with total responses over the whole radiation programme. It also quantified microparticle uptake (Kreuter, 1991, Couvreur and Puisieux, 1993), using a Morphological Index method: numbers in different compartments and cell types were recorded per circumference, as for

radiation-induced changes in crypt/microcolony numbers (Withers and Elkind, 1970). Sampling for particle counting covered nine proximo-distal segments. The conference was in honour of Marcello Malpighi, a 17th century Italian biologist and physician, regarded as the father of microscopical anatomy and histology. Authors were asked to mention him in their presentations: the references quoted in the paper by Carr et al (1997), have been used in Chapter 5 (Section 5.3) for a different purpose (Malpighi, 1687). The paper had almost no citation record and was included to illustrate the link between Chapters 3 and 4.

The second meeting, concentrating on particle uptake and translocation across epithelial membranes, was the central Symposium at the 1995 summer meeting of the Anatomical Society, in Belfast, hosted by the thesis author. The abstracts were published in the Journal of Anatomy, as were many of the full papers in 1996, after peer review, producing a total of 14 articles, including a one-page Introduction written by the organising committee. This contained representatives of gut biology, pharmacy, surgery and toxicology, reflecting the relevance of uptake to a range of subjects (Carr et al, 1996b). This short 'paper' noted: variation in methodology and data; villous epithelium and also Peyer's patches, important parts of gut-associated lymphoid tissue (GALT); factors influencing uptake; and the sensitivity of some population subsets. It also had almost no citation record and was included to illustrate the relationship of the Belfast particle uptake group's work with that of others in the field. However, it was cited as one of an introductory group of three papers in a 'meta-review' by Hussain et al (2001): the authors included S. Florence, one of the main authorities on uptake in the field and one of the co-organisers of the Belfast Symposium.

The third meeting, held in London, was entirely on microparticle uptake. The papers were published in the Journal of Drug Targeting and included Hodges et al (1995a), which summarised the literature (O'Hagan, 1990, Kreuter, 1991, Couvreur and Puisieux, 1993) and described its own two approaches to particle quantification: one used 'maceration' to produce solubilised tissues (LeFevre et al, 1989, Ebel, 1990) and the other specific methods for microscopy. Figures and data Tables of small intestine 30mins after oral administration of 2μ m latex microparticles included a confocal image of particles in the lumen and villous stroma, along with numbers per circumference in several villous and Peyer's patch compartments and also per gram of macerated tissue. More particles were associated with nearby villi than with the actual Peyer's patches and there was discussion of the interpretation of the data for the two techniques. The topics cited from this paper included the uptake of particulate material (Hillyer and Albrecht, 2001) and the lack of clear understanding of the uptake route of this process (Tabata et al, 1996), of such importance

to drug delivery. Several groups quoted it as having reported uptake through villous sites near Peyer's patches (Pietzonka et al, 2002b) or at enterocytes (Damge et al, 1996, Le Visage et al, 2001). Florence (1997) quoted it in a review as a source for percentage uptake.

Citation histories from now on have been mostly separated into the minority relating to techniques and the majority relevant to data: citations have not been given for all submissions and only some have them in both categories.

4.2 What model? Model 1, the standard in vivo in situ model

More information on this model was given in the next paper, published in a mainstream gut journal (Hodges et al, 1995b). As above, the baseline in vivo procedure involved oral exposure of rats to microparticles, with the intestine *in situ* and not as exteriorised loops. Particles were often 2 µm fluorescent latex. Mesenteric lymph nodes or other particle-poor regions were collected first, to avoid cross-contamination. Maceration counting on unfixed tissue (LeFevre et al, 1989, Ebel, 1990) was carried out on nodes and on proximal, middle and distal small intestinal 'Peyer's patch regions', which included not only domes but also adjacent villi. Epifluorescence and confocal microscopy were used for morphological analysis of full circumference fixed cryomicrotomy slices, from the proximal ninth only: these were termed 'Peyer's patch circumferences' and included domes and villi around the entire intestinal wall. The microscopy technique was more accurate than maceration, since it permitted identification and exclusion of luminal or surface particles and therefore allowed accurate quantification: this point has been touched on again below (Sections 4.8.1, 4.9.2). One of the citations to technique quoted this paper: it described the histology approach as only suitable as a semi-quantitative assay, although stating that 'precious information' was gathered by it (Delie, 1998). She pointed out the disadvantages of the technique and seemed to feel that quantitation through maceration was preferable (but see Sections 4.8.1, 4.9.2).

4.3 Model 1, variation in particle type

The principal particle type in the *in vivo in situ* model was, as stated above, fluorescent latex, often 2µm in diameter (Hodges et al, 1995a, 1995b, Sections 4.1, 4.2). Starch had also been used (Volkheimer and Schulz, 1968, Volkheimer et al, 1968a, b), but produced artefactual problems. Pilot experiments (McCullough et al, 1995) were also carried out, administering iron (Volkheimer et al, 1969, Parmley et al, 1984) as powder in milk and using light microscopy, backscattered detector electron imaging (BEI) with scanning

electron microscopy (SEM, Chapter 2, Sections 2.3, 2.4), transmission electron microscopy (TEM, Chapter 1) and X-ray microanalysis (XRMA, Chapter 2, Section 2.4). The citations related to this paper included a detailed description by Nagata (2004) of the techniques used.

There was another and serendipitous observation of 'particle' uptake (Cartwright-Shamoon et al, 1995), during investigation of the yeast Saccharomyces boulardii (Maupus et al, 1983, Vidon et al, 1986) as an anti-diarrhoeal agent, in collaboration with the Belfast Department of Child Health, headed by J. Dodge. Yeast uptake was studied with light microscopy, secondary detector SEM imaging and BEI SEM images. With respect to technical citations, Samonis et al (2011) referred to it as showing the yeast taken up in animal experiments as opposed to therapeutically, giving details of its protocol.

The final particle type (Carr et al, 2001) was fused aluminosilicate (FAPs), 2.2µm in diameter, able to be labelled with ⁵⁷Co to make its effects more like those of nuclear incidents such as that at Chernobyl (Cuddihy et al, 1989, Wakeford 2011a). This was carried out in collaboration with C. Collier and also J. Harrison and M. Youngman of the then Atomic Energy Authority (AEA) Technology and National Radiological Protection Board respectively, both at Harwell. The FAPs were visualised in resin sections of rat intestine by BEI and by TEM of freeze-fracture replicas. The techniques, however, were time-consuming and Model 1 with fluorescent latex remained the standard approach. This submission had no citation history, being in Abstract format only.

4.4 Model 1, developments in methodology

The *in vivo in situ* model was established by the time the paper by Hodges et al (1995b, Section 4.2) was published. Microparticle exposure time was usually between 30mins and 24hrs. Various technical changes were introduced, described in brief here, in approximate chronological order.

More detailed study of onward particle movement to secondary organs included microscopical analysis of mesenteric lymph nodes (MLNs, Hazzard et al, 1996), liver and spleen (McMinn et al, 1996). The percentage of administered dose was calculated for particles of different size, giving consideration also to the volume and surface area of particles taken up as well as to the numbers involved (Carr et al, 1996c). Citations to this paper on particle size included some to techniques, such as to the maceration protocol (Delie, 1998).

TEM images were used to explore the site of small intestinal uptake of yeast in mice, some infected with rotavirus to induce diarrhoea, (Cartwright-Shamoon et al, 1996). It was also applied to locate latex microparticles, in Model 1 routine ultrathin sections as well as those pre-selected by vibrotome sectioning of regions known to be particle-rich (Smyth et al, 2005). This paper explored the question of whether pregnant and nursing mothers were more at risk, by comparing uptake into male, virgin female, pregnant and lactating rats 5 or 30mins after particle administration. Other developments in the same paper included the use of gavage rather than oral/pharyngeal administration, the collection of particle numbers in cardiac blood and the examination of nine proximo-distal segments and Peyer's and non-Peyer's circumferences. The range of intestinal diameters in animals of different species or ages led to the recording of intestinal lengths and areas, giving stacked profile displays of 'area under the curve' numbers for luminal, surface and tissue particles down the intestinal length (Doyle-McCullough et al, 2007). Proximo-distal particle distribution was also shown for sites with different lymphoid profiles in a study of immune status (Smyth et al, 2008). Several aspects of the technique were also validated (Smyth et al, 2005, Doyle-McCullough et al, 2007).

4.5 What model? Model 2, the standard in vitro model

A major question had been identified (Hodges et al, 1995a, 1995b, Sections 4.1, 4.2) as to whether uptake was principally through villous epithelium (McClean et al, 1998, Hillyer and Albrecht, 2001) or through Peyer's patch microfold (M) cells (Ermak et al, 1995, Thomas et al, 1996, Beier and Gebert, 1998). Villous uptake implicated passage at tight junctions (Björk et al, 1995, McCullough et al, 1995, Section 4.3, Wiesner et al, 2002), which have attracted attention for their many roles (Aijaz et al, 2006). An *in vitro* protocol (Moyes et al, 2007) was constructed to explore the role of tight junctions in uptake, becoming more 'blue sky' than applied research. The aim was to construct *in vitro* an uptake model that simulated the *in vivo in situ* particle uptake at villous epithelium.

The choice for Model 2 of Caco-2 cells, derived from a human **co**lonic adeno**ca**rcinoma (Delie and Rubas, 1997) was related to their resemblance to a small intestinal enterocyte sheet (Beaulieu, 1999). Particles were usually but not exclusively fluorescent latex. The Caco-2 monolayers were grown on Transwell[®] plates and the inserts had 3μ m pores, allowing passage into the lower well of 2 μ m microparticles. Recording was made of transepithelial resistance (TER), reflecting permeability and the extent of closure of the tight junctions (Madara, 1998). Fluorescence microscopy was used for particle counting of a

'crosswire' across the middle of the epithelial monolayer. Confocal scanning laser microscopy images gave measurements for cell dimensions and estimated the proportions of particles associated with or sitting on the epithelial monolayer. Particle location was noted, either epithelial-associated or submembranous (EAPs, SMPs respectively). Any monolayers with discontinuities were discarded.

Objectives of the first *in vitro* paper (Moyes et al, 2007) were: to explore whether uptake was accompanied by permeability changes or affected by the removal of apical particles, mimicking peristalsis-driven onward movement of an intestinal bolus; to study the effect of the vasoactive intestinal peptide/polypeptide (VIP); and to compare the uptake data with those for *in vivo in situ* experiments (Sections 4.2, 4.4). With respect to technical citations, reference was made to its use of TER (Uematsu et al, 2016) and to the usefulness of Caco-2 cells for the assessment of uptake as for drug delivery (Lyu and Park, 2008, Derakhshandeh et al, 2011, Iannuccelli et al, 2011).

4.6 Model 2, developments in methodology

As was the case for Model 1 (Sections 4.2 - 4.4), there were methodological developments over the years, including further studies of culture conditions (Moyes et al, 2010a), a topic of interest to several groups (Hidalgo, 1996, Delie, 1998, Ranaldi et al, 2003, Zucco et al, 2005, Volpe, 2008). The developments included changes in cell sourcing and exposure to 'pre-treatments', to connective tissue cells found deep to the epithelium *in vivo* or to irradiation.

With respect to the two pre-treatments (Moyes et al, 2010a, Moyes et al, 2011), these were ethanol and temperature variations associated with the incubator (37°), culture hood (24°) or chilling (0°). Both pre-treatments became useful as comparator procedures for particle addition in identifying changes in junctional proteins. The technique-related citation record for the first pre-treatment paper (Moyes et al, 2010a) included references to it for the study of Caco-2 cells with TER (Yuan et al, 2011) or their use as a model of the intestinal epithelium (Li and Zhu, 2016).

With respect to the combined effects of external irradiation (Chapter 3) and particle exposure (Moyes et al, 2008), this was carried out during an attachment to the MRC Radiation and Genome Stability Unit (RAGSU) at Harwell. The collaborating team there included M. Kadhim (radiobiologist) and M. Hill (physicist). The protocol used 2Gy X-rays (de Carvalho et al, 2006), at the level of one fraction in human radiotherapy

(International Commission on Radiological Protection, ICRP, 1990, published 1991), with or without subsequent particle exposure. The importance of sham treatment (Chandler et al, 1993, Oude Elferink et al, 1993, Armitage et al, 1994) was explored to control for the different temperature of the irradiation room (Moyes et al, 2008). Derived data assessed included cellular surface area and the volume of cells or intercellular spaces (Lieb et al, 1977, Porvaznik, 1979, Somosy et al, 1993, de Carvalho et al, 2006).

With respect to exposure to connective tissue cells, THP-1 derived macrophages (Manabe et al, 2003, Satsu et al, 2006) were used: these were known to open Caco-2 cell tight junctions (Kanzato et al, 2001) and to phagocytose particles (Hallab and Jacobs, 2009), moving them away from the epithelium (Wells et al, 1988). In Moyes et al (2010b), the two cell types were mutually exposed for 24hrs in indirect contact through the medium, more 'compartmentalised' culture (Parlesak et al, 2004) than 'co-culture'. Some technical citations for this paper were to the Caco-2 model itself: for instance, Stevanović et al (2012) quoted it for Caco-2 cells being useful for testing the efficiency of drug delivery.

Alternatively, Raji lymphocytes in the lower well produced monolayers of M cells (Kernéis et al, 1997, Gullberg et al, 2000, Tyrer et al, 2002, des Rieux et al, 2007, Miyazawa et al, 2010), as found in Peyer's patch epithelium (Carr, personal communication, 2018). These were also produced through a compartmentalised culture, since a true 'co-culture' contained gaps that distorted the data (Moyes et al, 2009). Uptake was compared with that in the Caco-2 model (above), which simulated a sheet of villous enterocytes, testing claims that it was greater across M cells (van der Lubben et al, 2002, Lai and D'Souza, 2008).

Developments in assays included adjustments to the protocols for both pre- and postfixation stages. Pre-fixation developments included the addition of the measurement of cytokine levels in the medium (Moyes et al, 2010b). TER, which provided information on permeability (Delie and Rubas, 1997), but not epithelial integrity (Mukherjee et al, 2004) was originally reported as the recorded, 'raw data' (Delie and Rubas, 1997, Moyes et al, 2007). Other approaches involved the comparison of two values, either through subtraction, giving 'delta' values (Moyes et al, 2010a, b) or through ratios, producing percentages (Moyes et al, 2011).

Post-fixation developments relating to cell dimensions (Moyes et al, 2007, Moyes et al, 2008) included the production of diagrammatic representation of cell shapes. Junctional protein (JP) immunolocalisation at intercellular junctions, put in context by their relationships to actin (Shaw et al, 2005), was developed (Shi et al, 1993, Ferruzza et al,

1999, Musch et al, 2006) to produce uniform localisation of the proteins over a wide area of control monolayers, allowing comparison with treated samples (Moyes et al, 2010b, 2011). Tight junction (TJ) proteins occludin and zonula occludens (ZO-1) in belt format met these criteria, as did adhering junction (AJ) protein E cadherin, found as belts and spots.

Other post-fixation developments involved particle counting, initially mostly through epithelial-associated particle (EAPS, Moyes et al, 2007, 2008), then more through submembranous numbers (SMPs, Moyes et al, 2010a, 2010b, 2011) and recently through both, for Caco-2 and M cell monolayers (Carr, personal communication, 2018). There were also developments in comparisons of microparticle uptake *in vitro* and *in vivo in situ*, (Moyes et al, 2007).

4.7 What model? Invited Review on microparticle uptake

The review (Carr et al, 2012) was written around the time of the 2011 Fukushima incident (Wakeford, 2011b), underlining the field's long-running relevance (Wakeford, 2007). It contained more information on some aspects of uptake, such as Peyer's patches and their M cells (Cornes, 1965, Owen and Jones, 1974, Regoli et al, 1995) and the history of uptake and its terminology (Steffens, 1995), including 'persorption' (Volkheimer and Schulz 1968, Volkheimer et al, 1968a, b) or, given Herbst's early paper in 1844, 'the Herbst-Volkheimer effect', (Prokop, 1990, Beukeveld and Wolthers, 1995).

The review differed from those in Chapters 2 and 3 in containing proportionally more otherwise unpublished data and comment. Model 1 topics dealt with further included: particle characteristics and localisation; animal status; quality control and model validation; calculation of percentage uptake; and data interpretation. In addressing further the question of 'at risk' populations, in this case after radiation exposure following incidents like those at Chernobyl (Wakeford, 2011a), the protocol involved a dose of 10Gy whole body X-rays, given before a 30mins Model 1 exposure. For Model 2, there was further description of the formation of an M cell monolayer and of immunocytochemistry localisation with I. Dublineau of IRSN, Paris, of the effects on permeability of depleted uranium, iron and latex (Carr et al, 2012). The review also reported briefly on the hypotheses that intercellular junction proteins participated in uptake, a topic requiring further analysis (Carr, personal communication, 2018).

4.8 Q. 1 Do microparticles enter tissues by an intestinal route ?

4.8.1 In vivo in situ data (Model 1)

The answer to the first question was that, after *in vivo in situ* administration, particles did cross the small intestinal epithelium (Hodges et al, 1995a, b, Sections 4.1, 4.2). Morphological analysis showed particles in proximal small intestine, confirming their position by optical slicing to be in and not artefactually on the tissues. Validation checks (Smyth et al, 2005, Section 4.4) were satisfactory and particles were seen by TEM at the microvillous surface, between epithelial cells and in blood vessels. Particles were also identified in lymph nodes.

The review (Carr et al, 2012, Section 4.7) brought together values for several aspects of uptake. Mean figures for percentage administered dose were higher from maceration (1.80%) than from microscopy (0.13%), which were nearer to the even lower literature values (Ebel, 1990). Returning to the point discussed above (Section 4.2), maceration not only overestimated uptake by including luminal and tissue particles, but also underestimated it to a variable extent because of the fading of particle fluorescence after storage in fluid containing residual potassium hydroxide. There was inter-animal variation, possibly related to handling-related stress but fasting had little or no effect, unlike literature reports on this and diet (Worthington and Syrotuck, 1976, Ebel, 1990, Simon et al, 1997).

Further information below has been set out in five parts (Sections 4.8.2 to 4.8.6).

4.8.2 Uptake time-frame (Model 1)

Papers by Hodges et al (1995b, Section 4.2) and McMinn et al (1996, Section 4.4) showed that, in macerated tissue (LeFevre et al, 1989, Ebel, 1990), particles were found 30mins after administration, confirming reports of 30mins to 2hrs as significant (Kreuter et al, 1989, Jenkins et al, 1994). Microscopy showed that the proportion in the lumen fell with time from 2 to 8hrs, while that for epithelium increased from 0.5 to 4hrs: the proportion in non-epithelial tissues varied with time (Hodges et al, 1995b). With respect to available citation information, this paper was the top cited article in the thesis in Scopus. It was quoted by Bakhru et al (2013), as one of four papers from the group, being included in their Table reviewing the literature on uptake. They also quoted it with respect to uptake occurring within 30mins of administration.

Extension earlier of the time-frame showed uptake as early as 5mins after administration (Hazzard et al, 1996, Section 4.4), confirming previous reports of it taking place within a few minutes (Volkheimer and Schultz, 1968, Sass et al, 1990, Jenkins et al, 1994). With respect to citations, the Hazzard paper was quoted by Åkesson et al (2011) as describing uptake as 'rapid' and by Bakhru et al (2013) as reporting it as happening at 5mins. The 'speed of movement' of particles from lumen to serosa was up to 4 times per second the diameter of a particle, or $3-8\mu$ m/second, allowing passage through the wall within 2mins, the earliest exposure time studied (Carr et al, 2012, Section 4.7).

Extension of the last sampling time later to 5 weeks (ibid) showed a fall in total particle numbers and the proportions in villous sites, while those at Peyer's patches rose, particularly for lymphoid tissue, by then at 100%, albeit of a much smaller total number. In other words, the earlier the more villous, the later the more Peyer's, confirming that the predominant villous uptake was partly due to the comparatively early time-points originally sampled.

Also relevant, however, were data on transit (Carr et al, 2012), provided by J. Harrison (then National Radiological Protection Board, NRPB, later Health Protection Agency, Harwell) and C. Collier (then AEA, Harwell): these showed that at the often used 30mins time-point, 89% of 10µm particles were still in the oesophagus or stomach and only 6% had reached the small intestine. This showed that quoting uptake figures for only one time-point could not have revealed the whole amount taken up over time.

4.8.3 Uptake sites, proximo-distal and villous/Peyer's patches (Model 1)

For proximo-distal distribution (Hodges et al, 1995b, Section 4.2), 30 mins after exposure, substantial numbers were seen from segment 1 through to segment 5, with more at segment 3 and fewer distally (Carr et al, 2012, Section 4.7). Uptake of fused aluminosilicate particles (Carr et al, 2001, Section 4.3) occurred even more proximally and to a lesser extent than for latex. The paper by Hodges et al (1995b) was quoted as reporting uptake to be proximal (Hussain et al, 2001, Bakhru et al, 2013).

For the primary uptake site as in villous or Peyer's patches, microscopy quantification of numbers per proximal circumference (Hodges et al, 1995a, Section 4.1, 1995b) recorded that most was villous (Wells et al, 1988), although several groups had reported that Peyer's patches were the key sites (LeFevre et al, 1985, Owen et al, 1986, Pappo and Ermak, 1989, Eldridge et al, 1990, Pappo et al, 1991, Jepson et al, 1993). Although the observed predominance of villous uptake was pointed out clearly (Hodges et al, 1995b), attention was

distracted from this by the use of the terms 'Peyer's patch region' and 'Peyer's patch circumference' and several groups have misquoted the results: the details of this have not been included here. Citations relevant to the importance of villous uptake included one to the Hodges et al paper (1995b) by Norris et al (1998) which included a definition of the term 'Peyer's patch rings' and a reproduction of one of the Tables. Other references to this paper describing uptake to be heavily villous or involving enterocytes included one by Lwin et al (2009) and a review by Snoeck et al (2005). Reineke et al (2013), in their Introduction, mentioned the paper on uptake and immune status (Smyth et al, 2008, Section 4.4) and the Carr et al (2012) review as identifying non-FAE tissue to be a site of uptake, commenting that, since M cells only made up a very small proportion of the surface cells, targeting them was not easy.

Confirmation of this important predominantly villous site in the *in vivo in situ* model within 30mins of microparticle administration came from three sets of experiments.

Firstly, there was no difference in uptake between Peyer's and non-Peyer's circumference rings (Smyth et al, 2005, Section 4.4). Secondly, Peyer's patches were not essential for uptake, since mice with virtually none of these lymphoid aggregations did not have reduced levels of uptake, as shown through collaborative work with U. Schumacher's group in Hamburg (Smyth et al, 2008). This studied microparticle exposure in combined immunodeficient (SCID) mice, known to take up micro-organisms (Ohsugi et al, 1996, Havell et al, 1999, Mutwiri et al, 2001) and parasites (Mead et al, 1991, Koudela et al, 1999, Umemiya et al, 2005). BALBc and SCID particle uptake were respectively \geq 90% and 100% villous. The percentage of administered dose taken up at 30mins was 0.27% for BALBc mice, similar to those in the literature (Ebel, 1990, Jenkins et al, 1994): the corresponding value for SCID mice was not significantly different.

Thirdly, results from an inter-laboratory study (Carr et al, 2012) showed that uptake in entirely villous samples was observed by three groups. One was headed by T. Dayan, at St Bartholomew's Hospital, another by J. Harrison, at the then National Radiological Protection Board (NRPB), Harwell and the third by the thesis author. All three laboratories detected substantial particle numbers in non-Peyer's sites: data courtesy of Dayan, Simon, Harrison and Carr. The Belfast group also showed by microscopy that 93% of the tissue particles in proximal segments were villous.

4.8.4 Uptake pathway, paracellular/apical (Model 1)

Information on this has come from work on four particle types, all relating to villous locations.

Firstly, TEM showed latex particles within intercellular spaces (Smyth et al, 2005, Section 4.4), implying a paracellular pathway, but not excluding apical/lateral access. Secondly, iron deposits (McCullough et al, 1995, Section 4.3) were seen at microvilli and intercellular boundaries, confirming apical uptake of metallic substances (Lever and Duncumb, 1961, Volkheimer et al, 1969, Rehnberg et al, 1981, Nwokolo et al, 1992). The identification of iron near tight junctions was an early indicator of their possible involvement. The deposits were patchy, perhaps because of variation in either iron receptors or nutritional status (Parmley et al, 1984). With respect to citations, the data were quoted by Nagata (2004) as implicating both intracellular and paracellular pathways, a conclusion reached also by Löbenberg et al (1997), which included J. Kreuter as one of the authors. Other groups cited it more specifically as having pointed to the paracellular pathway, (Srinivasu et al, 2015) and/or to particle access through tight junctions (Hillyer and Albrecht, 2001, Zhang et al, 2003, Rohner et al, 2007).

Thirdly, freeze-fracture TEM data from preliminary FAP studies (Carr et al, 2001, Section 4.3) also pointed to villous tight junctions as possibly involved in uptake, supporting paracellular passage. As indicated above, this submission had no citations. Fourthly, Cartwright-Shamoon et al (1995, Section 4.3) showed yeast 'particles' of Saccharomyces boulardii (Maupus et al, 1983, Toothaker and Elmer, 1984, Vidon et al, 1986) within the villous mucosal epithelial layer, implying translocation as suggested by Stone et al (1974), but not specifying whether uptake was paracellular. TEM (Cartwright-Shamoon et al, 1996, Section 4.4) showed the yeast between and to a lesser extent within the epithelial cells. This suggested that the more important pathway was paracellular, implicating tight junctions again. Samonis et al (2011) cited the 1995 paper as evidence for the yeast travelling from the intestinal lumen through the mucosa. The 1996 paper was quoted as reporting yeast uptake in a review by Lewis and Freedman (1998), by Marteau and Shanahan (2003) and Llopis et al (2014) and also with respect to identifying it as a possible risk factor, especially for immune-suppressed individuals (McCullough, M.J. et al, 1998).

4.8.5 Onward movement (Model 1)

With respect to secondary organs (Section 4.4), small numbers were seen in mesenteric lymph nodes as early as 5mins after administration (Hazzard et al, 1996), supporting previous reports of onward movement (Jani et al, 1989, O'Hagan, 1990): various sites were involved. Particles were also found to a lesser extent at 30mins in liver and spleen (McMinn et al, 1996) and not in cardiac blood until that time-point (Smyth et al, 2005). With respect to citations, Snoeck et al (2005), in their review of the role of enterocytes in the intestinal barrier, quoted Hazzard et al (1996) for the onward movement of particles to mesenteric lymph nodes 5 to 30 minutes after exposure. Yang et al (1999) and Akande et al (2010) cited McMinn et al (1996) for movement to secondary organs.

4.8.6 Particle-associated factors (Model 1)

These included particle administration, size, aggregation or tissue loading. With respect to administration, as reported elsewhere (Ebel, 1990, Seifert et al, 1996), dilution in water of the particle dose decreased uptake (Carr et al, 2012, Section 4.7), but the use of milk (ibid) as the diluent decreased rather than increased uptake (Le Ray et al, 1994): the lack of agreement was possibly associated with protocol differences.

With respect to size, (LeFevre et al, 1980, Ebel, 1990, Eldridge et al, 1990, Jenkins et al, 1994, Florence al, 1995), the cut-off depended on assay technique (Carr et al, 1996c, Section 4.4) but more was taken up of the smaller latex particles, which also penetrated further than larger ones. However, consideration of their dimensions showed that a diameter of 6µm could be the most efficient for delivery to lymph nodes, a subject of interest to pharmaceutical specialists, among others. With respect to citations, it was quoted with respect to uptake at 'non-lymphoid intestinal tissue' (e.g. Behrens et al, 2002) and used as a source for uptake level being < 1% (Møller et al, 2008). Bakhru et al (2013) quoted it for greater numbers of smaller particles being taken up, with another inclusion in their Table of the group's work, through this paper, as for Hodges et al, 1995b (Section 4.2). Uddin et al (2009) also noted its report that, when volume was considered, the next larger size (5 to 6μ m) was more efficient for onward movement to nodes. Hussain et al (2001) quoted it with respect to loading capacity. Size was also important for iron uptake (McCullough et al, 1995, Section 4.3), since the particles were described as $6 - 9\mu m$ in diameter, but some deposits were so small that they were 'nanoparticles' and less suitable than latex for microparticle uptake studies.

With respect to aggregation, standard latex particles were seen singly, in superficial linear arrays and in deeper clusters (Carr et al, 2012). For tissue loading, in TEM samples there was 1 particle/1000 cells and even in heavily laden sections harvested by vibrotome slicing, the ratio only rose to 1/50 cells. The imbalance between uptake and cell numbers was also important, goblet cells having higher uptake but fewer cells than enterocytes. Aggregation and loading could affect microdosimetry and tissue damage, with possible impact on the pathophysiology of the response to particle exposure, particularly for larger particles, which carried more dose per unit particle (Carr et al, 1996c). Tissue damage (Fabian, 1983, Freedman, 1991) was confirmed as occurring at fractures where particles were located, especially when aggregation was present.

4.8.7 Particle-free in vitro monolayers (Model 2)

Laboratory conditions were standardised to define control Caco-2 monolayer parameters (Moyes et al, 2007, 2010a, Sections 4.5, 4.6) for comparison with other reports (Delie and Rubas, 1997, Sambuy et al, 2005, Hughes et al, 2007, Volpe, 2008). The mean control TER value was 457 Ω cm², with a range of 328-728 Ω cm² (Moyes et al, 2010a), in line with the literature (Delie and Rubas, 1997, Ingels and Augustijns, 2003). Cell height was 27.2 µm, in keeping with other reported values for cells *in vitro* (Hidalgo et al, 1989, van't Hof and van Meer, 1990, Nicklin et al, 1992) or human *in vivo* cells (Delie and Rubas, 1997): the apical dimensions were smaller than those in the literature (des Rieux et al, 2005, de Carvalho et al, 2006), although the protocols were not identical. The citation record for the first pre-treatment paper (Moyes et al, 2010a) included reference to it for the height of a Caco-2 cell (Ozeki et al, 2015), used when calculating permeability coefficients. Citations for the macrophage paper (Moyes et al, 2010b, Section 4.6) included confirmation of its report for cell packing (Xiao et al, 2014), any one cell being surrounded by six others (Uskoković et al, 2012a).

With respect to immunolabelling (Moyes et al, 2010b), actin was arranged in apical, lateral and basal zones (Shaw et al, 2005) and apical cell shape was clearly outlined by junctional protein localisation (de Carvalho et al, 2006, Musch et al, 2006). This included occludin and ZO-1 at tight junctions (TJs), seen as single lines in apical belts, as was adhering junction (AJ) E cadherin (Ferruza et al, 1999, Obert et al, 2000, Musch et al, 2006), which was, however, also found as basal arrays of spots (Fujimoto, 1995). As observed with TEM (Farquar and Palade, 1963), the three belts were part of the junctional complex, along with the most apical E cadherin spot, the others being classified as desmosomes. Where more

than two cells met at junction points, some of these had brighter immunolabelling (Moyes et al, 2010b), termed multicellular junctions (MCJs, Moyes et al, 2011).

The two pre-treatments (Moyes et al, 2010a) caused different effects. Ethanol decreased the TER and increased permeability, confirming previous reports of TJ loosening and related changes (Bjarnason et al, 1984Ma et al, 1999, Banan et al, 2007, Joseph et al, 2008, Amin et al, 2009). As for canine kidney cells (MDCK, Armitage et al, 1994), cooling produced a temperature-dependent increase in TER. The paper was quoted with respect to data on ethanol-induced changes in junctional occludin, including its diffuse profiles, by Tria et al (2013a). A subsequent paper (Moyes et al, 2011, Section 4.6) showed that after exposure times of 60mins, ethanol loosened and disorganised TJs, while cooling did the opposite: the junctions were also made more diffuse or sharp, respectively.

For connective tissue cell compartmentalised culture, macrophages caused the Caco-2 TER to decrease (Moyes et al, 2010b), loosening tight junctions, possibly by cytokine effects (Satsu et al, 2006, Al-Sadi et al, 2009). Cells in the Caco-2/macrophage group were shorter and wider, with less regular outlines, some with double linear junctions. Occludin labelling was more diffuse and ZO-1 more irregular, as seen after exposure to TNF α (Ma et al, 2004), possibly related to actin redistribution (Fanning et al, 1998, 2002). Control M cell permeability was higher, the cells were shorter and more irregular and there were differences in junctional protein distribution between the two monolayers.

4.8.8 Monolayer uptake routes (Model 2)

Addition of particles to control Caco-2 monolayers produced an initial marked TER increase followed by a slight decrease: addition of medium produced only the initial increase (Moyes et al, 2007, Section 4.5). The increase was possibly related to cooling, since particles were stored at low temperature (Carr et al, 2012, Section 4.7): certainly, when cells were chilled to 0°C for long enough, there was a TER decrease when particles, now warmer than cells, were added. The main decrease could have been associated with tight junction loosening and with uptake, as reported after exposure to other challenges (Stenson et al, 1993, Ferruzza et al, 1999, Araki et al, 2005) or to substances known to loosen tight junctions (González-Mariscal et al, 2005).

Increases in epithelial-associated particle numbers (EAPs, Moyes et al, 2007) took place over similar time-frames to the TER decrease. Submembranous particle numbers (SMPs) were higher after longer exposure (Moyes et al, 20010a, Section 4.6). Simulation of *in vivo*

peristalsis by removal of particles from the upper well caused a TER decrease and more EAPs in some groups (Moyes et al, 2007). Addition of vasoactive intestinal peptide (VIP) affected TER (Blais et al, 1997), but interpretation was complicated by its half-life (Gololobov et al, 1998) and the distance between the administration site and the receptors (Anderson et al, 2003). Citations to the data in this paper (Moyes et al, 2007) included references to particles reaching the epithelial surface through the unstirred water layer (Sugano, 2010), to its reports of paracellular uptake (Coppi et al, 2010, with Iannuccelli) by an enterocyte route (Iannuccelli et al, 2011, with Coppi) and to 'direct cellular uptake' with respect to micronutrients and nutroceuticals in a review by Velikov and Pelan (2008). Powell et al (2010), in their article on dietary nano and microparticles, made reference to it for the integrity of tight junctions being altered by particle 'endocytosis'.

Ethanol allowed more particles through the membrane, as well as loosening and disorganising TJs, while cooling had an opposite effect, but only after a sufficient exposure time (Moyes et al, 2011, Section 4.6). The first pre-treatment paper (Moyes et al, 2010a) was cited in a recent 'advanced review' on food-borne nanomaterials (Bouwmeester et al, 2018, with M. Jepson as an author): it was quoted for its report of *in vitro* effects of ethanol and for tight junction involvement, during comments on possible increased uptake associated with epithelial damage, such as at sites of cell turnover or in disease. Citations for data in the second pre-treatment paper (Moyes et al, 2011) included its use as a source for a TER increase/decrease after treatment (Tria et al, 2013a) and for the linkage of this to changes in ion permeable tight junction pores (Tria et al, 2013b).

After particle addition in the macrophage experiments (Moyes et al, 2010b, Section 4.6), Caco-2/macrophages had a lower TER. Submembranous particle numbers (SMPs) were high in the Caco-2/macrophage group, with many at or near the macrophages. These cells therefore loosened Caco-2 TJs and encouraged uptake, facilitated by their phagocytic abilities and the reduced epithelial cell height, possibly associated with actin remodelling. The macrophage paper (Moyes et al, 2010b) was cited by Uskoković et al (2012b), who referred to it during a description of the jagged nature of ZO-1 after particle exposure, categorising them and double line junctions as indicative of membrane disruption but not dysfunction. Li et al (2013) mentioned the relevance of the paper's report on increased uptake by Caco-2 monolayers in the presence of macrophages, especially for low permeability drugs.

After particle exposure, the M cell permeability increase was more obvious than that of Caco-2 cell monolayers. The initial report (Carr et al, 2012) that uptake was similar in both groups applied to the end of the experiment after exposure to particles for 120mins. However, in the early group, when the lack of an underlying connective tissue compartment as *in vivo* was less marked. M cell monolayers had more epithelium-associated particles and higher uptake rates (Carr, personal communication, 2018, Section 4.6). As in the macrophage experiments above (Moyes et al, 2010b), particles were often located at intercellular junctions of both control and treated groups. This was now described as producing two intercellular junction subsets, most without retained particles ('bare') and fewer with ('decorated'). Early particle-induced junctional protein plasticity occurred at adhering as well as at tight junctions and was often dissimilar at 'decorated' and 'bare' junctions. This suggested that uptake was paracellular, an echo of earlier TEM sightings there of latex particles (Smyth et al, 2005, Section 4.4), identifying the junctions as a possible uptake route. The higher early uptake by M cell monolayers could have been linked to cell shortening or the different junctional protein responses in apical edge levels or other parameters.

4.8.9 Particle-associated factors (Model 2)

Model 2 produced quantitative data, showing that most particles were often 'singles' and that the proportion of aggregates increased with exposure time (Moyes et al, 2008, Section 4.6, Carr et al, 2012, Section 4.7). *In vitro* and *in vivo in situ* particle numbers (Moyes et al, 2007, Section 4.5) were in the same range, especially for data calculated per 1000 cells (Moyes et al, 2008) or for ATCC cells (Carr et al, 2012). Particle numbers on and associated with the monolayer increased with time, more so than in the *in vivo* model, which had underlying connective tissue, facilitating onward particle movement to secondary organs, and avoiding build up at the epithelium. Data citations to the first pre-treatment paper (Moyes et al, 2010a, Section 4.6) included a reference to the good link between permeability *in vitro* and 'absorption' *in vivo* (Ferraro et al, 2014).

4.9 Q. 2 Are any groups 'at risk', such as pregnant or lactating females?

4.9.1 Pregnancy and lactation (Model 1)

The question of whether pregnant or nursing mothers and their offspring were more at risk after Chernobyl as in other exposures (Bhattacharya, 1983, Leazer et al, 2002) was an

important issue in drug delivery (Florence, 1997) and radionuclide dosimetry (Harrison, 1998). Proximo-distal profiles of uptake varied across groups with different reproductive profiles (Smyth et al, 2005, Section 4.4) and particle numbers for late pregnant and early lactating rats were higher than those for virgin female animals, significantly so 5 minutes after administration: the differences were related to numbers associated with villous enterocytes and goblet cells. Factors implicated included gastric emptying (Paterson et al, 2000), endocrine influence (Chang et al, 1995) and the role of mucus (Sharma et al. 1995): the higher particle numbers at goblet cells in early lactating groups could have reflected corresponding changes to mucus elsewhere (Gadegone and Bhiwagde, 1981). The main uptake route was confirmed to be villous for *in vivo in situ* uptake. This highlighted again the relevance of tight junctions, given their possible response to hormone modifiers and their leakiness in pregnant animals (Nguyen and Neville, 1998, Cereijido et al, 2000, Nusrat et al, 2000). Available data citations for the paper on reproductive status (Smyth et al, 2005) surprisingly did not include any on its main topic of 'at risk' groups. In terms of general citations, Wu et al (2011) quoted this and three others of the group's papers, along with two from S. Florence, with respect to several aspects of uptake, including the involvement of enterocytes as well as M cells. Kobori et al (2012) quoted it for uptake being probably paracellular through tight junctions, with respect to novel food particles being taken up.

As part of this work on groups with different reproductive history, pilot studies showed that particles were found in the stomachs of the three rat pups examined (Carr et al, 2012, Section 4.7).

4.9.2 Species, age and pathophysiology (Model 1)

The lack of difference in uptake across species (Doyle-McCullough et al, 2007, Section 4.4) was difficult to compare with the literature at the time, due to protocol variations (McClean et al, 1998). Age was, however, important: uptake in young adult rats was higher than that of young, mature and old groups, confirming some earlier data (Harrison and Fritsch, 1992, Seifert et al, 1996) but not others (LeFevre et al, 1989, Simon et al, 1994). Some young adult females, therefore, had a double risk of taking up harmful particles *in vivo in situ*, since this sensitive age coincided with their active reproductive period (Smyth et al, 2005, Section 4.4). Both factors were also important during dose calculations for vaccines or pharmacological agents. Citations were made to the Doyle-McCullough (2007) paper's data for particles being in the unstirred water layer on the intestinal surface in a review by Sugano and Terada (2015). There were references to it for evidence of uptake (Lai et al, 2008,

Jünemann and Dressman's 2012 review, Schleh et al, 2012) and for it taking place through 'direct cellular delivery (Velikov and Pelan's review, 2008) or at goblet cells (Johnson et al, 2011). There were also comments in a paper on dietary nano- and micro-particles (Powell et al, 2010), quoting the data, with those from two other papers from the group, as demonstrating a predominantly villous route. One paper highlighted correctly the core point of the paper, in its mention of the importance of the intestinal size and of the higher uptake in younger animals (Woitiski et al, 2008).

With respect to pathophysiology, the possible effect of particle exposure on individuals with another health problem (McMinn et al, 1996, Section 4.4), was explored using the model for streptozotocin-induced diabetes described by Ettarh and Carr (1997, Chapter 2, Section 2.5) prior to the study of post-irradiation responses (Ettarh et al, 2000, Section 3.9). Small intestines were longer and wider in diabetic groups, confirming earlier findings (Mantle et al, 1989). The maceration data wrongly implied that there were many more particles in the diabetic intestine, highlighting the need for microscopy, which showed that those supposed higher numbers were found only in luminal compartments and had not yet been taken into the tissues. The numbers in the different parts of the diabetic intestinal wall implied reduced particle movement, related perhaps to the thicker basal lamina (Mantle et al, 1989). For secondary organs, with no luminal compartments to complicate the issue, the maceration data were not inaccurate: lymph node and liver had fewer particles in diabetic groups and the microscopy data did not contradict this. The conclusion overall was therefore that diabetic individuals did not form another at-risk group. Data were cited for confirmation of 'non-lymphoid' uptake (Behrens et al, 2002) and secondary organ involvement (Yang et al, 1999, Akande et al, 2010). Lopes et al (2014) in a review article quoted it as demonstrating that diabetes could influence the extent of uptake and therefore the dosimetry associated with insulin delivery systems.

4.9.3 Irradiation and related exposures (Models 1 and 2)

Another 'at risk' situation related to the project having its roots in the accidental release of radioactive particles (Cuddihy et al, 1989, Wakeford, 2011a, b), when ambient radiation could be a factor. Studies were therefore carried out on radiation-induced changes in particle uptake.

For Model 1, microscopy of nodes showed more particles at the subcapsular sinuses 3 days after irradiation (Carr et al, 2012, Section 4.7). Other experiments indicated that reserpine

(ibid), known to affect intestinal motility (Plummer et al, 1955) and simulate irradiation effects (Indran et al, 1985, 1991, Section 3.7), increased total uptake, in keeping with the fact that atropine, with an opposite action (Carr et al, 1991a, Section 3.7), led to decreased uptake (Volkheimer et al, 1968c, Volkheimer, 1977).

For Model 2, the work related to radiation effects and tight junction responses (Moyes et al, 2008, Section 4.6). Radiation-induced changes in small intestine had been widely reported, including in reviews (Carr, 2001, Chapter 3, Section 3.10, Wang and Hauer-Jensen, 2003) and books, Galland and Spencer, 1990, Dubois et al, 1995, Fajardo et al, 2001). Tight junctions (TJs), implicated in particle uptake (McCullough et al, 1995, Section 4.3, Moyes et al, 2007, Section 4.5, Carr et al, 2012, Section 4.7), were disrupted by radiation (Porvaznik, 1979, Somosy et al, 1993). Immunocytochemistry had shown changes in the intercellular transmembrane proteins (Dublineau et al, 2004, de Carvalho et al, 2006). In Moyes et al (2008), irradiation produced lower and broader cells. The increase in calculated intercellular space was similar to that seen directly by TEM after radiation in vivo (Lieb et al, 1977) and *in vitro* (de Carvalho et al, 2006). This change was linked to the relocation of junctional proteins (Dublineau et al, 2004), possibly giving desmosomal loosening, which, if happening at hemi-desmosomes, could have produced disorganised basal laminae and epithelial/stromal stripping (Carr et al, 1981d, Chapter 3, Section 3.6) or loss of epithelial integrity (Dublineau et al, 2004). For submembranous (SMP) numbers, irradiated groups allowed more through than controls. In conclusion, external irradiation could have affected uptake, with possible links between this and changes in permeability. Citations to this paper included those to data for uptake being influenced by various factors, including external radiation (Wu et al, 2011) and radiation-induced alterations in permeability and microparticle uptake (Morini et al, 2017). References are also made to it for irradiation causing disruption or opening of tight junctions (Stern et al, 2013, Strup-Perrot et al, 2016). In the review already mentioned above for ethanol (Section 4.8.8), the paper was quoted for irradiation affecting paracellular uptake and tight junctions (Bouwmeester al 2018, with M. Jepson as an author).

In the pilot Model 2 study with I. Dublineau, depleted uranium, iron and latex all produced the increase/decrease TER response pattern, particularly the higher concentration of the uranium (Carr et al, 2012, data courtesy of Dublineau, Moyes, Steffani, Carr). This project was not continued when support ended as the grant body's radioprotection programme was reviewed.

4.10 Overview and conclusions

The review (Carr et al, 2012, Section 4.7) made several points, including: the advantages of the microscopy technique; the small percentage of administered particles taken up; the preponderant *in vivo in situ* villous uptake and the possible role of tight junctions, confirmed *in vitro*; the location of many particles near or at intercellular junctions; and the considerable plasticity of proteins at adhering as well as tight junctions.

This position in the account of Chapter 4 has been chosen to put the review (ibid) in context with the later literature. It has been quoted in articles on additional applications of uptake, including microsphere design (Kazazi-Hyseni et al, 2014), drug encapsulation (Ramazani et al, 2016), engineered nanomaterials in soils (Cornelis et al, 2014), regulation of nanoparticle drug delivery (Hussain, 2016), environmental plastics in food production (Bouwmeester et al, 2015) and satiety-simulating food additives (Naftalovich et al, 2016). Data topics cited included the villous, non-follicle-associated epithelial site of uptake (Braakhuis et al, 2015, Reineke et al, 2013), the significance of this with respect to the time after particle administration and whether or not to target drugs to M cells (Hussain, 2016), the possible involvement of tight junctions (Cornelis et al, 2014) and the percentage uptake (Bouwmeester et al, 2015). The article was also cited for the importance of particle size (Kazazi-Hyseni et al, 2014, Ramazani et al, 2016), while Naftalovich et al (2016), describing it as having made 'an extensive review of the topic', quoted it verbatim for the relationship between size, uptake and penetration. Types of citing articles included a review article (Braakhuis et al, 2015), a critical review (Bouwmeester et al, 2015) an article on 'unique insights' into the fate of microspheres (Reineke et al, 2013) and an article forming part of a special journal issue to honour the contributions of S. Florence to the field (Hussain, 2016).

In conclusion, the answers to the two questions posed post-Chernobyl were as follows.

To the first question, 'Do microparticles enter tissues by an intestinal route ?', the answer was that, after *in vivo in situ* exposures as short as 5 minutes, particles did cross the small intestinal epithelium, moving on to secondary organs such as lymph nodes: particle dose and size were important. Entry at times soon after uptake was, unexpectedly, mostly through villous epithelium and not the M cells of lymphocyte-rich Peyer's patches. Particles were observed at intercellular junctions or spaces, highlighting the possible involvement of enterocyte tight junctions. *In vitro* work addressed this, showing that ethanol loosened junctions and increased uptake, while cooling produced opposite effects and different

changes in immunolabelled junctional proteins. Particle retention at intercellular junctions was also seen in this Model, further implicating tight junctions and their associated proteins.

To the second question, 'Are any groups 'at risk', such as pregnant or lactating females ?', the answer was that uptake could be higher in late pregnancy and early lactation, was more affected by age than species, was not increased by diabetes, but could be increased by irradiation.

There has therefore been continuing evidence of the relevance of microparticle uptake, not only to diet and therefore environmental issues (Volkheimer and Schultz, 1968), but also to drug and vaccine delivery (O'Hagan, 1990), to clinical topics such as multiple organ dysfunction syndrome (Nieuwenhuijzen et al, 1996) and to the additional areas listed in the citations entries for the review (this section, above). The data have presented strong confirmation of the earlier firm statement that 'persorption of solid particles in the micrometer range is a fact' (Volkheimer, 1977) and the equally definite description of the process as 'neither exceptional nor unusual' (Florence, 1997).

4.11 Context and comment

The theme here was the application of the understanding of mammalian small intestine acquired in Chapters 1 and 2 and of the multi-parameter approach of Chapter 3 to the exploration of the sites and pathways involved in microparticle uptake across the small intestinal wall from lumen to serosa. The main addresses were Belfast, Harwell and Oxford, covering only twenty years, publications running from 1995 to 2012: this was a shorter span than Chapters 2 or 3 and included a smaller number of publications.

For output destination, fewer were in anatomical journals, although the main review was in Progress in Histochemistry and Cytochemistry. There was also a small number of publications in general biological journals and more in those specialising in radiobiology, gastroenterology, or pharmaceutical science.

With respect to collaborative research, the topic of microparticle uptake involved proportionally more in-house work, in Belfast and Oxford. There were, however, still important networks. These involved already established links, such as that with Imperial Cancer Research Fund (ICRF) in London, who moved one of their staff and accompanying equipment to Belfast during a reorganisation. There were collaborations with the National Radiological Board (NRPB), Atomic Energy Authority, St Bartolomew's Hospital, London and the University Medical Centre, Hamburg Eppendorf, Germany. These and other links

brought contacts with groups interested in toxicology, drug delivery, gastrointestinal biology and clinical applications, such as surgery. The establishment of the Caco-2 *in vitro* model relied on the advice and help of Professor Ian Sanderson and his colleagues, Queen Mary University, London.

Roles and contributions changed during this period, but in a slightly different way, given the end of the commitment as Director of School and a two-year sabbatical in Harwell and Oxford and then early retirement to allow these attachments to continue. Much of the work was done while supervisor, Head of Department/School or group leader, but there was a continuing involvement in protocol design, the provision of technical training, data quality control and writing of reports and papers. Most recently, the role has been as Principal Investigator on the Department of Health Grant: while Academic Visitor in Anatomy, Department of Physiology, Anatomy and Genetics, University of Oxford, involvement has included writing grant renewal proposals, presenting the outcome at annual funding body meetings, guiding the direction of the programme, supervising the younger scientists involved and contributing in a major way to the production of reports and papers: time was also once more available for contributions to data recording.

With respect to links to the next Chapter, the summary of the data on uptake has provided a good bridge to Chapter 5, which has been constructed to provide an overview of the contents of all four Chapters in context with the wider literature.

Chapter 5 Conclusions – key findings and significance

Preface - extract from Presubmission Abstract (Chapter 5)

The Chapters above concentrate on the scientific findings, from work done when on staff in Universities in Glasgow and Belfast and when a Guest researcher/Visitor in Berkeley, Harwell or Oxford. These Chapters also: comment on the work's significance; discuss article and journal choice; describe collaborative links with London, Chicago, Hamburg and others; and mention where 'hands-on' work gave way to supervisory input. Chapter 5 draws these all together and highlights the linking theme of the importance of probing surfaces as well as exploring underlying tissues.

5.1 Background to output

Now that all the data sets and their interpretation have been set out in Chapters 1 to 4, an overview has been constructed of their content and possible influence. As set out in the Preface, all but 2/103 submissions (Carr et al, 2001, Chapter 4, Section 4.3, Carr, personal communication, 2018, Chapter 4, Section 4.6) were published as articles in journals, the records of conference proceedings or as chapters or books. In the 101 full publications, 80% were in one or other 'lead' position in the authorship list. In only one (recollected to be Ettarh and Carr, 1996b, Chapter 2, Section 2.5), was input to writing minimal or non-existent, taking place after the graduate student had moved on to another Department: this paper was, however, included because it was the culmination of several mini-projects on the subject, initiated and supervised by the thesis author.

Output was published in journals associated with the disciplines of anatomy, histology or microscopy, including the Journal of Anatomy and the Journal of Microscopy, associated with the Anatomical and Royal Microscopical Societies respectively. The papers in Scanning Electron Microscopy were placed there because of the advantages of meeting other early scanning electron microscope (SEM) practitioners at the annual conferences where the pre-publication papers were delivered as oral presentations. Those in the Journal of Submicroscopic Cytology/Pathology were submitted there because of the excellence of their micrograph reproduction, a matter of importance when developments in imaging or interpretation were being reported.

During the time of publication of the work, there has been increasing emphasis on journal Impact Factors. Although these were not borne in mind during selection of output destination when the accompanying material was being worked on, an analysis has been made recently of the Factors for the years 1999, 2006 and 2016. None of the Journals used in the thesis was as highly cited as journals such as Nature or Cell. However, some of the Journals were well placed in their group, such as Gut, in the top first or second position in a group of between 87 and 134 Journals in the Gastroenterology subject area and the Journal of Investigative Dermatology in the top place in a group of more than 100 Journals. The Pharmaceutical Journals were also on the whole in good positions, Pharmaceutical Research in the top three in 1999, The International Journal of Pharmaceutics approximately in the top 10% and The Journal of Drug Targeting within the top 20%. The Journal of Anatomy and The Journal of Microscopy were less well placed, but usually in the top third of their lists.

5.2 Influence of publications in Chapters 1 to 4

Chapters 1 through 4 have given an account of the scientific flow and development of the work, including citations to the individual publications. This Subsection has looked at the publications more generally, in three ways, each aiming to comment on the most important five or so publications: this demanding criterion was used to minimise the space taken up by this aspect of the thesis.

5.2.1 Representative papers

These five papers have been selected to reflect retrospective interest in aspects of the main thrust of the work, taking it from the systematic observation of surfaces with correlative SEM, through analysis of irradiated surfaces and their underlying responses to uptake of microparticles by an 'unexpected' route.

The first (Carr et al, 1983a, Chapter 2, Section 2.4) came from the account of the emergence of scanning electron microscopy. It was the introductory low temperature paper, representative because it was early in the use of its particular technique for the examination of small intestinal villi and had used correlative techniques to maximise the information produced.

The second and third papers were from the Chapter on correlative morphology of irradiated small intestinal surfaces. The second (Carr et al, 1983c, Chapter 3, Section 3.5) applied two

dimensional information from the literature to produce the objective scoring system used to take the description of radiation-induced villous collapse beyond the merely descriptive. It analysed data from the study of surfaces and linked them with possible underlying stromal changes, in this case defining villous pegs and pericryptal plates as important. The third paper (Carr et al, 1992a, Chapter 3, Section 3.8) took this theme further, in particular the importance of non-epithelial tissues. This was the use of the Morphological Index display to compare the effects of changing dose and time after X-irradiation. It extended data collection from crypt count assays to include several villous and cryptal parameters, illustrating in graphical form their wide range of responses as dose and time changed.

The fourth and fifth papers were from the Chapter on microparticles crossing intestinal surfaces. The fourth (Smyth et al, 2005, Chapter 4, Section 4.4) was representative of the in vivo in situ model for uptake in using maceration and microscopy. It used nine proximodistal segments and gave proof of validation of the techniques and evidence for the predominant uptake pathway being villous and not at Peyer's patches. It illustrated with transmission electron microscopy the presence of latex microparticles in intercellular spaces, thus confirming a possible role for tight junctions. It also showed that particles had moved on to secondary compartments or organs. The fifth paper, in a study of the effect of macrophages on uptake in the *in vitro* model (Moyes et al, 2010b, Chapter 4, Section 4.6), illustrated a full range of typical in vitro techniques, including measurement of transepithelial resistance (TER), cytokine levels, cell dimensions, particle counts for the lower well and also immunolocalisation of junctional proteins. The data supported the involvement of tight junctions during exposure to particles and highlighted the importance to uptake of the underlying cells, in this case macrophages. It also showed particles associated with intercellular junctions, complementing their sighting in vivo in intercellular spaces (Smyth et al, 2005, above) and again implicating tight junctions and a paracellular uptake route.

All five, although selected to be representative scientifically, also summarised the research in other ways, through the scientific centres involved, collaborative input, student contributions or Journal choice. The addresses reflected the move from hands-on contribution as a Glasgow University staff member through involvement as School Director in Biomedical Science in Belfast to being part of a small research grouping in Oxford University. Collaborative links were also representative, including Radiotherapeutics in Glasgow, The MRC Cyclotron Unit at the Hammersmith Hospital, London, the Lawrence Berkeley Laboratory, California, USA and the University of Washington, Seattle, USA. The composition of the author list was also typical of the whole, including academic colleagues in Belfast and Oxford, technicians in Glasgow and Belfast and graduate students in Belfast and Oxford. The Journal choice spanned the types of output destination used, with three anatomical, one clinical radiology and one radiation science journal.

In conclusion, these five papers represented the whole body of work, although they were selected by subjective criteria, relying entirely on the author's opinion of the papers. The use of this 'author's hindsight' approach to select subjects of interest has also drawn attention to the fact that some have received less attention than was optimal, partly due to research focus changing or unavailability of funding. Selection of one 'sidelined' topic in each main Chapter focussed attention on the following areas that required more research: the significance of microridges (Shasha'a et al, 1993, Chapter 2, Section 2.5, Robertson et al, 1987, Chapter 3, Section 3.3), the possible overlapping effects of existing abnormal conditions and external irradiation (Ettarh et al, 2000, Chapter 3, Section 3.9) and the effects of irradiation or radioactive and associated particles (Moyes et al, 2008, Carr et al, 2012, Chapter 4, Sections 4.6, 4.7)

A different subjective author's view of the whole output has been the identification of a common approach throughout. This was a tendency to look at additional topics that complemented what was regarded at the time as a main theme in the research field. For the SEM work, this involved an interest not only in the production of surface images, but also the use of non-standard images and correlative methods. For the subcellular radiobiology, it took the form of reporting not only what could have caused side-effects in developing therapeutic treatments, but also collecting data on non-proliferative structures not identified classically as 'targets'. For the particle uptake work, the finding that, as an early phenomenon *in vivo in situ*, it occurred predominantly at villous sites, led to interest in the plasticity of junctional proteins and to the observation by others that over-concentration on targeting M cells could be unproductive.

5.2.2 Highly cited papers

This analysis, carried out during thesis preparation in 2017, used the top five papers in the Scopus citation data base. Only the peer-reviewed publications put forward in the thesis were included. This involved disregarding three papers within the top eight, two being duplicate coverage of material already in submitted papers and one stemming from a colleague's research area. In addition, a note was made of papers not in the Scopus list, but

ranked by Google Scholar in the top fifteen, giving consideration to publications of a type not included in Scopus. None of the research fields was large and none of the papers attracted long lists of citations, the largest here being 75 in Scopus and 135 in Google Scholar: all those selected had 30 or more citations.

The first paper in the top five of the Scopus list, with 75 citations, described the mainly villous route of microparticle uptake *in vivo in situ*: it was published in Digestive Diseases and Sciences (Hodges et al, 1995b, Chapter 4, Section 4.2). The second in the list, with 66 citations, was on the effect of dietary fibre, also in a specialist gastrointestinal Journal, Gut (McCullough, J.S. et al, 1998, Chapter 1, Section 1.2). The third, with 65 citations, was a further paper on particles, this time on their size, in Pharmaceutical Research (Carr et al, 1996c, Chapter 4, Section 4.4). The fourth, with 35 citations, was another paper on diet, this time on fat and in the British Journal of Nutrition (Sagher et al, 1991, Chapter 1, Section 1.2). The fifth paper, with 33 citations, was again a microparticle uptake paper, on the effect of diabetes on microparticle uptake, in the Journal of Anatomy (McMinn et al, 1996, Chapter 4, Section 4.4).

Consultation of Google Scholar produced two other publications for consideration, again omitting papers from the top fifteen of a similar type to those excluded from the Scopus list. Its surprising top scorer with 135 citations was the book on the digestive system (Toner et al, 1971, Chapter 2, Section 2.2), with some early SEM images. Another publication in the same Chapter and not in the Scopus list, was the Letter to the Editor of Lancet, describing briefly with SEM images the three dimensional structure of small intestinal villi (Carr and Toner, 1968, Chapter 2, Section 2.1): it had 31 Google Scholar citations and in was in thirteenth place.

In conclusion, the Scopus top five were either in Chapter 1, reflecting contributions to strong collaborative groupings, or in Chapter 4, whose more 'in-house' subject of microparticle uptake has been of continuing interest recently, with relevance to environmental issues. The Google Scholar additions from Chapter 2 included here were from early SEM publications, often reaching a wide audience due to the images providing scientific and educational information in what was then a novel format. Another notable point was that although neither Chapter 2 nor Chapter 3 had any papers in the Scopus top five, when the filter was relaxed to include the top ten, one article in Chapter 2 and two in Chapter 3 would have been included. With respect to Journal choice, all but one of the Scopus top five were in specialist journals, mostly in the gastrointestinal field.

5.2.3 Citations of interest

The five publications here have also been chosen using Scopus and Google Scholar, usually from papers in the top 20% of the lists: the selection was made mainly because of the topic highlighted, although the standing of the citing group was also taken into account.

In the first two Chapters, one of the most interesting citations was to the work on cells settling on glass (Carr and Carr, 1970, Chapter 2, Section 2.1) for its role in the identification of the Kupffer cells of the liver as of the macrophage line (Muto, 1975, Motta, 1975). Another citation of note was to the SEM review (Carr, 1971, Chapter 2, Section 2), where it was used as one of three key papers setting the scene in an overview of 'trends and prospects' in SEM (Hayes, 1974): it was also of interest because of the reputation of the author quoting it in this way.

For the work on the effects of irradiation, the main review (Carr, 2001, Chapter 3, Section 3.10) was included, for its range of citation dates, up to 2017, but also for the range of topics quoted. These included citations to the different responses of constituent cells and tissues of the intestinal wall (Wang and Hauer-Jensen, 2003, Veerginadis et al, 2017), demonstrating that individuals with different interests were accessing information by reading the Index Tables as they had been designed. It was also cited with reference to mechanistic aspects (Senesse et al, 2009) and to clinical topics (Cameron et al, 2012, Barnett et al, 2013), including possible post-irradiation treatments (Ishihara et al, 2011). Another citation, by Hauer-Jensen (2014), one of the key authorities in the field, was also notable because of its description of the paper as a 'very comprehensive review'.

From the work on microparticle uptake, Doyle-McCullough et al (2007, Chapter 4, Section 4.4) was cited on its key finding that uptake was higher in young animals (Woitiski et al, 2008), relevant to pollution, drug delivery and multiple organ dysfunction syndrome (MODS): this paper also had a citation for percentage uptake and the predominance of villous involvement in a paper on dietary nano- and micro-particles (Powell et al, 2010), referring to it as part of the 'extensive studies from Kate (KE) Carr's group in Belfast, and now Oxford'. Citations to the review on microparticle uptake by Carr et al (2012, Chapter 4, Section 4.7) included one crucially covering the villous site of uptake and its relevance to whether or not drug targeting should be aimed at M cells (Hussain, 2016): it was also of interest because it came from a paper published as part of a special issue dedicated to S. Florence, marking his special contributions to particle uptake.

In conclusion, this selection was as subjective as the first one listing representative papers (Section 5.2.1), since it also depended entirely on the author's' choice, but it did have hidden semi-objective information on how other groups had viewed the work.

5.2.4 Summary of output analysis

All the publications were in academic outlets and of the seventeen selected, most were in standard Journals, a few in review series, one from a book and one a letter to an editor. Although original articles took all the Scopus top five citation places numerically, review articles appeared to have more influence overall than has recently been thought. Of the topics covered, most were entirely or mainly on gut, a few on macrophages or their influence on gut cells and one was methodological, dealing with instrumentation and image interpretation. Over 80% of the selected publications had the thesis author in one of the 'lead' positions, similar to the percentage for the whole published submission list. When comparison was made of the output selected by these three different ways of assessing their contributions, none featured in more than one list, leading to the unsurprising conclusion that what seemed to be of most importance or interest to an author did not necessarily attract attention from a wider readership.

5.3 Context and comment - probes and surfaces

The theme of probing has been found throughout the thesis, at more than one level, probing both to produce images and also for a response to environmental challenge, whether or not the research was more applied, as in post-Chernobyl *in vivo* microparticle work, or more 'blue sky' as in some of the corresponding *in vitro* experiments.

In both Chapters 1 and 2, the probing was mainly but not exclusively, at one level, relating therefore substantially to imaging. In Chapter 1, electrons or light produced images of sectioned material, working out the relationship of the constituent structures. In Chapter 2, one-level probing was by a de-magnified electron beam interacting with the superficial part of specimens and producing images of different types, often needing correlative data to interpret them. Malpighi (1687, in Carr et al, 1997, Chapter 4, Section 4.1) would perhaps have had a comment, 'Observation by means of the microscope will reveal more wonderful things than those viewed in regard to mere structure and connexion'.

In Chapter 3, the probing was usually at two levels, addressing further the point of 'structure and connexion'. Firstly, the sample was probed experimentally by external irradiation,

interacting with control multicellular mammalian tissues, to produce a range of outcomes and sometimes to facilitate interpretation of control structure. Secondly, these samples were then examined as in Chapters 1 or 2, probed by an electron beam to produce images of the radiation-induced responses, leading to correlation, not just structurally, but with parameters associated with the pathophysiology of intestinal function. The appropriate comment by Malpighi (ibid) could have been 'The solution of those problems may prepare the way for greater things and will place the operation of nature more clearly before the eyes'.

In Chapter 4, the 'greater things' could have been related to the fact that there were more examples of up to four types of probe, at two levels. Firstly, experimental probing involved the addition of microparticles, which explored the surfaces of mammalian tissues or cells and found pathways through to underlying tissues or compartments. A proportion of the submissions also had a second example of this type of experimental probing, such as subjects with immune or other pathophysiological issues or samples exposed to ethanol, ice or irradiation. Secondly, there was probing as in Chapters 1 and 2, by the production of confocal or electron microscope images, in some cases with a further level of probe, using immunocytochemistry to chart the presence and distribution of junctional proteins and explore their responses. And the final comment from Malpighi (ibid), perhaps with tongue in cheek, could have been 'And when we try to unravel the obscure things in the viscera of animals, at length by our efforts, and only with great weariness, we conclude that the truth of our observations is made out'.

Complete list of References in Harvard format

- Abbas, B., Boyle, F.C., Wilson, D.J., Nelson, A.C. & Carr, K.E. 1990a, "Radiation induced changes in the blood capillaries of rat duodenal villi: a corrosion cast, light and transmission electron microscopical study", *Journal of submicroscopic cytology and pathology*, vol. 22, no. 1, pp. 63–70.
- Abbas, B., Hayes, T.L., Wilson, D.J. & Carr, K.E. 1989, "Internal structure of the intestinal villus: morphological and morphometric observations at different levels of the mouse villus", *Journal of anatomy*, vol. 162, pp. 263– 273.
- Abbas, B., Hume, S.P., McCullough, J.S., Wilson, D.J., Stewart, P.C. & Carr, K.E. 1990b, "Early morphological changes in blood capillaries of mouse duodenal villi induced by X-irradiation", *Journal of submicroscopic cytology and pathology*, vol. 22, no. 4, pp. 609–614.
- Abu-Hijleh, M.F., Reid, O., McGadey, J. & Scothorne, R.J. 1994, "Distribution of mesothelial stomata in the rat", *Clinical anatomy*, vol. 7, no. 4, pp. 189–203.
- Ackerman, L., Piros, J., de Carle, D. & Christensen, J. 1976, "A scanning electron microscopic study of esophageal mucosa". *Scanning electron microscopy*, vol. 5, pp. 247–252.
- Aijaz, S., Balda, M.S. & Matter, K. 2006, "Tight junctions: molecular architecture and function", *International review of cytology*, vol. 248, pp.261–298.
- Akande, J., Yeboah, K.G., Addo, R.T., Siddig, A., Oettinger, C.W. & D'Souza, M.J. 2010, "Targeted delivery of antigens to the gut-associated lymphoid tissues: 2. Ex vivo evaluation of lectin-labelled albumin microspheres for targeted delivery of antigens to the M-cells of the Peyer's patches", *Journal of microencapsulation*, vol. 27, no. 4, pp. 325–336.
- Åkesson, C.P., McGovern, G., Dagleish, M.P., Espenes, A., Press, C.M., Landsverk, T. & Jeffrey, M. 2011, "Exosomeproducing follicle associated epithelium is not involved in uptake of PrPd from the gut of sheep (Ovis aries): an ultrastructural study", *PloS one*, vol. 6, no. 7, p.e22180, 10 pages.
- Akpolat, M., Kanter, M. & Uzal, M.C. 2009, "Protective effects of curcumin against gamma radiation-induced ileal mucosal damage", *Archives of toxicology*, vol. 83, no. 6, pp. 609–617.
- Akpolat, M., Topçu-Tarladaçalişir, Y., Dökmeci, D., Turan, F.N. & Uzal, M.C. 2011, "Protective effects of ibuprofen and L-carnitine against whole body gamma irradiation-induced duodenal mucosal injury", *Balkan medical journal*, vol. 28, no. 1, pp. 80-85.
- Alam, I.A. & Sadiq, M. 1989, "Metal contamination of drinking water from corrosion of distribution pipes", *Environmental pollution*, vol. 57, no. 2, pp. 167–178.
- Al-Jaff, D.M.A., Cook, G.T., Carr K.E., & Duncan H.J. 1982, "Further studies on bracken morphology in relation to herbicide uptake", in *The plant cuticle. Linnean society symposium series 10*, Eds. Cutler, D. F., Alvin, K. L. & Price, C. E., Academic Press, London, pp. 293–301.

- Allgood, J.W., Langberg, C.W., Sung, C.C. & Hauer-Jensen, M. 1996, "Timing of concomitant boost irradiation affects incidence and severity of intestinal complications", *International journal of radiation oncology biology physics*, vol. 34, no. 2, pp. 381–387.
- Alonso-Amelot, M.E. & Rodulfo-Baechler, S. 1996, "Comparative spatial distribution, size, biomass and growth rate of two varieties of bracken fern (Pteridium aquilinum L. Kuhn) in a neotropical montane habitat". *Vegetatio*, vol. 125, no. 2, pp. 137-147.
- Alpen, E.L. 1998, "Radiation Biophysics", 2nd edition, Academic Press, San Diego.
- Alpen, E.L., Powers-Risius, P. & McDonald, M. 1980, "Survival of intestinal crypt cells after exposure to high Z, high-energy charged particles", *Radiation research*, vol. 83, no. 3, pp. 677-687.
- Al-Sadi, R., Boivin, M., & Ma, T. 2009, "Mechanism of cytokine modulation of epithelial tight junction barrier", *Frontiers in bioscience*, vol. 14, no. 7, pp. 2765–2778.
- Altmann, G.G. 1974, "Changes in the mucosa of the small intestine following methotrexate administration or abdominal X-irradiation", *Developmental dynamics*, vol. 140, no. 2, pp. 263–279.
- Amer, S., Na, K.J., Motobu, M., El-Abasy, M., Nakamura, K., Koge, K. & Hirota, Y. 2005, "Radioprotective effect of sugar cane extract in chickens", *Phytotherapy research*, vol. 19, no. 6, pp. 496–500.
- Amin, P.B., Diebel, L.N. & Liberati, D.M. 2009, "Dose-dependent effects of ethanol and E. coli on gut permeability and cytokine production", *Journal of surgical research*, vol. 157, no. 2, pp. 187–192.
- Anderson, C.M.H., Mendoza, M.E., Kennedy, D.J., Raldua, D. & Thwaites, D.T. 2003, "Inhibition of intestinal dipeptide transport by the neuropeptide VIP is an anti-absorptive effect via the VPAC1 receptor in a human enterocyte-like cell line (Caco-2)", *British journal of pharmacology*, vol. 138, no. 4, pp. 564–573.
- Anderson, J.H. & Withers, R.H. 1973, "Scanning electron microscope studies of irradiated rat intestinal mucosa", *Scanning electron microscopy*, vol. 3, pp.565 – 572.
- Araki, Y., Katoh, T., Ogawa, A., Bamba, S., Andoh, A., Koyama, S., Fujiyama, Y. & Bamba, T. 2005, "Bile acid modulates transepithelial permeability via the generation of reactive oxygen species in the Caco-2 cell line", *Free radical biology and medicine*, vol. 39, no. 6, pp. 769–780.
- Arendt, T. 1991, "Penetration of lanthanum through the main pancreatic duct epithelium in cats following exposure to infected human bile". *Digestive diseases and sciences*, vol. 36, no. 1, pp.75–81.
- Argenzio, R.A., Liacos, J.A., Levy, M.L., Meuten, D.J., Lecce, J.G. & Powell, D.W. 1990, "Villous atrophy, crypt hyperplasia, cellular infiltration, and impaired glucose-NA absorption in enteric cryptosporidiosis of pigs", *Gastroenterology*, vol. 98, no. 5, part 1, pp. 1129–1140.
- Armitage, W.J., Juss, B.K. & Easty, D.L. 1994, "Response of epithelial (MDCK) cell junctions to calcium removal and osmotic stress is influenced by temperature", *Cryobiology*, vol. 31, no. 5, pp. 453–460.
- Asikainen, P., Mikkonen, J.J. & Kullaa, A.M. 2014, "Surface morphology of superficial cells in irradiated oral mucosa: an experimental study in beagle dog", *Ultrastructural Pathology*, vol. 38, no. 4, pp. 268–72.
- Asikainen, P., Sirviö, E., Mikkonen, J.J.W., Singh, S.P., Schulten, E.A.J.M., Ten Bruggenkate, C.M., Koistinen, A.P. & Kullaa, A.M. 2015, "Microplicae - specialized surface structure of epithelial cells of wet-surfaced oral Mucosa", *Ultrastructural pathology*, vol. 39, no. 5, pp. 299–305.
- Atwell, W. 1994, "Anatomical models for space radiation applications: an overview", *Advances in space research*, vol. 14, no. 10, pp. 415–422.
- Badran, D., Soutar, D.S., Robertson, A.G., Payne, A.P., McDonald, S.W. & Scothorne, R.J. 1994, "Scanning electron microscopy of the surface morphology of superficial cells of buccal mucosa is unlikely to be useful in monitoring radiotherapy", *Clinical anatomy*, vol. 7, no. 1, pp. 34–41.
- Bakhru, S.H., Furtado, S., Morello, A.P. & Mathiowitz, E. 2013, "Oral delivery of proteins by biodegradable nanoparticles", *Advanced drug delivery reviews*, vol. 65, no. 6, pp. 811–821.
- Balcerzak, S.P., Lane, W.C. & Bullard, J.W. 1970, "Surface structure of intestinal epithelium", *Gastroenterology*, vol. 58, no. 1, pp. 49–55.
- Banan, A., Keshavarzian, A., Zhang, L., Shaikh, M., Forsyth, C.B., Tang, Y. & Fields, J.Z. 2007, "NF-κB activation as a key mechanism in ethanol-induced disruption of the F-actin cytoskeleton and monolayer barrier integrity in intestinal epithelium", *Alcohol*, vol. 41, no. 6, pp. 447–460.
- Bansil, R., Celli, J.P., Hardcastle, J.M. & Turner, B.S. 2013, "The influence of mucus microstructure and rheology in Helicobacter pylori infection", *Frontiers in immunology*, vol. 4, no. 310, pp. 1–12.
- Baradi, A.F. & Hope, J. 1964, "Observations on ultrastructure of rabbit mesothelium", *Experimental cell research,* vol. 34, no. 1, pp. 33-44.
- Barnett, A., Cedar, A., Siddiqui, F., Herzig, D., Fowlkes, E. & Thomas, C.R. 2013, "Colorectal cancer emergencies", *Journal of gastrointestinal cancer*, vol. 44, no. 2, pp. 132–142.
- Basson, M.D. 2003, "Paradigms for mechanical signal transduction in the intestinal epithelium category: molecular, cell, and developmental biology", *Digestion*, vol. 68, no. 4, pp. 217–225.
- Beaulieu, J.F. 1999, "Integrins and human intestinal cell functions", *Frontiers in bioscience: a journal and virtual library*, vol. 4, pp. D310–D321.
- Becciolini, A., Cremonini, D., Balzi, M., Fabbrica, D. & Cinotti, S. 1982, "Irradiation at different times of the day. Morphology and kinetics of the small intestine", *Acta radiologica, oncology*, vol. 21, no. 3, pp. 169–175.
- Behrens, I., Vila Pena, A.I., Alonso, M.J. & Kissel, T. 2002, "Comparative uptake studies of bioadhesive and nonbioadhesive nanoparticles in human intestinal cell lines and rats: the effect of mucus on particle adsorption and transport", *Pharmaceutical research*, vol. 19, no. 8, pp. 1185–1193.

- Beier, R. & Gebert, A. 1998, "Kinetics of particle uptake in the domes of Peyer's patches", *American journal of physiology gastrointestinal and liver physiology*, vol. 275, no. 1, pp. G130–G137.
- Berg, J. W. & Howell, M. A., 1974, "The geographic pathology of bowel cancer", Cancer, vol. 34, pp. 805–814.
- Bernatchez, S.F., Parks, P.J., Grussing, D.M., Matalas, S.L. & Nelson, G.S. 1998, "Histological characterization of a delayed wound healing model in pig", *Wound repair and regeneration*, vol. 6, no. 3, pp. 223–233.
- Berthrong, M. & Fajardo, L.F. 1981, "Radiation injury in surgical pathology: part II, alimentary tract", *The American journal of surgical pathology*, vol. 5, no. 2, pp. 153–178.
- Besterman, H.S., Adrian, T.E., Mallinson, C.N., Christofides, N.D., Sarson, D.L., Pera, A., Lombardo, L., Modigliani,
 R. & Bloom, S.R. 1982, "Gut hormone release after intestinal resection", *Gut*, vol. 23, no. 10, pp. 854–861.
- Beukeveld, G.J.J. & Wolthers, B.G. 1995, "Cholestyramine orally administered to patients with erythropoietic protoporphyria results in urinary excretion of protoporphyrin: to be explained by the Herbst-Volkheimer effect?", *Clinica chimica acta*, vol. 233, no. 1–2, pp. 119–126.
- Bhattacharyya, M.H. 1983, "Bioavailability of orally administered cadmium and lead to the mother, fetus, and neonate during pregnancy and lactation: an overview", *Science of the total environment*, vol. 28, no. 1–3, pp. 327–342.
- Bhattathiri, N.V., Bharathykkutty, C., Prathapan, R., Chirayathmanjiyil, D.A. & Nair, K.M. 1998, "Prediction of radiosensitivity of oral cancers by serial cytological assay of nuclear changes", *Radiotherapy and oncology*, vol. 49, no. 1, pp. 61–65.
- Bhuiyan, N.U. & Poston, J.W. Sr. 2005, "A revised model for electron dosimetry in the human small intestine", *Health physics*, vol. 88, no. 1, pp. 23–36.
- Bird, S.D. 2004, "Mesothelial primary cilia of peritoneal and other serosal surfaces", *Cell biology international*, vol. 28, no. 2, pp. 151–159.
- Bjarnason, I., Ward, K. & Peters, T. 1984, "The leaky gut of alcoholism: possible route of entry for toxic compounds", *the lancet*, vol. 323, no. 8370, pp.179–182.
- Björk, E., Isaksson, U., Edman, P. & Artursson, P. 1995, "Starch microspheres induce pulsatile delivery of drugs and peptides across the epithelial barrier by reversible separation of the tight junctions", *Journal of drug targeting*, vol. 2, no. 6, pp. 501–507.
- Blais, A., Aymard, P. & Lacour, B. 1997, "Paracellular calcium transport across Caco-2 and HT29 cell monolayers", *Pflugers archiv european journal of physiology*, vol. 434, no. 3, pp. 300–305.
- Bouwmeester, H., Hollman, P.C.H. & Peters, R.J.B. 2015, "Potential health impact of environmentally released micro- and nanoplastics in the human food production chain: experiences from nanotoxicology", *Environmental science and technology*, vol. 49, no. 15, pp. 8932–8947.

- Bouwmeester, H., van der Zande, M.A. & Jepson, M.A. 2018, "Effects of food-borne nanomaterials on gastrointestinal tissues and microbiota" *Wiley's interdisciplinary reviews nanomedicine nanobiotechnology*, vol. 10, no. 1481, pp.1012, ePub.
- Bowen, I.D., Lewis, G.H.J. & Winters, C. 1983, "The localization of histochemical and autoradiographic products in the scanning electron microscope by means of atomic number contrast", *The histochemical journal*, vol. 15, no. 12, pp. 1191–1202.
- Boyde, A. 1974, "Photogrammetry of stereo pair SEM images using separate measurements from the two images", *Scanning electron microscopy*, vol. 1, pp. 101–107.
- Boyde, A. & Lester, K.S. 1968, "A method of preparing bacterial plaque lining carious cavities for examination by scanning electron microscopy", *Archives of oral biology*, vol. 13, no. 12, pp. 1413–1419.
- Braakhuis, H.M., Kloet, S.K., Kezic, S., Kuper, F., Park, M.V.D.Z., Bellmann, S., van der Zande, M., Le Gac, S., Krystek, P., Peters, R.J.B., Rietjens, I.M.C.M. & Bouwmeester, H. 2015, "Progress and future of in vitro models to study translocation of nanoparticles", *Archives of toxicology*, vol. 89, no. 9, pp. 1469–1495.
- Brady, M.E. & Hayton, W.L. 1977a, "GI drug absorption in rats exposed to cobalt-60 γ-radiation I: extent of absorption", *Journal of pharmaceutical sciences*, vol. 66, no. 3, pp. 361–365.
- Brady, M.E. & Hayton, W.L. 1977b, "GI drug absorption in rats exposed to cobalt-60 γ-radiation II: in vivo rate of absorption", *Journal of pharmaceutical sciences*, vol. 66, no. 3, pp. 366–370.
- Brady, M.E. & Hayton, W.L. 1977c, "GI drug absorption in rats exposed to cobalt-60 γ-radiation III: in situ intestinal absorption", *Journal of pharmaceutical sciences*, vol. 66, no. 5, pp. 680–683.
- Branicki, F.J., Evans, D.F., Jones, J.A., Ogilvie, A.L., Atkinson, M. & Hardcastle, J.D. 1984, "A frequency-duration index (FDI) for the evaluation of ambulatory recordings of gastro-oesophageal reflux", *British journal of surgery*, vol. 71, no. 6, pp. 425–430.
- Brennan, P.C., Carr, K.E., Seed, T. & McCullough, J.S. 1998, "Acute and protracted radiation effects on small intestinal morphological parameters", *International journal of radiation biology*, vol. 73, no. 6, pp. 691–698.
- Brennan, P.C., McCullough, J.S. & Carr, K.E. 1999, "Variations in cell and structure populations along the length of murine small intestine", *Cells tissues organs*, vol. 164, no. 4, pp. 221–226.
- Bright, D.N.E. 1928, "The effects of exposure upon structure of certain heath-plants", *Journal of ecology*, vol. 16, pp. 323–365.
- Brownlee, I.A. 2011, "The physiological roles of dietary fibre" Food hydrocolloids, vol.25, no. 2, pp. 238-250.
- Brummer, M.E.G., Lowrie, P.M. & Tyler, W.S. 1975, "A technique for sequential examination of specific areas of large tissue blocks using SEM, LM and TEM", *Scanning electron microscopy*, vol. 8, pp. 333–340.

- Buell, M.G. & Harding, R.K. 1989, "Proinflammatory effects of local abdominal irradiation on rat gastrointestinal tract", *Digestive diseases and sciences*, vol. 34, no. 3, pp. 390–399.
- Burnett, W.T., Burke, A.W. Jr. & Upton, A.C. 1952, "Effect of acetyl-beta-methylcholine, carbamylcholine and atropine on X-Irradiated mice", *Federation proceedings*, vol. II, pp. 328–329.
- Burrell, J.S., Bradley, R.S., Walker-Samuel, S., Jamin, Y., Baker, L.C.J., Boult, J.K.R., Withers, P.J., Halliday, J., Waterton, J.C. & Robinson, S.P. 2012, "MRI measurements of vessel calibre in tumour xenografts: comparison with vascular corrosion casting", *Microvascular research*, vol. 84, no. 3, pp. 323–329.
- Busch, D.B. 1990, "Pathology of the radiation-damaged bowel", in *Radiation enteritis*, Eds. Galland, R.B. & Spencer, J., Edward Arnold, London, pp. 66–87.
- Cameron, S., Schwartz, A., Sultan, S., Schaefer, I.-M., Hermann, R., Rave-Fränk, M., Hess, C.F., Christiansen, H. & Ramadori, G. 2012, "Radiation-induced damage in different segments of the rat intestine after external beam irradiation of the liver", *Experimental and molecular pathology*, vol. 92, no. 2, pp. 243–258.
- Carr, K.E. 1967, "Fine structure of crystalline inclusions in the globule leucocyte of the mouse intestine", *Journal of anatomy*, vol. 101, no. 4, pp. 793–803.
- Carr, K.E. 1970a, "Intercalation and oxidation effects on graphite of a mixture of sulphuric and nitric acids", *Carbon*, vol. 8, no. 2, pp.155-166.
- Carr, K.E. 1970b, "Electron microscope study of the formation of graphite oxide", Carbon, vol. 8, no. 2, pp. 245-247.
- Carr, K.E. 1970c, "Scanning electron microscope studies of human skin", *British journal of plastic surgery*, vol. 23, pp. 66–72.
- Carr, K.E. 1971. "Applications of scanning electron microscopy in biology", *International review of cytology*, vol. 30, pp. 183–255. Academic Press.
- Carr, K.E. 1980, "Scanning electron microscopy of macrophages", In '*The reticuloendothelial system*' Vol I. Eds. Carr, I. & Daems, W.T., Plenum Press Corporation, pp. 259–295.
- Carr, K.E. 1981, "Scanning electron microscopy of tissue response to irradiation", *Scanning electron microscopy*, vol. 4, pp. 35-46.
- Carr, K.E. 2001, "Effects of radiation damage on intestinal morphology", *International review of cytology*, vol. 208, pp.1–119.
- Carr, K.E. 2018, Personal communication.
- Carr, K.E. & Carr, I. 1970, "How cells settle on glass: a study by light and scanning electron microscopy of some properties of normal and stimulated macrophages", *Zeitschrift für zellforschung und mikroskopische anatomie*, vol. 105, no. 2, pp. 234–241.

- Carr, K.E. & McGadey, J. 1974, "Staining of biological material for the scanning electron microscope", *Journal of microscopy*, vol. 100, no. 3, pp. 323–330.
- Carr, K.E. & Seed, T.M. 1983, "Ultrastructural effects of radiation on tissues and cells" (excerpt), soft-cover book, 17 papers, Scanning electron microscopy.
- Carr, K. E. & Toner, P. G. 1968, "Scanning electron microscopy of rat intestinal villi", The lancet, vol. 2, pp. 570-571,
- Carr, K.E. & Toner, P.G. 1972, "Surface studies of acute radiation injury in the mouse intestine", *Virchows archiv B zell-pathologie*, vol. 11, no. 1, pp. 201–210.
- Carr, K.E. & Toner, P.G. 1982. "Cell structure. An introduction to biomedical electron microscopy" (3rd edition, excerpt), Churchill Livingstone.
- Carr, K.E., & Toner, P.G. 1984, "Morphology of the intestinal mucosa", *Handbook of experimental pharmacology*, vol. 70, no. 1, pp. 1–50.
- Carr, K.E. & Whur, P. 1968, "Ultrastructure of globule leucocyte inclusions in the rat and mouse", *Zeitschrift für zellforschung und mikroskopische anatomie*, vol. 86, no. 2, pp. 153–162.
- Carr, K.E., Arbuthnott, J.P., Toner, P.G. & Gemmell, C.G. 1967, "The fine structure of rat liver cells in suspension", Zeitschrift für zellforschung und mikroskopische anatomie, vol. 79, no. 2, pp. 265–271.
- Carr, K.E., Bullock, C., Ryan, S.S., McAlinden, M.G. & Boyle, F.C. 1991a, "Radioprotectant effects of atropine on small intestinal villous shape", *Journal of submicroscopic cytology and pathology*, vol. 23, no. 4, pp. 569–577.
- Carr, K.E., Dayan, A.D., Florence, A.T., Goodlad, R.A., Hodges, G.M. & Rowlands, B.J. 1996b, "Symposium on particle translocation across epithelial membranes", *Journal of anatomy*, vol. 189, no. 3, p. 475.
- Carr, K.E., Ellis, S., Hayes, T.L. & Ainsworth, E.J. 1989, "Multiview technique for SEM study of intestinal villi", *Scanning*, vol. 11, no. 2, pp. 109–110.
- Carr, K.E., Ellis, S. & Michalowski, A. 1986b, "Surface studies of duodenal lesions induced by thoracic irradiation", *Scanning electron microscopy*, vol. 1, pp. 209–219.
- Carr, K.E., Hamlet, R., Nias, A.H.W., Boyle, F.C. & Fife, M.G. 1985, "Stromal damage in the mouse small intestine after Cobalt 60 gamma or D-T neutron irradiation", *Scanning electron microscopy*, vol. 5, pp. 1615–1621.
- Carr, K.E., Hamlet, R., Nias, A.H.W. & Watt, C. 1979b, "Lack of correlation between villus and crypt damage in irradiated mouse intestine", *British journal of radiology*, vol. 52, no. 618, pp. 485–493.
- Carr, K.E., Hamlet, R., Nias, A.H.W. & Watt, C. 1981a, "Multinucleate giant enterocytes in small intestinal villi after irradiation", *Journal of microscopy*, vol. 123, no. 2, pp. 169–176.

- Carr, K.E., Hamlet, R., Nias, A.H.W. & Watt, C. 1983c, "Damage to the surface of the small intestinal villus: an objective scale of assessment of the effects of single and fractionated radiation doses", *British journal of radiology*, vol. 56, no. 667, pp. 467–475.
- Carr, K.E., Hamlet, R., Nias, A.H.W. & Watt, C. 1984c, "Morphological differences in the response of mouse small intestine to radiobiologically equivalent doses of X and neutron irradiation", *Scanning electron microscopy*, vol. 5, no. 1, pp. 445–454.
- Carr, K.E., Hamlet, R. & Watt, C. 1981c, "Scanning electron microscopy, autolysis, and irradiation as techniques for studying small intestinal morphology", *Journal of microscopy*, vol. 123, no. 2, pp. 161–168.
- Carr, K.E., Hayes, T.L., Abbas, B. & Ainsworth, E.J. 1990, "Collared crypts in irradiated small intestine", *Journal of submicroscopic cytology and pathology*, vol. 22, no. 2, pp. 265–271.
- Carr, K.E., Hayes, T.L., Hume, S.P. & Kamel, H.M.H. 1983b, "Qualitative and quantitative SEM of gastrointestinal tract", *Proceedings of the 41st annual meeting of the electron microscope society of America*, pp. 496-499.
- Carr, K.E., Hayes, T.L., Indran, M., Bastacky, S.J., McAlinden, G., Ainsworth, E.J. & Ellis, S. 1987b, "Morphological criteria for comparing effects of X-rays and neon ions on mouse small intestine", *Scanning microscopy*, vol. 1, no. 2, pp. 799–809.
- Carr, K.E., Hayes, T.L., McKoon, M., Bastacky, S.J. & Kamel, H.M. 1984b, "Etched surfaces of plastic embedded and frozen hydrated gastrointestinal tissue", *Journal of submicroscopic cytology*, vol. 16, no. 2, pp. 219–226.
- Carr, K.E., Hayes, T.L., McKoon, M., Sprague, M. & Bastacky, S.J. 1983a, "Low temperature scanning electron microscope studies of mouse small intestine", *Journal of microscopy*, vol. 132, no. 2, pp. 209–217.
- Carr, K.E., Hayes, T.L., Watt, A., Bastacky, S.J., Klein, S. & Fife, M. 1986a, "Specimen handling and data interpretation in X-ray microanalysis of frozen hydrated etched gastrointestinal tract", *Journal of electron microscopy technique*, vol. 4, no. 4, pp. 371–379.
- Carr, K.E., Hayes, T.L., Watt, A., Klein, S., McKoon, M., Bastacky, S.J. & Ellis, S. 1987a, "Biological variation in cryomicroanalytical data from mouse small intestinal villi", *Journal of electron microscopy technique*, vol. 5, no. 1, pp. 65–74.
- Carr, K.E., Hazzard, R.A., Reid, S. & Hodges, G.M. 1996c, "The effect of size on uptake of orally administered latex microparticles in the small intestine and transport to mesenteric lymph nodes", *Pharmaceutical research,* vol. 13, no. 8, pp. 1205–1209.
- Carr, K.E., Hume, S.P., Ettarh, R., Carr, E.A. & McCullough, J.S., 1995, "Radiation-induced changes to epithelial and non-epithelial tissue", in *Radiation and the gastrointestinal tract*, Eds. Dubois A., King, G.L. & Livengood. D.R., Taylor and Francis, pp. 113–128.

- Carr, K.E., Hume, S.P., Marigold, J.C.L. & Michalowski, A. 1982b, "Scanning and transmission electron microscopy of the damage to small intestinal mucosa following X-irradiation or hyperthermia", *Scanning electron microscopy*, no. 1, pp. 393–402.
- Carr, K.E., Hume, S.P., Nelson, A.C., O'Shea, O., Hazzard, R.A. & McCullough, J.S. 1996a, "Morphological profiles of neutron and X-Irradiated small intestine", *Journal of radiation research*, vol. 37, no. 1, pp. 38-48.
- Carr, K.E., Joffe, S.N., Toner, P.G. & Watt, C. 1979a, "A preliminary study of scanning electron microscopic changes of the duodenum during healing of duodenal ulcers", *Scandinavian journal of gastroenterology*, vol. 14, supplement 54, pp. 78–83.
- Carr, K.E., Kamel, H.M.H., Toner, P.G., McGadey, J. & Wong, A. 1984a, "Correlative scanning electron microscopy in gastrointestinal pathology", *Scanning electron microscopy*, vol. 5, pp. 761–772.
- Carr, K.E., McCullough, J.S., Brennan, P., Hayes, T.L., Ainsworth, E.J. & Nelson, A.C. 1994, "Heavy ion induced changes in small intestinal parameters", *Advances in space research*, vol. 14, no. 10, pp. 521–530.
- Carr, K.E., McCullough, J.S., Gilmore, R. S., Abbas, B., Hume, S.P., Nelson, A.C., Hayes, T.L. & Ainsworth, E.J. 1993, "Effects of neon ions, neutrons and X rays on small intestine", in *Biological effects and physics of solar and* galactic cosmic radiation. NATO ASI Series (Series A: life sciences), Eds. Swenberg, C.E., Horneck, G. & Stassinopoulos, E.G., Springer, Boston, MA, Plenum Press, New York, pp. 203–216.
- Carr, K.E., McCullough, S.J., Hazzard, R.A., Brennan, P., McGuiness, C., Seed, T. & Hodges, G.M. 1997, "Structural and functional profiles of mammalian small intestine", in *Recent advances in microscopy of cells, tissues and organs*, Ed. Motta P.M., pp. 381-386.
- Carr, K.E., McCullough, J.S., McCallion, W., Steele, J., Nelson, A.C. & Hanson, W. 1991b, "Morphological effects of radioprotectants on different intestinal cells", in *Radiation research: a twentieth century perspective*, Eds. Chapman, J.D., Dewey, W.C. & Whitmore, G.F., Academic Press Inc, Toronto, vol. 1, p. 370.
- Carr, K.E., McCullough, J.S., Nelson, A.C., Hume, S.P., Nunn, S. & Kamel, H.H.M. 1992b, "Relationship between villous shape and mural structure in neutron irradiated small intestine", *Scanning microscopy*, vol. 6, no. 2, pp. 561–572.
- Carr, K.E., McCullough, J.S., Nunn, S., Hume, S.P. & Nelson, A.C. 1991c, "Neutron and X-ray effects on small intestine summarized by using a mathematical model or paradigm", *Proceedings of the royal society B: biological sciences*, vol. 243, no. 1308, pp. 187–194.
- Carr, K.E., Nelson, A.C., Hume, S.P. & McCullough, J.S. 1992a, "Characterization through a data display of the different cellular responses in X-irradiated small intestine", *Journal of radiation research*, vol. 33, no. 2, pp. 163–177.
- Carr, K.E., Shaw Dunn, J. & Toner, P.G. 1974, "Scanning electron microscopy of the alimentary tract", *Scottish medical journal*, vol. 19, no. 5, pp. 211–220.

- Carr, K.E., Smyth, S.H., McCullough, M.T., Morris, J.F. & Moyes, S.M. 2012, "Morphological aspects of interactions between microparticles and mammalian cells: intestinal uptake and onward movement", *Progress in histochemistry and cytochemistry*, vol. 46, no. 4, pp. 185–252.
- Carr, K.E., Toner, P.G., McLay, A.L.C. & Hamlet, R. 1981d, "The ultrastructure of some gastrointestinal lesions in experimental animals and man", *Scandinavian journal of gastroenterology, supplement*, vol. 16, pp. 107–128.
- Carr, K.E., Toner, P.G. & Saleh, K.M. 1982a, "Scanning electron microscopy", Histopathology, vol. 6, no. 1, pp. 3-24.
- Carr, K.E., Wong, A.L., Young, D.G., Toner, P.G. & Watt, C. 1981b, "Scanning electron microscopy and resin histology of large bowel biopsies", *Scottish medical journal*, vol. 26, no. 2, pp. 103–114.
- Carr, K.E., Youngman, M.J., Barr, M., Colhoun, L. & Harrison, J.D. 2001, "Visualisation of fused aluminoscilicate microparticles taken up by intestinal tissue", *Journal of anatomy*, vol. 199, pp. 225–226.
- Carter, H.W. 1980, "Clinical applications of scanning electron microscopy (SEM) in North America with emphasis on SEM's role in comparative microscopy", *Scanning electron microscopy*, vol. 3, pp. 115–120, 154.
- Cartwright-Shamoon, J., Dickson, G.R., Dodge, J. & Carr, K.E. 1995, "Morphological aspects of particle translocation in vivo following ingestion of the yeast Saccharomyces boulardii", *Journal of drug targeting,* vol. 3, no. 1, pp. 61–63.
- Cartwright-Shamoon, J., Dickson, G.R., Dodge, J. & Carr, K.E. 1996, "Uptake of yeast (Saccharomyces boulardii) in normal and rotavirus treated intestine", *Gut*, vol. 39, no. 2, pp. 204–209.
- Casteels, R. & Login, I.S. 1983, "Reserpine has a direct action as a calcium antagonist on mammalian smooth muscle cells", *The journal of physiology*, vol. 340, no. 1, pp. 403–414.
- Cereijido, M., Shoshani, L. & Contreras, R.G. 2000, "Molecular physiology and pathophysiology of tight junctions.
 I. Biogenesis of tight junctions and epithelial polarity", *American journal of physiology gastrointestinal and liver physiology*, vol. 279, no. 3, pp. G477–G482.
- Chachalis, D., Reddy, K.N., Elmore, C.D. & Steele, M.L. 2001, "Herbicide efficacy, leaf structure, and spray droplet contact angle among Ipomoea species and small flower morning glory". *Weed science*, vol. 49, no. 5, pp. 628–634.
- Chamberlain, S.M., Kirk, J. & Nias, A.H.W. 1980, "Variation in neutron RBE values for human lymphocytes", International journal of radiation oncology, biology and physics, vol. 6, pp. 341–344.
- Chandler, C.E., Zaccaro, L.M. & Moberly, J.B. 1993, "Transepithelial transport of cholyltaurine by Caco-2 cell monolayers is sodium dependent", *American journal of physiology – gastrointestinal and liver physiology,* vol. 264, no. 6, pp. G1118–G1125.
- Chang, F.Y., Lee, S., Yeh, G. & Wang, P. 1995, "Influence of pregnancy and uterine weight on rat gastrointestinal transit", *Journal of gastroenterology and hepatology*, vol. 10, no. 5, pp. 585–588.

- Ch'Ang, H.J., Lin, L.M., Chang, P.Y., Luo, C.W., Chang, Y.H., Chou, C.K. & Chen, H.H. 2012, "Bone marrow transplantation enhances trafficking of host-derived myelomonocytic cells that rescue intestinal mucosa after whole body radiation", *Radiotherapy and oncology*, vol. 104, no. 3, pp. 401-407.
- Chen, D., Hosm, H., Tanaka, K. & Murakami, G. 1995, "Postnatal development of lymphoid follicles in rat Peyer's patches, with special reference to increased follicle number", *Archives of histology and cytology*, vol. 58, no. 3, pp. 335-343.
- Cheng, H. & Leblond, C.P. 1974, "Origin, differentiation and renewal of the four main epithelial cell types in the mouse small intestine I. Columnar cell", *American journal of anatomy*, vol. 141, no. 4, pp. 461-479.
- Chikui, T., Kitamoto, E., Kawano, S., Sugiura, T., Obara, M., Simonetti, A.W., Hatakenaka, M., Matsuo, Y., Koga, S., Ohga, M., Nakamura, K. & Yoshiura, K. 2012, "Pharmacokinetic analysis based on dynamic contrastenhanced MRI for evaluating tumor response to preoperative therapy for oral cancer", *Journal of magnetic resonance imaging*, vol. 36, no. 3, pp. 589–597.
- Chomette, G., Daburon, F., Auriol, M. & Garnier, H. 1977, "High dose irradiation in the pig small intestine histoenzymology and electron microscopic study", *Virchows archiv abteilung B cell pathology*, vol. 23, no. 3, pp. 237–256.
- Clarke, J.A. & Salsbury, A.J. 1967, "Surface ultramicroscopy of human blood cells", *Nature*, vol. 215, no. 5099, pp. 402–404.
- Classen, J., Belka, C., Paulsen, F., Budach, W., Hoffmann, W. & Bamberg, M. 1998, "Radiation-induced gastrointestinal toxicity. Pathophysiology, approaches to treatment and prophylaxis", *Strahlentherapie und onkologie*, vol. 174, supplement 3, pp. 82–84.
- Cobb, R.A. & Galland, R.B. 1990, "Early reports of radiation-damaged gut", in *Radiation enteritis*, Eds. Galland, R.B. & Spencer, J., Edward Arnold, London, pp. 1–6.
- Cocco, A.E., Dohrmann, M.J. & Hendrix, T.R. 1966, "Reconstruction of normal jejunal biopsies: three dimensional histology", *Gastroenterology*, vol. 51, pp. 24–31.
- Coleman, C.N. & Harris, J.R. 1998, "Current scientific issues related to clinical radiation oncology", *Radiation research*, vol. 150, no. 2, pp. 125–133.
- Cook, G.T., Carr, K.E. & Duncan, H.J. 1979, "The influence of morphological differences in bracken pinnules on the foliar uptake of aminotriazole", *Annals of applied biology*, vol. 93, no. 3, pp.311–317.
- Coppi, G., Montanari, M., Rossi, T., Bondi, M. & Iannuccelli, V. 2010, "Cellular uptake and toxicity of microparticles in a perspective of polymyxin B oral administration", *International journal of pharmaceutics*, vol. 385, no. 1– 2, pp. 42–46.

- Cornelis, G., Hund-Rinke, K., Kuhlbusch, T., van den Brink, N. & Nickel, C. 2014, "Fate and bioavailability of engineered nanoparticles in soils: a review, *Critical reviews in environmental science and technology*, vol. 44, no. 24, pp. 2720–2764.
- Cornes, J.S. 1965, "Number, size and distribution of Peyer's patches in the human small intestine", *Gut*, vol. 6, pp. 225–229.
- Couvreur, P. & Puisieux, F. 1993, "Nano- and microparticles for the delivery of polypeptides and proteins", *Advanced drug delivery reviews*, vol. 10, no. 2-3, pp. 141-162.
- Crewe, A.V., 1971, "A high-resolution scanning electron microscope", *Scientific American*, vol. 224, no. 4, pp. 26-35.
- Croci, T., Landi, M., Galzin, A. & Marini, P. 2003, "Role of cannabinoid CB 1 receptors and tumor necrosis factor-α in the gut and systemic anti-inflammatory activity of SR 141716 (Rimonabant) in rodents", *British journal of pharmacology*, vol. 140, no. 1, pp. 115–122.
- Cuddihy, R.G., Finch, G.L., Newton, G.J., Hahn, F.F., Mewhinney, J.A., Rothenberg, S.J. & Powers, D.A. 1989, "Characteristics of radioactive particles released from the Chernobyl nuclear reactor", *Environmental science and technology*, vol. 23, no. 1, pp. 89–95.
- Currie, L.J., Sharpe, J.R. & Martin, R. 2001, "The use of fibrin glue in skin grafts and tissue-engineered skin replacements: a review", *Plastic and reconstructive surgery*, vol. 108, no. 6, pp. 1713–1726.
- Damgé, C., Aprahamian, M., Marchais, H., Benoit, J.P. & Pinget, M. 1996, "Intestinal absorption of PLAGA microspheres in the rat", *Journal of anatomy*, vol. 189, no. 3, pp. 491–501.
- Dannenberg, A.M., Walter, P.C. & Kapral, F.A. 1963, "A histochemical study of phagocytic and enzymatic functions of rabbit mononuclear and polymorphonuclear exudate cells, and alveolar macrophages", *Journal of immunology*, vol. 90, pp. 448–465.
- Davidson, M. & Wasserman, R. 1966, "The irritable colon of childhood", Journal of pediatrics, vol. 60, pp. 1207–1238.
- De Carvalho, A.D., De Souza, W. & Morgado-Díaz, J.A. 2006, "Morphological and molecular alterations at the junctional complex in irradiated human colon adenocarcinoma cells, Caco-2", *International journal of radiation biology*, vol. 82, no. 9, pp. 658–668.
- Delie, F. 1998, "Evaluation of nano- and microparticle uptake by the gastrointestinal tract", *Advanced drug delivery reviews*, vol. 34, no. 2–3, pp. 221–233.
- Delie, F. & Rubas, W. 1997, "A human colonic cell line sharing similarities with enterocytes as a model to examine oral absorption: advantages and limitations of the Caco-2 model", *Critical reviews in therapeutic drug carrier systems*, vol. 14, no. 3, pp. 221-286.

- Demling, L., Becker, V. & Classen, M. 1969, "Examinations of the mucosa of the small intestine with the scanning electron microscope", *Digestion*, vol. 2, no. 1, pp. 51–60.
- Derakhshandeh, K., Hochhaus, G. & Dadashzadeh, S. 2011, "In-vitro cellular uptake and transport study of 9nitrocamptothecin PLGA nanoparticles across caco-2 cell monolayer model", *Iranian journal of pharmaceutical research*, vol. 10, no. 3, pp. 425-434.
- des Rieux, A., Fievez, V., Théate, I., Mast, J., Préat, V. & Schneider, Y.J. 2007, "An improved in vitro model of human intestinal follicle-associated epithelium to study nanoparticle transport by M cells", *European journal of pharmaceutical sciences*, vol. 30, no. 5, pp. 380–391.
- des Rieux, A., Ragnarsson E.G., Gullberg E., Préat V., Schneider Y.J. & Artursson P. 2005, "Transport of nanoparticles across an in vitro model of the human intestinal follicle associated epithelium", *European journal of pharmaceutical sciences*, vol. 25 no. 4-5, pp. 455-465.
- Dickson, G.B., Kamel, H.M.H., Hume, S.P., Jaber, M. & Carr, K.E. 1992, "Early effects on the morphology of mouse small intestine of single or combined modality treatment with hyperthermia and X-irradiation", *Scanning Microscopy*, vol. 6, no.3, pp. 847 – 854.
- Dickson, G.R., McKenna, S., McColl, K. & Carr, K.E. 1989, "Examination of mucosal surfaces by scanning electron microscopy before and after removal of debris", *Journal of microscopy*, vol. 153, no. 1, pp. 75–79.
- Domagala, W., Kahan, A.V, & Koss, L.G. 1979, "A simple method of preparation and identification of cells for scanning electron microscopy", *Acta cytologica*, vol. 23, pp. 140–146.
- Domènech, A., Pasquinelli, G., de Giorgio, R., Gori, A., Bosch, F., Pumarola, M. & Jiménez, M. 2011, "Morphofunctional changes underlying intestinal dysmotility in diabetic RIP-I/hIFNβ transgenic mice", *International journal of experimental pathology*, vol. 92, no. 6, pp. 400-412.
- Dörr, W, 1995, Impact of time between X-ray fractions on acute changes in the function of the urinary bladder in the mouse, *Radiation research*, vol. 141, pp. 219–222.
- , W., Alheit, H., Appold, S., Enghardt, W., Haase, M., Haberer, T., Hinz, R., Jäkel, O., Kellerer, A.M., Krämer, M., Kraft, G., Kumpf, R., Nitzsche, H., Scholz, M., Voigtmann, L. & Herrmann, T. 1999, "Response of pig lung to irradiation with accelerated 12^c-ions", *Radiation and environmental biophysics*, vol. 38, no. 3, pp. 185–194.
- Dourov, N. 1984, "Scanning electron microscopy contribution in oral pathology", *Scanning electron microscopy*, vol. 1, pp.243–248.
- Doyle-McCullough, M., Smyth, S.H., Moyes, S.M. & Carr, K.E. 2007, "Factors influencing intestinal microparticle uptake in vivo", *International journal of pharmaceutics*, vol. 335, no. 1–2, pp. 79–89.
- Dublineau, I., Ksas, B., Aigueperse, J., Gourmelon, P. & Griffiths, N.M. 1998, "In vivo alterations of fluid and electrolyte fluxes in rat colon by gamma irradiation", *Digestive diseases and sciences*, vol. 43, no. 3, pp. 652– 662.

- Dublineau, I., Ksas, B., Joubert, C., Aigueperse, J., Gourmelon, P. & Griffiths, N.M. 2002, "Alterations in water and electrolyte absorption in the rat colon following neutron irradiation: influence of neutron component and irradiation dose", *International journal of radiation biology*, vol. 78, no. 12, pp. 1127–1138.
- Dublineau, I., Lebrun, F., Grison, S. & Griffiths, N.M. 2004, "Functional and structural alterations of epithelial barrier properties of rat ileum following X-irradiation", *Canadian journal of physiology and pharmacology,* vol. 82, no. 2, pp. 84–93.
- Dubois, A., King, G.L. & Livengood, D.R. Eds., 1995, *Radiation and the gastrointestinal tract,* CRC Press, Boca Raton, FL.
- Durmus-Altun, G., Vatansever, U., Arzu Vardar, S., Altaner, S. & Dirlik, B. 2011, "Scintigraphic evaluation of small intestinal transit in the streptozotocin induced diabetic rats", *Hippokratia*, vol. 15, no. 3, pp. 262–264.
- Ebel, J.P. 1990, "A method for quantifying particle absorption from the small intestine of the mouse", *Pharmaceutical research: an official journal of the American association of pharmaceutical scientists*, vol. 7, no. 8, pp. 848-851.
- Echlin, P. 1968, "The use of the scanning reflection electron microscope in the study of plant and microbial material", *Journal of the Royal microscopical society*, vol. 88, no. 3, pp. 407–418.
- Echlin, P. 1978, "Low temperature scanning electron microscopy: a review", *Journal of microscopy*, vol. 112, pp. 47–61.
- Echlin, P., Lai, C.E. & Hayes, T.L. 1981, "The distribution relative concentration of potassium in the root tips of Lemna minor L. analysed by low temperature X-ray microanalysis", *Scanning electron microscopy*, vol. 2, pp. 489-498.
- Eckhaus, A.A., Fish, J.S., Skarja, G., Semple, J.L. & Sefton, M.V. 2008, "A preliminary study of the effect of poly(methacrylic acid-co-methyl methacrylate) beads on angiogenesis in rodent skin grafts and the quality of the panniculus carnosus", *Plastic and reconstructive surgery*, vol. 122, no. 5, pp. 1361–1370.
- Edemskiĭ, A.I. 1983, "Morphologic features of the structure of the major duodenal papilla in healthy subjects and pathology of the biliopancreatic region", *Arkhiv patologii*, vol. 45, no. 9, pp. 42–48.
- Edemskiĭ, A.I. & Svishchev, A.V. 1986, "Anatomo-stereologic characteristics of the mucous membrane of the major duodenal papilla", *Arkhiv anatomii, gistologii i embriologii*, vol. 90, no. 3, pp. 61–66.
- Egawa, J. & Ishioka, K. 1978, "Radiation effects on the fine blood vessels in abdominal organs of mice", *Acta radiologica. Oncology radiation therapy physics and biology*, vol. 17, no. 5, pp. 414–422.
- Eigenbrodt, M.L., Eigenbrodt, E.H., Kneitz, J.S. & Thiele, D.L. 1995, "Cellular blebbing in superficial colonic epithelial cells occurring with murine graft-versus-host disease", *Virchows archiv*, vol. 426, no. 1, pp. 61-67.

- Eldridge, J.H., Hammond, C.J., Meulbroek, J.A., Staas, J.K., Gilley, R.M. & Tice, T.R. 1990, "Controlled vaccine release in the gut-associated lymphoid tissues, I, orally administered biodegradable microspheres target the Peyer's patches", *Journal of controlled release*, vol. 11, no. 1–3, pp. 205–214.
- Elliott, T.B., Ledney, G.D., Harding, R.A., Henderson, P.L., Gerstenberg, H.M., Rotruck, J.R., Verdolin, M.H., Stille, C.M. & Krieger, A.G. 1995, "Mixed-field neutrons and γ photons induce different changes in ileal bacteria and correlated sepsis in mice", *International journal of radiation biology*, vol. 68, no. 3, pp. 311–320.
- Epperly, M.W., Shen, H., Jefferson, M. & Greenberger, J.S. 2004, "In vitro differentiation capacity of esophageal progenitor cells with capacity for homing and repopulation of the ionizing irradiation-damaged esophagus", *In vivo*, vol. 18, no. 6, pp. 675–686.
- Eriksson, B., Johnson, L. & Lundqvist, P.G. 1983, "Ultrastructural aspects of capillary function in irradiated bowel: an experimental study in the cat", *Scandinavian journal of gastroenterology*, vol. 18, no. 4, pp. 473–480.
- Ermak, T.H., Dougherty, E.P., Bhagat, H.R., Kabok, Z. & Pappo, J. 1995, "Uptake and transport of copolymer biodegradable microspheres by rabbit Peyer's patch M cells", *Cell and tissue research*, vol. 279, no. 2, pp. 433-436.
- Ettarh, R.R. & Carr, K.E. 1993, "Structural and morphometric analysis of murine small intestine after indomethacin administration", *Scandinavian journal of gastroenterology*, vol. 28, no. 9, pp. 795-802.
- Ettarh, R.R. & Carr, K.E. 1996a, "Morphometric analysis of the small intestinal epithelium in the indomethacintreated mouse", *Journal of anatomy*, vol. 189, no. 1, pp. 51–56.
- Ettarh, R.R. & Carr, K.E. 1996b, "Ultrastructural observations on the peritoneum in the mouse", *Journal of anatomy*, vol. 188, no. 1, pp. 211–215.
- Ettarh, R.R. & Carr, K.E. 1997, "A morphological study of the enteric mucosal epithelium in the streptozotocindiabetic mouse", *Life Sciences*, vol. 61, no. 18, pp. 1851–1858.
- Ettarh, R.R., Hodges, G.M. & Carr, K.E. 2000, "Radiation effects in the small bowel of the diabetic mouse", *International journal of radiation biology*, vol. 76, no. 2, pp. 241–248.
- Fabian, G. 1983, "Persorption the way of large sized corpuscle particles via the lymphatic system", *Lymphology,* vol. 16, no. 1, pp. 43–48.
- Fajardo, L.F. 1989, "Morphologic patterns of radiation injury 1", in *Radiation tolerance of normal tissues*, Karger Publishers, vol. 23, pp. 75–84.
- Fajardo, L.F., Berthrong, M. & Anderson, R.E. 2001, Radiation pathology, Oxford University Press.
- Fanning, A. S., Jameson, B. J., Jesaitis, L. A. & Anderson, J. M. 1998, "The tight junction protein ZO-1 establishes a link between the transmembrane protein occludin and the actin cytoskeleton", *Journal of biological chemistry*, vol. 273, no. 45, pp. 29745-29753.

- Fanning, A. S., Ma, T. Y. & Anderson, J. M. 2002, "Isolation and functional characterization of the actin binding region in the tight junction protein Z0-1", *The FASEB journal: official publication of the federation of American societies for experimental biology*, vol. 1, no. 13, pp. 1835–1837.
- Farquar, M.G. & Palade, G.E. 1963, "Junctional complexes in various epithelia", *Journal of cell biology*, vol. 17, pp. 375-412.
- Fastinger, N.D., Karr-Lilienthal, L.K., Spears, J.K., Swanson, K.S., Zinn, K.E., Nava, G.M., Ohkuma, K., Kanahori, S., Gordon, D.T. & Fahey, G.C. Jr. 2008, "A novel resistant maltodextrin alters gastrointestinal tolerance factors, fecal characteristics, and fecal microbiota in healthy adult humans" *Journal of the American college of nutrition*, vol. 27, no. 2, pp. 356-366.
- Fatemi, S.H., Antosh, M., Cullan, G.M. & Sharp, J.G. 1985, "Late ultrastructural effects of heavy ions and gamma irradiation in the gastrointestinal tract of the mouse", *Virchows archiv B cell pathology including molecular pathology*, vol. 48, no. 1, pp. 325–340.
- Faucheron, J.L. 1999, "Toxicity of non-steroidal anti-inflammatory drugs in the large bowel", *European journal of gastroenterology and hepatology*, vol. 11, no. 4, pp. 389–392.
- Fenger, C. & Knoth, M. 1981, "The anal transitional zone: a scanning and transmission electron microscopic investigation of the surface epithelium", *Ultrastructural pathology*, vol. 2, pp. 163–173.
- Ferguson, A. & Parrott, D.M.V. 1973, "Histopathology and time course of rejection of allografts of mouse small intestine", *Transplantation*, vol. 15, no. 6, pp. 546–554.
- Ferguson, A., Carr, K.E., Macdonald, T.T. & Watt, C. 1978, "Hypersensitivity reactions in the small intestine, iv influence of allograft rejection on small intestinal mucosal architecture: a scanning and transmission electron microscope study", *Digestion*, vol. 18, no. 1–2, pp. 56–63.
- Ferguson, A., Maxwell, J.D. & Carr, K.E. 1969, "Progressive changes in the small-intestinal villous pattern with increasing length of gestation", *The journal of pathology*, vol. 99, no. 1, pp. 87–91.
- Ferguson, P.C., Boynton, E.L., Wunder, J.S., Hill, R.P., O'Sullivan, B., Sandhu, J.S. & Bell, R.S. 1999, "Intradermal injection of autologous dermal fibroblasts improves wound healing in irradiated skin", *Journal of surgical research*, vol. 85, no. 2, pp. 331-338.
- Ferraro, V., Ferreira Jorge, R., Cruz, I.B., Antunes, F., Sarmento, B., Castro, P.M.L. & Pintado, M.E. 2014, "In vitro intestinal absorption of amino acid mixtures extracted from codfish (Gadus morhua L.) salting wastewater", *International journal of food science and technology*, vol. 49, no. 1, pp. 27-33.
- Ferreira, M.N. & Leblond, C.P. 1971, "Argentaffin and other "endocrine" cells of the small intestine in the adult mouse", II. Renewal. *Developmental dynamics*, vol. 131, no. 3, pp. 331–351.

- Ferruzza, S., Scarino, M.L., Rotilio, G., Ciriolo, M.R., Santaroni, P., Muda, A.O. & Sambuy, Y. 1999, "Copper treatment alters the permeability of tight junctions in cultured human intestinal Caco-2 cells", *American journal of physiology - gastrointestinal and liver physiology*, vol. 277, no. 6, pp. G1138–G1148.
- Flessner, M.F., Lofthouse, J. & Williams, A. 2001, "Increasing peritoneal contact area during dialysis improves mass transfer", *Journal of the American society of nephrology*, vol. 12, no. 10, pp. 2139–2145.
- Flickinger, J.C., Bloomer, W.D. & Kinsella, T.J. 1990, "Intestinal tolerance of radiation injury", in *Radiation enteritis,* Eds. Galland, R.B. & Spencer, J., Edward Arnold, London, pp. 49–65.
- Florence, A.T. 1997, "The oral absorption of micro- and nanoparticulates: neither exceptional nor unusual", *Pharmaceutical research*, vol. 14, no. 3, pp. 259–266.
- Florence, A.T., Hillery, A.M., Hussain, N. & Jani, P.U. 1995, "Factors affecting the oral uptake and translocation of polystyrene nanoparticles: histological and analytical evidence", *Journal of drug targeting*, vol. 3, no. 1, pp. 65–70.
- Forrester, J.C., Hunt, T.K., Hayes, T.L. & Pease, R.F.W. 1969, "Scanning electron microscopy of healing wounds", *Nature*, vol. 221, no. 5178, pp. 373–374.
- Forsgren, S., Höckerfelt, U., Norrgård, O., Henriksson, R. & Franzén, L. 2000, "Pronounced substance P innervation in irradiation-induced enteropathy a study on human colon", *Regulatory peptides*, vol. 88, no. 1–3, pp. 1–13.
- Foureau, D.M., Mielcarz, D.W., Menard, L.C., Schulthess, J., Werts, C., Vasseur, V., Ryffel, B., Kasper, L.H. & Buzoni-Gatel, D. 2010, "TLR9-dependent induction of intestinal α-defensins by Toxoplasma gondii", *Journal of immunology*, vol. 184, no. 12, pp. 7022–7029.
- François, A., Dublineau, I., Lebrun, F., Ksas, B. & Griffiths, N.M. 1999, "Modified absorptive and secretory processes in the rat distal colon after neutron irradiation: in vivo and in vitro studies", *Radiation research*, vol. 151, no. 4, pp. 468–478.
- Freedman, B.J. 1991, "Persorption of raw starch: a cause of senile dementia?", *Medical hypotheses*, vol. 35, no. 2, pp. 85-87.
- Freeman, S.L., Hossain, M. & MacNaughton, W.K. 2001, "Radiation-induced acute intestinal inflammation differs following total-body versus abdominopelvic irradiation in the ferret", *International journal of radiation biology*, vol. 77, no. 3, pp. 389–395.
- Freudenhieim, J.L., Graham, S., Marshall, J.R., Haughey, B.P. & Wilkinson, G. 1990, "A case-control study of diet and rectal cancer in western New York", *American journal of epidemiology*, vol. 131, no. 4, pp. 612–624.

Friberg, L. G. 1980, "Effects of irradiation on the small intestine of the rat", Ph.D. Thesis, Lund University.

- Friedman, M., Ryan, U. S., Davenport, W. C., Chaney, E. L., Strickland, D. L. & Kwock, L. 1986, "Reversible alterations in cultured pulmonary artery endothelial cell monolayer morphology and albumin permeability induced by ionizing radiation", *Journal of cellular physiology*, vol. 129, no. 2, pp. 237–249.
- Fuchs, W. & Fuchs, H. 1980, "The use of frozen-hydrated bulk specimens for X-ray microanalysis", *Scanning electron microscopy*, vol. 2, pp. 371-382.
- Fujimoto, K. 1995, "Freeze-fracture replica electron microscopy combined with SDS digestion for cytochemical labeling of integral membrane proteins application to the immunogold labeling of intercellular junctional complexes", *Journal of cell science*, vol. 108, no. 11, pp. 3443–3449.
- Fukuo, Y., Shinohara, H. & Matsuda, T. 1990, "The distribution of lymphatic stomata in the diaphragm of the golden hamster", *Journal of anatomy*, vol. 169, pp. 13–21.
- Gadegone, M.M. & Bhiwagde, D.A. 1981, "Mucosubstances in the cervix uteri of the Indian langur monkey, Presbytis entellus entellus (Dufrense)", *Zeitschrift fur mikroskopisch-anatomische forschung – abteilung 2*, vol. 95, no. 2, pp. 223–229.
- Galland, R.B. & Spencer, J. Eds, 1990, *Radiation enteritis*, Edward Arnold, London.
- Galloway, D.J., Jarrett, F., Boyle, P., Indran, M., Carr, K., Owen, R.W. & George, W.D. 1987, "Morphological and cell kinetic effects of dietary manipulation during colorectal carcinogenesis", *Gut*, vol. 28, no. 6, pp. 754–763.
- Geisinger, K.R., Scobey, M.W., Northway, M.G., Cassidy, K.T., & Castell, D.O., 1990, "Radiation-induced proctitis cystica profunda in the rat", *Digestive diseases and sciences*, vol. 35, no. 7, pp. 833-9.
- Geissinger, H.D. 1972, "The use of silver nitrate as a stain for scanning electron microscopy of arterial intima and paraffin sections of kidney", *Journal of microscopy*, vol. 95, no. 3, pp. 471–481.
- Gentilhomme, E., Reano, A., Pradel, D., Bergier, J., Schmitt, D. & Neveux, Y. 1998, "In vitro dermal intoxication by bis(chloroethyl)sulfide. Effect on secondary epidermization", *Cell biology and toxicology*, vol. 14, no. 1, pp. 1–11.
- Gerard, J.P., Coquard, R., Romestaing, P., Ardiet, J.M., Rocher, F.P., Trillet Lenoir, V. & Sentenac, I. 1995,
 "Prevention of radiation related complications in the treatment of rectal adenocarcinoma the importance of the dose volume relationship", *Tumori*, vol. 81, no. 3 supplement, pp. 114–116.
- Giampaolo, C., Poulsen, S.S., Szabo, S., Reynolds, E.S. & Feldman, D. 1978. "Stereomicroscopic and ultrastructural characterization of propionitrile-induced duodenal ulcer in the rat" *Laboratory investigation*, vol. 38, no. 1, pp. 1–7.
- Gilloteaux, J., Jamison, J.M., Neal, D., Arnold, D., Taper, H.S. & Summers, J.L. 2013, "Human prostate DU145 carcinoma cells implanted in nude mice remove the peritoneal mesothelium to invade and grow as carcinomas", *Anatomical record*, vol. 296, no. 1, pp. 40–55.

- Goldstein, J., Newbury, D., Echlin, P., Joy, D., Fiori, C. & Lifshin, E. 1981, *Scanning electron microscopy and X-ray microanalysis*, Plenum Press, New York.
- Gololobov, G., Noda, Y., Sherman, S., Rubinstein, I., Baranowska-Kortylewicz, J. & Paul, S. 1998, "Stabilization of vasoactive intestinal peptide by lipids", *Journal of pharmacology and experimental therapeutics*, vol. 285, no. 2, pp. 753–758.
- González-Mariscal, L., Nava, P. & Hernández, S. 2005, "Critical role of tight junctions in drug delivery across epithelial and endothelial cell layers", *Journal of membrane biology*, vol. 207, no. 2, pp. 55–68.
- Goodlad, R.A. 2017, "Quantification of epithelial cell proliferation, cell dynamics, and cell kinetics in vivo", *Wiley interdisciplinary reviews, developmental biology*, vol. 6, no. 4, pp. 1–18.
- Goodlad, R.A. & Wright, N.A. 1990, "Changes in intestinal cell proliferation, absorptive capacity and structure in young, adult and old rats", *Journal of anatomy*, vol. 173, pp. 109–118.
- Gorea, G., Demy, M., Tran Van Nhieu, J., Tigori, J., Aubé, C., Cherqui, D., Oberti, F., Caroli-Bosc, F. & Calès, P. 2010, "Radiation-induced cholangitis with hepatocellular carcinoma", *Gastroenterologie clinique et biologique*, vol. 34, no. 1, pp. 35–39.
- Gorgani-Firouzjaee, T., Farshid, A.A. & Naem, S. 2015, "First ultrastructural observations on gastritis caused by Physaloptera clausa (Spirurida: Physalopteridae) in hedgehogs (Erinaceus europeaus)", *Parasitology research*, vol. 114, no. 10, pp. 3693-3698.
- Griffiths, N.M. 1999, "Long-term effects of X-irradiation on gastrointestinal function and regulatory peptides in monkeys", *International journal of radiation biology*, vol. 75, no. 2, pp.183-191.
- Gullberg, E., Leonard, M., Karlsson, J., Hopkins, A.M., Brayden, D., Baird, A.W. & Artursson, P. 2000, "Expression of specific markers and particle transport in a new human intestinal M-cell model", *Biochemical and biophysical research communications*, vol. 279, no. 3, pp. 808–813.
- Gundersen, H.J.G. 1977, "Notes on the estimation of the numerical density of arbitrary profiles: the edge effect", *Journal of microscopy*, vol. 111, no. 2, pp. 219–223.
- Gunter-Smith, P.J. 1989, "Gamma radiation affects active electrolyte transport by rabbit ileum. II. Correlation of alanine and theophylline response with morphology", *Radiation research*, vol. 117, no. 3, pp. 419–432.
- Hagemann, R.F., Sigdestad, C.P. & Lesher, S. 1970, "A quantitative description of the intestinal epithelium of the mouse", *Developmental dynamics*, vol. 129, no. 1, pp. 41–51.
- Hallab, N. J., & Jacobs, J. J. 2009, "Biologic effects of implant debris", *Bulletin of the NYU hospital for joint diseases*, vol. 67, no. 2, pp. 182-188.
- Hamlet, R., Carr, K.E., Nias, A.H.W. & Watt, C. 1981, "Surface damage in the small intestine of the mouse after X or neutron irradiation", *Scanning electron microscopy*, no. 4, pp. 73–78.

- Hamlet, R., Carr, K.E., Toner, P.G. & Nias, A.H.W. 1976, "Scanning electron microscopy of mouse intestinal mucosa after cobalt 60 and D-T neutron irradiation", *British journal of radiology*, vol. 49, no. 583, pp. 624–629.
- Hanker, J.S., Silverman, M.S. & Romanovicz, D.K.1977, "Catalase in salivary gland striated and excretory duct cells. II. φ Body: an ellipsoidal peroxisomal organelle with crystalloid axial projections", The Histochemical Journal, vol. 9, no. 6, pp.729-744.
- Hanson, W.R., Fry, R.J.M., Sallese, A.R., Frischer, H., Ahmad, T. & Ainsworth E.J. 1987, "Comparison of intestine and bone marrow radiosensitivity of the BALB/c and the C57BL/6 mouse strains and their B6CF1 offspring", *Radiation research*, vol. 110, pp. 340–352
- Harary, I. & Masamitsu, M. 1978, "Lead inhibition of the preparation for fusion of the L6line of skeletal muscle cells", *Developmental biology*, vol. 65, no. 1, pp. 28–37.
- Harrison, J.D. 1998, "Gut transfer and doses from environmental plutonium and americium", *Journal of radiological protection*, vol. 18, no. 2, pp. 73–76.
- Harrison, J.D. & Fritsch, P. 1992, "The effect of age on the absorption and intestinal retention of ingested radionuclides", *Radiation protection dosimetry*, vol. 41, no. 2–4, pp. 71–76.
- Harrison, J.D. & Stather, J.W. 1996, "The assessment of doses and effects from intakes of radioactive particles", *Journal of anatomy*, vol. 189, no. 3, pp. 521–530.
- Hauer-Jensen, M. 1990, "Late radiation injury of the small intestine clinical, pathophysiologic and radiobiologic aspects: a review", *Acta oncologica*, vol. 29, no. 4, pp. 401–415.
- Hauer-Jensen, M., Denham, J.W. & Andreyev, H.J.N. 2014, "Radiation enteropathy pathogenesis, treatment and prevention", *Nature reviews gastroenterology and hepatology*, vol. 11, no. 8, pp. 470–479.
- Hauer-Jensen, M.H., Sauer, T., Devik, F. & Nygaard, K. 1983a, "Late changes following single dose roentgen irradiation of rat small intestine", *Acta radiologica, oncology*, vol. 22, no. 4, pp. 299–303.
- Hauer Jensen, M., Sauer, T., Devik, F. & Nygaard, K. 1983b, "Effects of dose fractionation on late roentgen radiation damage of rat small intestine", *Acta radiologica, oncology*, vol. 22, no. 5, pp. 381–384.
- Havell, E.A., Beretich Jr., G.R. & Carter, P.B. 1999, "The mucosal phase of Listeria infection", *Immunobiology*, vol. 201, no. 2, pp. 164–177.
- Hayes, T.L. 1974, "Special Invited review, trends and prospects in scanning electron microscopy", *Journal of microscopy*, vol. 100, no. 2, pp. 133–142.
- Hayes, T.L., Pease, R.F. & McDonald, L.W. 1966, "Applications of the scanning electron microscope to biologic investigations", *Laboratory investigation*, vol. 15, no. 8, pp. 1320–1326.
- Hazzard, R.A., Hodges, G.M., Scott, J.D., McGuinness, C.B. & Carr, K.E. 1996, "Early intestinal microparticle uptake in the rat", *Journal of anatomy*, vol. 189, no. 2, pp. 265–271.

- Heine, H. & Schaeg, G. 1977, "Herkunft und funktion des globulären leukozyten", *Cells tissues organs,* vol. 98, no. 3, pp. 275–280.
- Heinzmann, U., Marquart, K.H. & Kriegel, H. 1978, "Electron microscopic investigation of X-irradiation effects on developing ependyma", *Scanning electron microscopy*, vol. 2, pp. 871–878.
- Hendry, J.H. & Potten, C.S. 1974, "Cryptogenic cells and proliferative cells in intestinal epithelium", *International journal of radiation biology*, vol. 25, pp. 583–588.
- Hidalgo, I.J. 1996, "Cultured intestinal epithelial cell models", *Pharmaceutical biotechnology*, vol. 8, pp. 35-50.
- Hidalgo, I.J., Raub, T.J. & Borchardt, R.T. 1989, "Characterization of the human colon carcinoma cell line (Caco-2) as a model system for intestinal epithelial permeability", *Gastroenterology*, vol. 96, no. 2, pp. 736-749.
- Hillyer, J.F. & Albrecht, R.M. 2001, "Gastrointestinal persorption and tissue distribution of differently sized colloidal gold nanoparticles", *Journal of pharmaceutical sciences*, vol. 90, no. 12, pp. 1927–1936.
- Höckerfelt, U., Franzén, L., Kjörell, U. & Forsgren, S. 2000, "Parallel increase in substance P and VIP in rat duodenum in response to irradiation", *Peptides*, vol. 21, no. 2, pp. 271–281.
- Höckerfelt, U., Henriksson, R., Franzén, L., Norrgård, O. & Forsgren, S. 1999, "Irradiation induces marked immunohistochemical expression of vasoactive intestinal peptide in colonic mucosa of man", *Digestive diseases and sciences*, vol. 44, no. 2, pp. 393-401.
- Hodges, G.M. 1982, "Tumour formation: the concept of tissue (stroma-epithelium) regulatory dysfunction" In *Functional integration of cells in animal tissues*, Eds. Pitts, J.D, & Finbow, ME, Cambridge University Press, Cambridge, UK, pp. 333–356.
- Hodges, G.M., Carr, E.A., Hazzard, R.A., O'Reilly, C. & Carr, K.E. 1995a, "A commentary on morphological and quantitative aspects of microparticle translocation across the gastrointestinal mucosa", *Journal of drug targeting*, vol. 3, no. 1, pp. 57–60.
- Hodges, G.M., Carr, E.A., Hazzard, R.A. & Carr, K.E. 1995b, "Uptake and translocation of microparticles in small intestine morphology and quantification of particle distribution", *Digestive diseases and sciences*, vol. 40, no. 5, pp. 967–975.
- Hodges, G.M., Carr, K.E., Hume, S.P., Marigold, J.C., Southgate, J. & Marshall, J.F. 1985, "Changes in surface structure and concanavalin A-binding capacity of urothelium in the mouse bladder after whole-body neutron irradiation", *Scanning electron microscopy*, vol. 4, pp. 1603–1614.
- Hodges, G.M, Smolira, M.A. & Trejdosiewicz, L.K. 1982, "Urothelium-specific antibody and lectin surface mapping of bladder urothelium", *The histochemical journal*, vol. 14, no. 5, pp. 755–766.

- Hopewell, J.W. 1980, "The importance of vascular damage in the development of late radiation effects in normal tissues", in *Radiation biology in cancer research*, Eds. Meyn, R.E. & Withers, H.R., Raven Press, New York, pp. 449–459.
- Hopper, A.F., Wannemacher, R.W. & Mcgovern, P.A. 1968, "Cell population changes in the intestinal epithelium of the rat following starvation and protein-depletion", *Proceedings of the society for experimental biology and medicine*, vol. 128, no. 3, pp. 695–698.
- Hornsey, S. 1970, "The relative biological effectiveness of fast neutrons for intestinal damage", *Radiology*, vol. 97, pp. 649–652.
- Hornsey, S., & Field, S. B. 1974, "The RBE of cyclotron neutrons for effects on normal tissues", *European journal of cancer*, vol.10, pp. 231–234.
- Hughes, P., Marshall, D., Reid, Y., Parkes, H. & Gelber, C. 2007, "The costs of using unauthenticated, over-passaged cell lines: how much more data do we need?", *BioTechniques*, vol. 43, no. 5, pp. 575–586.
- Hume, S.P., Robinson, J.E. & Hand, J.W. 1979a, "The influence of blood flow on temperature distribution in the exteriorized mouse intestine during treatment by hyperthermia" *The British journal of radiology*, vol. 52, no. 615, pp. 219–222.
- Hume, S.P., Marigold, J.C. & Field, S.B. 1979b, "The effect of local hyperthermia on the small intestine of the mouse" *The British journal of radiology*, vol. 52, no. 620, pp. 657–662.
- Humphreys, W.J. & Henk, W.G. 1979, "Ultrastructure of cell organelles by scanning electron microscopy of thick sections surface-etched by an oxygen plasma, *Journal of microscopy*, vol. 116, no. 2, pp. 255–264.
- Hunt, G.J. 1998, "Transfer across the human gut of environmental plutonium, americium, cobalt, caesium and technetium: studies with cockles (Cerastoderma edule) from the Irish Sea", *Journal of radiological protection*, vol. 18, no. 2, pp. 101-109.
- Hussain, N. 2016, "Regulatory aspects in the pharmaceutical development of nanoparticle drug delivery systems designed to cross the intestinal epithelium and M-cells", *International journal of pharmaceutics*, vol. 514, no. 1, pp. 15–23.
- Hussain, N., Jaitley, V. & Florence, A.T. 2001, "Recent advances in the understanding of uptake of microparticulates across the gastrointestinal lymphatics", *Advanced drug delivery reviews*, vol. 50, no. 1– 2, pp. 107–142.
- Iannuccelli, V., Montanari, M., Bertelli, D., Pellati, F. & Coppi, G. 2011, "Microparticulate polyelectrolyte complexes for gentamicin transport across intestinal epithelia", *Drug delivery*, vol. 18, no. 1, pp. 26–37.
- Ijiri, K. & Potten, C. S. 1990, "The circadian-rhythm for the number and sensitivity of radiation-induced apoptosis in the crypts of mouse small-intestine", *International journal of radiation biology*, vol. 58, pp. 165–175.

- Inagaki, E., Natori, Y., Ohgishi, Y., Hayashi, H. & Suzuki, Y. 2005, "Segmental difference of mucosal damage along the length of a mouse small intestine in an Ussing chamber", *Journal of nutritional science and vitaminology*, vol. 51, no. 6, pp. 406–412.
- Indran, M., Boyle, F.C. & Carr, K.E. 1985, "Simulation with pharmacological agents of radiation damage to small intestinal villi", *Scanning electron microscopy*, vol. 3, pp. 1165–1175.
- Indran, M., Carr, K.E. & Boyle, F.C. 1988, "Radiation-induced changes in mouse duodenal papilla", *British journal of radiology*, vol. 61, no. 731, pp. 1039-1042.
- Indran, M., Carr, K.E., Gilmore, R.S. & Boyle, F.C. 1991, "Mucosal changes in mouse duodenum after gammairradiation or reserpine treatment", *Journal of submicroscopic cytology and pathology*, vol. 23, no. 2, pp. 267–278.
- Ingels, F.M. & Augustijns, P.F. 2003, "Biological, pharmaceutical, and analytical considerations with respect to the transport media used in the absorption screening system, Caco-2", *Journal of pharmaceutical sciences,* vol. 92, no. 8, pp. 1545–1558.
- Inoué, T. & Koike, H. 1989, "High-resolution low-temperature scanning electron microscopy for observing intracellular structures of quick frozen biological specimens", *Journal of microscopy*, vol. 156, no. 2, pp. 137-147.
- International Commission on Radiological Protection (ICRP) 1991, "Recommendations of the international commission on radiological protection.", 1990, Publication 60, *Annals of the ICRP*, vol. 21, no. 1–3, pp. 1–201.
- Ishihara, H., Tanaka, I., Yakumaru, H., Tanaka, M., Satoh, A., Ishiwata, A., Yokochi, K., Kurematsu, A., Ueda, J., Shibata, T., Hachiya, M. & Akashi, M. 2011, "Acceleration of regeneration of mucosa in small intestine damaged by ionizing radiation using anabolic steroids", *Radiation research*, vol. 175, no. 3, pp. 367–374.
- Jacob, S.T. & Bhargava, P.M. 1962, "A new method for the preparation of liver cell suspensions", *Experimental cell research*, vol. 27, no. 3, pp. 453–467.
- Jahnson, S., Christofferson, R.H. & Gerdin, B. 1998, "Reduced mucosal perianastomotic capillary density in rat small intestine with chronic radiation damage", *Radiation research*, vol. 150, no. 5, pp. 542–548.
- Jani, P., Halbert, G.W., Langridge, J. & Florence, A.T. 1989, "The uptake and translocation of latex nanospheres and microspheres after oral administration to rats", *Journal of pharmacy and pharmacology*, vol. 41, no. 12, pp. 809–812.
- Jaques, W.E., Coalson, J. & Zervins, A. 1965, "Application of the scanning electron microscope to human tissues. A preliminary study", *Experimental and molecular pathology*, vol. 4, no. 6, pp. 576–580.
- Jenkins, P.G., Howard, K.A., Blackball, N.W., Thomas, N.W., Davis, S.S. & O'hagan, D.T. 1994, "Microparticulate absorption from the rat intestine", *Journal of controlled release*, vol. 29, no. 3, pp.339–350.

- Jepson, M.A., Simmons, N.L., Savidge, T.C., James, P.S. & Hirst, B.H. 1993, "Selective binding and transcytosis of latex microspheres by rabbit intestinal M cells", *Cell and tissue research*, vol. 271, no. 3, pp. 399–405.
- Johnson, C.J., McKenzie, D., Pedersen, J.A., & Aiken, J.M. 2011, "Meat and bone meal and mineral feed additives may increase the risk of oral prion disease transmission", *Journal of toxicology and environmental health, Part A*, vol. 74, no. 2–4, pp. 161–166.
- Johnsson, F., Joelsson, B., Gudmundsson, K. & Greiff, L. 1987, "Symptoms and endoscopic findings in the diagnosis of gastroesophageal reflux disease", *Scandinavian journal of gastroenterology*, vol. 22, no. 6, pp. 714–718.
- Johnston, B.T., Nunn, S., Sloan, J.M., Collins, J.S.A., Mcfarland, R.J., Parkin, S., Carr, K.E. & Collins, B.J. 1996, "The application of microridge analysis in the diagnosis of gastro-oesophageal reflux disease", *Scandinavian journal of gastroenterology*, vol. 31, no. 2, pp. 97–102.
- Jongebloed, W.L., Stokroos, I., Van Der Want, J.J.L. & Kalicharan, D. 1999, "Non-coating fixation techniques or redundancy of conductive coating, low kV FE-SEM operation and combined SEM/TEM of biological tissues", *Journal of microscopy*, vol. 193, no. 2, pp. 158–170.
- Joseph, R.A., Shepard, B.D., Kannarkat, G.T., Rutledge, T.M., Tuma, D.J. & Tuma, P.L. 2008, "Microtubule acetylation and stability may explain alcohol-induced alterations in hepatic protein trafficking", *Hepatology*, vol. 47, no. 5, pp. 1745–1753.
- Jünemann, D. & Dressman, J. 2012, "Analytical methods for dissolution testing of nanosized drugs", *Journal of pharmacy and pharmacology*, vol. 64, no. 7, pp. 931–943.
- Kallfass, E., Krämling, H. & Schultz-Hector, S. 1996, "Early inflammatory reaction of the rabbit coeliac artery wall after combined intraoperative (IORT) and external (ERT) irradiation", Radiotherapy and oncology, vol. 39, no. 2, pp. 167–178.
- Kamel, H.M.H., Carr, K.E., Hume, S.P. & Marigold, J.C.L. 1985, "Structural changes in mouse small intestinal villi following lower body hyperthermia", *Scanning electron microsccopy*, vol. 2, pp. 849–858.
- Kamel, H.M.H., Hume, S.P., Carr, K.E., Marigold, J.C.L. & Michalowski, A. 1988, "Development of villous damage in mouse small intestine after local hyperthermia or irradiation", *Journal of submicroscopic cytology and pathology*, vol. 20, no. 1, pp. 185–193.
- Kanimozhi, G., Rajendra Prasad, N., Ramachandran, S. & Pugalendi, K.V. 2012, "Umbelliferone protects wholebody irradiated Swiss albino mice: study on animal survival, tissue antioxidant status and DNA damage", *Biomedicine and preventive nutrition*, vol. 2, no. 3, pp. 186–192.
- Kanter, M. & Akpolat, M. 2008, "Vitamin C protects against ionizing radiation damage to goblet cells of the ileum in rats", *Acta histochemica*, vol. 110, no. 6, pp. 481–490.

- Kanzato, H., Manabe, M. & Shimizu, M. 2001, "An in vitro approach to the evaluation of the cross talk between intestinal epithelium and macrophages", *Bioscience, biotechnology and biochemistry*, vol. 65, no. 2, pp. 449-451.
- Kazazi-Hyseni, F., Landin, M., Lathuile, A., Veldhuis, G.J., Rahimian, S., Hennink, W.E., Kok, R.J. & Van Nostrum, C.F. 2014, "Computer modeling assisted design of monodisperse PLGA microspheres with controlled porosity affords zero order release of an encapsulated macromolecule for 3 months", *Pharmaceutical research*, vol. 31, no. 10, pp. 2844–2856.
- Kent, T.H., Cardelli, R.M. & Stamler, F.W. 1969, "Small intestinal ulcers and intestinal flora in rats given indomethacin", *American journal of pathology*, vol. 54, no. 2, pp. 237–249.
- Kernéis, S., Bogdanova, A., Kraehenbuhl, J.P. & Pringault, E. 1997, "Conversion by Peyer's patch lymphocytes of human enterocytes into M cells that transport bacteria", *Science*, vol. 277, no. 5328, pp. 949-952.
- Kiefer, J., Schenk-Meuser, K. & Kost, M., 1996. "Radiation biology", in *Biological and medical research in space*, Eds. Moore, D., Bie, P. & Oser, H., Springer, Berlin, Heidelberg, pp. 300–367.
- Kobori, T., Watanabe, J. & Nakao, H. 2012, "Gold nanoparticles as localization markers for direct and live imaging of particle absorption through a Caco-2 cell monolayer using dark-field microscopy", *Analytical Sciences*, vol. 28, no. 1, pp.61–61.
- Köst, S., Keinert, K. & Glaser, F. 1998, "D-xylose test of resorption as a method to determine radiation side effects in small intestine", *Strahlentherapie und onkologie*, vol. 174, no. 9, pp. 462–467.
- Koudela, B., Modrý, D., Svobodová, M., Votýpka, J., Vávra, J. & Hudcovic, T. 1999, "The severe combined immunodeficient mouse as a definitive host for Sarcocystis muris", *Parasitology research*, vol. 85, no. 8-9, pp. 737-742.
- Kreuter, J. 1991, "Peroral administration of nanoparticles", Advanced drug delivery reviews, vol. 7, no. 1, pp. 71-86.
- Kreuter, J., Müller, U. & Munz, K. 1989, "Quantitative and microautoradiographic study on mouse intestinal distribution of polycyanoacrylate nanoparticles", *International journal of pharmaceutics*, vol. 55, no. 1, pp. 39-45.
- Kunwar, A., Priyadarsini, K.I., & Jain, V.K. 2011,"Organoselenium compounds: a new generation of radioprotectors", *BARC Newsletter*, vol. 319, pp.1–7.
- Kurul, S., Dinçer, M., Kizir, A., Uzunismail, A. & Darendeliler, E. 1997, "Plastic surgery in irradiated areas: analysis of 200 consecutive cases", *European journal of surgical oncology*, vol. 23, no. 1, pp. 48–53.
- Kwan, D. K., & Norman, A. 1977, "Radiosensitivity of human lymphocytes and thymocytes", *Radiation research*, vol. 69, pp. 143—151.

- Laferla, G.A., Shashaa, S., Crean, G.C., Gilmore, R.S. & Carr, K.E. 1988, "Topographical variation in the mucosal surface of oesophageal biopsies", *Journal of submicroscopic cytology and pathology*, vol. 20, no. 4, pp. 731-739.
- Lai, Y., Chiang, P.C., Blom, J.D., Li, N., Shevlin, K., Brayman, T.G., Hu, Y., Selbo, J.G. & Hu, L.G. 2008, "Comparison of in vitro nanoparticles uptake in various cell lines and in vivo pulmonary cellular transport in intratracheally dosed rat model", *Nanoscale research letters*, vol. 3, no. 9, pp. 321–329.
- Lai, Y.H. & D'Souza, M. 2008, "Microparticle transport in the human intestinal M cell model", *Journal of drug targeting*, vol. 16, no. 1, pp. 36-42.
- Lametschwandtner, A., Lametschwandtner, U. & Weiger, T. 1984, "Scanning electron microscopy of vascular corrosion casts-technique and applications", *Scanning electron microscopy*, vol 2, pp. 663-695.
- Langenbach, R., Morham, S.G., Tiano, H.F., Loftin, C.D., Ghanayem, B.I., Chulada, P.C., Mahler, J.F., Lee, C.A., Goulding, E.H., Kluckman, K.D., Kim, H.S. & Smithies, O. 1995, "Prostaglandin synthase 1 gene disruption in mice reduces arachidonic acid-induced inflammation and indomethacin-induced gastric ulceration", *Cell*, vol. 83, no. 3, pp. 483–492.
- Larkin, J.C. 1949, "Effect of atropine on acute irradiation sickness in mice", *American journal of roentgenology*, vol. 62, pp. 547–549.
- Laschi, R. & Govoni, E. 1978, "Staining methods for semithin sections" In *Electron microscopy in human medicine*, vol. 1, pp. 187–198.
- Le Ray, A.M., Vert, M., Gautier, J.C. & Benoît, J.P. 1994, "Fate of [14C]poly(DL-lactide-co-glycolide) nanoparticles after intravenous and oral administration to mice", *International journal of pharmaceutics*, vol. 106, no. 3, pp. 201–211.
- Le Visage, C., Couvreur, P., Mysiakine, E., Breton, P., Bru, N. & Fattal, E. 2001, "In vitro and in vivo evaluation of poly(methylidene malonate 2.1.2) microparticles behavior for oral administration", *Journal of drug targeting*, vol. 9, no. 2, pp. 141–153.
- Leazer, T.M., Liu, Y. & Klaassen, C.D. 2002, "Cadmium absorption and its relationship to divalent metal transporter-1 in the pregnant rat", *Toxicology and applied pharmacology*, vol. 185, no. 1, pp. 18-24.
- Lebrun, F., Francois, A., Vergnet, M., Lebaron-Jacobs, L., Gourmelon, P. & Griffiths, N.M. 1998, "Ionizing radiation stimulates muscarinic regulation of rat intestinal mucosal function", American journal of physiologygastrointestinal and liver physiology, vol. 275, no. 6, pp. G1333-G1340.
- Lee, J. & Schmid-Schönbein, G.W. 1995, "Biomechanics of skeletal muscle capillaries: hemodynamic resistance, endothelial distensibility, and pseudopod formation", *Annals of biomedical engineering*, vol. 23, no. 3, pp. 226–246.

- LeFevre, M.E., Boccio, A.M. & Joel, D.D. 1989, "Intestinal uptake of fluorescent microspheres in young and aged mice", *Proceedings of the society for experimental biology and medicine,* vol. 190, no. 1, pp. 23–27.
- LeFevre, M.E., Hancock, D.C. & Joel, D.D. 1980, "Intestinal barrier to large particulates in mice", *Journal of toxicology and environmental health*, vol. 6, no. 4, pp. 691–704.
- LeFevre, M.E., Joel, D.D. & Schidlovsky, G. 1985, "Retention of ingested latex particles in Peyer's patches of germfree and conventional mice", *Proceedings of the society for experimental biology and medicine*, vol. 179, no. 4, pp. 522–528.
- Legraverend, C. & Jay, P. 2013, "Random chromosome segregation in mouse intestinal epithelial stem cells", *Chromosome research*, vol. 21, no. 3, pp. 213–224.
- Lehy, T., Dessirier, V., Attoub, S., Bado, A., Griffiths, N.M. & Linard, C. 1998, "Exposure to ionizing radiation modifies circulating gastrin levels and gastrointestinal endocrine cell densities in the rat", *International journal of radiation biology*, vol. 73, no. 3, pp. 331–340.
- Leith, J., McDonald, M., & Howard, J. 1982, "Residual skin damage in rats 1 year after exposure to X rays or accelerated heavy ions", *Radiation research*, vol.89, no.1, pp. 209–213.
- Lev, R. 1983, "Contributions of electron microscopy to our understanding of childhood diarrhea", *Journal of pediatric gastroenterology and nutrition*, vol. 2, no. 1, pp. 6–9.
- Lever, J.D. & Duncumb, P. 1961, "The detection of intracellular iron in rat duodenal epithelium", in *Electron microscopy in anatomy*, Eds., Boyd, J.D., Johnson, F.R. & Lever, J.D., Edward Arnold, London, pp. 278–286.
- Levi, S. & Hodgson, H.J. 1990, "The medical management of radiation enteritis", in *Radiation enteritis*, Eds. Galland, R.B. & Spencer, J., Edward Arnold, London, 120–135.
- Levi, S., Goodlad, R.A., Lee, C.Y., Walport, M.J., Wright, N.A. & Hodgson, H.J.F. 1992, "Effects of nonsteroidal antiinflammatory drugs and misoprostol on gastroduodenal epithelial proliferation in arthritis", *Gastroenterology*, vol. 102, no. 5, pp. 1605–1611.
- Lewis, S.J. & Freedman, A.R. 1998, "Review article: the use of biotherapeutic agents in the prevention and treatment of gastrointestinal disease", *Alimentary pharmacology and therapeutics*, vol. 12, no. 9, pp. 807–822.
- Li, N., Wang, D., Sui, Z., Qi, X., Ji, L., Wang, X. & Yang, L. 2013, "Development of an improved three-dimensional in vitro intestinal mucosa model for drug absorption evaluation", *Tissue engineering part C: methods*, vol. 19, no. 9, pp. 708–719.
- Li, W. 2015, "Modelling methods for in vitro biomechanical properties of the skin: a review", *Biomedical engineering letters*, vol. 5, no. 4, pp. 241-250.

- Li, Y. & Zhu, C. 2016, "Enhanced hepatic-targeted delivery via oral administration using nanoliposomes functionalized with a novel DSPE-PEG-cholic acid conjugate", *RSC (Royal society of chemistry) advances,* vol. 6, no. 33, pp. 28110–28120.
- Liang, F., Kachar, B., Ding, M., Zhai, Z., Wu, X.R. & Sun, T.T. 1999, "Urothelial hinge as a highly specialized membrane: detergent-insolubility, urohingin association, and in vitro formation", *Differentiation*, vol. 65, no. 1, pp.59–69.
- Lieb, R.J., McDonald, T.F. & McKenney, J.R. 1977, "Fine-structural effects of 1200-R abdominal X irradiation on rat intestinal epithelium", *Radiation research*, vol. 70, no. 3, pp. 575–584.
- Linard, C., Griffiths, N.M., Esposito, V., Aigueperse, J. & Gourmelon, P. 1997, "Changes in gut neurotensin and modified colonic motility following whole-body irradiation in rat", *International journal of radiation biology*, vol. 71, no. 5, pp. 581–588.
- Llopis, S., Hernández-Haro, C., Monteoliva, L., Querol, A., Molina, M. & Fernández-Espinar, M.T. 2014, "Pathogenic potential of Saccharomyces strains isolated from dietary supplements", *PLoS ONE*, vol. 9, no. 5.
- Löbenberg, R., Araujo, L. & Kreuter, J. 1997, "Body distribution of azidothymidine bound to nanoparticles after oral administration", *European journal of pharmaceutics and biopharmaceutics*, vol. 44, no. 2, pp. 127–132.
- Loehry, C. A., & Creamer, B., 1966, "Post-mortem study of small-intestinal mucosa", *British medical journal*, vol. 1, no. 1 (5491), pp. 820–822 and 827–829.
- Lopes, M.A., Abrahim, B.A., Cabral, L.M., Rodrigues, C.R., Seiça, R.M.F., de Baptista Veiga, F.J. & Ribeiro, A.J. 2014, "Intestinal absorption of insulin nanoparticles: contribution of M cells", *Nanomedicine: nanotechnology, biology, and medicine*, vol. 10, no. 6, pp. 1139–1151.
- Luft, J.H. 1966, "Fine structures of capillary and endocapillary layer as revealed by ruthenium red", *Federation proceedings*, vol. 25, no. 6, pp. 1773–1783.
- Lumsden, J.S., Ferguson, H.W., Ostland, V.E. & Byrne, P.J. 1994, "The mucous coat on gill lamellae of rainbow trout (Oncorhynchus mykiss)", *Cell and tissue research*, vol. 275, no. 1, pp. 187–193.
- Lwin, S., Inoshima, Y., Atoji, Y., Ueno, H. & Ishiguro, N. 2009, "Immune cell types involved in early uptake and transport of recombinant mouse prion protein in Peyer's patches of calves", *Cell and tissue research*, vol. 338, no. 3, pp. 343–354.
- Lyu, S.Y. & Park, W.B. 2008, "Transport of mistletoe lectin by M cells in human intestinal follicle-associated epithelium (FAE) in vitro", *Archives of pharmacal research*, vol. 31, no. 12, pp. 1613–1621.
- Ma, Q., Hoper, M., Halliday, I. & Rowlands, B.J. 1996, "Diet and experimental colorectal cancer", *Nutrition research,* vol. 16, no. 3, pp. 413–426.

- Ma, T. Y., Iwamoto, G. K., Hoa, N. T., Akotia, V., Pedram, A., Boivin, M. A., & Said, H. M. 2004, "TNF-α-induced increase in intestinal epithelial tight junction permeability requires NF-κB activation", *American journal of physiology - gastrointestinal and liver physiology*, vol. 286, no. 3, pp. G367-G376.
- Ma, T.Y., Nguyen, D., Bui, V., Nguyen, H. & Hoa, N. 1999, "Ethanol modulation of intestinal epithelial tight junction barrier", *American journal of physiology - gastrointestinal and liver physiology*, vol. 276, no. 4, pp. G965– G974.
- MacNaughton, W.K. 2000, "Review article: new insights into the pathogenesis of radiation-induced intestinal dysfunction", *Alimentary pharmacology and therapeutics*, vol. 14, no. 5, pp. 523–528.
- MacNaughton, W.K., Leach, K.E., Prud'homme-Lalonde, L. & Harding, R.K. 1997, "Exposure to ionizing radiation increases responsiveness to neural secretory stimuli in the ferret jejunum in vitro", *International journal of radiation biology*, vol. 72, no. 2, pp. 219–226.
- Madara, J.L. 1998, "Regulation of the movement of solutes across tight junctions", *Annual review of physiology*, vol. 60, pp. 143–159.
- Malpighi, M., 1687, "Opera Omnia", Leiden (lugduni Batavarum).
- Manabe, M., Takenaka, R., Nakasa, T. & Okinaka, O. 2003, "Induction of anti-inflammatory responses by dietary Momordica charantia L. (bitter gourd)", *Bioscience, biotechnology and biochemistry*, vol. 67, no. 12, pp. 2512– 2517.
- Mantle, M., Thakore, E., Atkins, E., Mathison, R. & Davison, J.S. 1989, "Effects of streptozotocin-diabetes on rat intestinal mucin and goblet cells", *Gastroenterology*, vol. 97, no. 1, pp. 68–75.
- Mantovani, B., Rabinovitch, M. & Nussenzweig, V. 1972, "Phagocytosis of immune complexes by macrophages: different roles of the macrophage receptor sites for complement (C3) and for immunoglobulin (IgG)", *Journal of experimental medicine*, vol. 135, no. 4, pp. 780–792.
- Marsh, M.N. & Swift, J.A. 1969, "A study of the small intestinal mucosa using the scanning electron microscope", *Gut*, vol. 10, no. 11, pp. 940–949.
- Marteau, P. & Shanahan, F. 2003, "Basic aspects and pharmacology of probiotics: an overview of pharmacokinetics, mechanisms of action and side-effects", *Bailliere's best practice and research in clinical gastroenterology*, vol. 17, no. 5, pp. 725-740.
- Martin, D.E., Gould, K.G. & Warner, H. 1975, "Comparative morphology of primate spermatozoa using scanning electron microscopy. I. Families Hominidae, Pongidae, Cercopithecidae and Cebidae", *Journal of human evolution*, vol. 4, pp. 287–288.
- Mastaglia, F.L., McDonald, W.I., Watson, J.V. & Yogendran, K. 1976, "Effects of X-radiation on the spinal cord: an experimental study of the morphological changes in central nerve fibres", *Brain*, vol. 99, no. 1, pp. 101–122.

- Matravers, J. & Tyldesley, W.R. 1978, "Scanning electron microscopy of oral epithelial cells, Part I. Normal and malignant tissue", *British journal of oral surgery*, vol. 15, no. 3, pp.193–202.
- Maugel, T.K., Bonar, D.B., Creegan, W.J. & Small, E.B. 1980, "Specimen preparation techniques for aquatic organisms", *Scanning electron microscopy*, vol.2, pp. 57–78.
- Maupus, J.L., Champemont, P. & Delforge, M. 1983, "Treatment of irritable bowel syndrome. Double-blind trial of Saccharomyces boulardii", *Medicine chirurgic digestive*, vol. 12. No 1, pp.77–79.
- Mayhew, T.M. & Carson, F.L. 1989, "Mechanisms of adaptation in rat small intestine: regional differences in quantitative morphology during normal growth and experimental hypertrophy", *Journal of anatomy*, vol. 164, pp. 189–200.
- Mayhew, T.M., Dantzer, V., Sodring Elbrond, V. & Skadhauge, E. 1990, "A sampling scheme intended for tandem measurements of sodium transport and microvillous surface area in the coprodaeal epithelium of hens on high- and low-salt diets", *Journal of anatomy*, vol. 173, pp. 19–31.
- Mayr, N.A., Riggs C.E. Jr., Saag K.G., Wen B.C., Pennington, E.C. & Hussey D.H. 1997, "Mixed connective tissue disease and radiation toxicity", *Cancer*, vol. 79, pp. 612–618.
- McClean, P., Dodge, J.A., Nunn, S., Carr, K.E. & Sloan, J.M. 1996, "Surface features of small-intestinal mucosa in childhood diarrheal disorders", *Journal of pediatric gastroenterology and nutrition*, vol. 23, no. 5, pp. 538–546.
- McClean, S., Prosser, E., Meehan, E., O'Malley, D., Clarke, N., Ramtoola, Z. & Brayden, D. 1998, "Binding and uptake of biodegradable poly-DL-lactide micro- and nanoparticles in intestinal epithelia", *European journal of pharmaceutical sciences*, vol. 6, no. 2, pp. 153–163.
- McCullough, J.S., Hodges, G.M., Dickson, G.R., Yarwood, A. & Carr, K.E. 1995, "A morphological and microanalytical investigation into the uptake of particulate iron across the gastrointestinal tract of rats", *Journal of submicroscopic cytology and pathology*, vol. 27, no. 1, pp. 119–124.
- McCullough, J.S., Ratcliffe, B., Mandir, N., Carr, K.E. & Goodlad, R.A. 1998, "Dietary fibre and intestinal microflora: effects on intestinal morphometry and crypt branching", *Gut*, vol. 42, no. 6, pp. 799–806.
- McCullough, M.J., Clemons, K.V., McCusker, J.H. & Stevens, D.A. 1998, "Species identification and virulence attributes of Saccharomyces boulardii (nom. inval.)", *Journal of clinical microbiology*, vol. 36, no. 9, pp. 2613–2617.
- McElligott, T.F., Beck, I.T., Dinda, P.K. & Thompson, S. 1975, "Correlation of structural changes at different levels of the jejunal villus with positive net water transport in vivo and in vitro", *Canadian journal of physiology and pharmacology*, vol. 53, no. 3, pp. 439–450.
- McManus, J.P. & Isselbacher, K.J. 1970, "Effect of fasting versus feeding on the rat small intestine. Morphological, biochemical, and functional differences", *Gastroenterology*, vol. 59, no. 2, pp. 214–221.

- McMinn, L.H., Hodges, G.M. & Carr, K.E. 1996, "Gastrointestinal uptake and translocation of microparticles in the streptozotocin-diabetic rat", *Journal of anatomy*, vol. 189, no. 3, pp. 553–559.
- McPhail, C.D. & Duncan, H.J. 1981, "Studies on the foliar uptake of iron into potato leaves". In *EAPR conference papers*, pp. 216–217.
- Mead, J.R., Arrowood, M.J., Sidwell, R.W. & Healey, M.C. 1991, "Chronic Cryptosporidium parvum infections in congenitally immunodeficient SCID and nude mice", *Journal of infectious diseases*, vol. 163, no. 6, pp. 1297–1304.
- Messerschmidt, O. 1986, "Whole-body irradiation plus skin wound: animal experiments on combined injuries", *British journal of radiology supplement*, vol. 19, pp. 64–7.
- Michalowski, A., Uehara, S., Yin, W.B., Burgin, J. & Silvester, J.A. 1983, "Alternative types of duodenal ulcer induced in mice by partial X irradiation of the thorax", *Radiation research*, vol. 95, no. 1, pp. 78–86.
- Min, X.H., Yu, T., Qing, Q., Yuan, Y.H., Zhong, W., Chen, G.C., Zhao, L.N., Deng, N., Zhang, L.F. & Chen, Q.K. 2014, "Abnormal differentiation of intestinal epithelium and intestinal barrier dysfunction in diabetic mice associated with depressed Notch/NICD transduction in Notch/Hes1 signal pathway", *Cell biology international*, vol. 38, no. 10, pp. 1194-1204.
- Ming, S.C. & Goldman, H. 1998, "Pathology of the gastrointestinal tract", 2nd edition, Williams and Wilkins, Baltimore, MD.
- Miyazawa, K., Hondo, T., Kanaya, T., Tanaka, S., Takakura, I., Itani, W., Rose, M.T., Kitazawa, H., Yamaguchi, T. & Aso,
 H. 2010, "Characterization of newly established bovine intestinal epithelial cell line", *Histochemistry and cell biology*, vol. 133, no. 1, pp. 125–134.
- Mizukami-Murata, S., Iwahashi, H., Kimura, S., Nojima, K., Sakurai, Y., Saitou, T., Fujii, N., Murata, Y., Suga, S., Kitagawa, K., Tanaka, K., Endo, S. & Hoshi, M. 2010, "Genome-wide expression changes in Saccharomyces cerevisiae in response to high-LET ionizing radiation", *Applied biochemistry and biotechnology*, vol. 162, no. 3, pp. 855–870.
- Møller, P., Folkmann, J.K., Forchhammer, L., Bräuner, E.V., Danielsen, P.H., Risom, L. & Loft, S. 2008, "Air pollution, oxidative damage to DNA, and carcinogenesis", *Cancer letters*, vol. 266, no. 1, pp. 84–97.
- Montagne, L., Pluske, J.R. & Hampson, D.J. 2003, "A review of interactions between dietary fibre and the intestinal mucosa, and their consequences on digestive health in young non-ruminant animals", *Animal feed science and technology*, vol. 108, no. 1–4, pp.95–117.
- Moore, M.R., Meredith, P.A., Goldberg, A., Carr, K.E., Toner, P.G. & Lawrie, T.D. 1975, "Cardiac effects of lead in drinking water of rats", *Clinical science and molecular medicine*, vol. 49, no. 4, pp. 337-341.
- Morgan, J.L.L., Ritchie, L.E., Crucian, B.E., Theriot, C., Wu, H., Sams, C., Smith, S.M., Turner, N.D. & Zwart, S.R. 2014, "Increased dietary iron and radiation in rats promote oxidative stress, induce localized and systemic

immune system responses, and alter colon mucosal environment", *FASEB Journal*, vol. 28, no. 3, pp. 1486-1498.

- Morini, J., Babini, G., Barbieri, S., Baiocco, G., & Ottolenghi, A. 2017, "The interplay between radioresistant Caco-2 cells and the immune system increases epithelial layer permeability and alters signaling protein spectrum", *Frontiers in immunology*, vol. 8, no. 223, pp. 1–12.
- Morris, M.M. & Powell, S.N. 1997, "Irradiation in the setting of collagen vascular disease: acute and late complications", *Journal of clinical oncology*, vol. 15, no. 7, pp. 2728–2735.
- Motta, P. 1975, "A scanning electron microscopic study of the rat liver sinusoid endothelial and Kupffer cells", *Cell and tissue research*, vol. 164, no. 3, pp. 371–385.
- Mould, R.F. 1986, "Medical physics handbooks 7, radiotherapy treatment planning", Adam Hilger Ltd., Bristol, pp. 144–159.
- Moussavi, N., Gavino, V. & Receveur, O. 2008. "Could the quality of dietary fat, and not just its quantity, be related to risk of obesity?", *Obesity*, vol. 16, no. 1, pp.7–15.
- Moyes, S.M., Killick, E.M., Morris, J.F., Kadhim, M.A., Hill, M.A. & Carr, K.E. 2008, "Changes produced by external radiation in parameters influencing intestinal permeability and microparticle uptake in vitro", *International journal of radiation biology*, vol. 84, no. 6, pp. 467–486.
- Moyes, S.M., Morris, J.F. & Carr K.E. 2009, "The effect of lymphocytes on Caco-2 intestinal epithelial permeability", *Journal of anatomy*, vol. 214, no.5, p. 800.
- Moyes, S.M., Morris, J.F. & Carr, K.E. 2010a, "Culture conditions and treatments affect Caco-2 characteristics and particle uptake", *International journal of pharmaceutics*, vol. 387, no. 1-2, pp. 7-18.
- Moyes, S.M., Morris, J.F. & Carr, K.E. 2010b, "Macrophages increase microparticle uptake by enterocyte-like Caco-2 cell monolayers", *Journal of anatomy*, vol. 217, no. 6, pp. 740-754.
- Moyes, S.M., Morris, J.F. & Carr, K.E. 2011, "Roles of pre-treatment time and junctional proteins in Caco-2 cell microparticle uptake", *International journal of pharmaceutics*, vol. 407, no. 1–2, pp. 21–30.
- Moyes, S.M., Smyth, S.H., Shipman, A., Long, S., Morris, J.F. & Carr, K.E. 2007, "Parameters influencing intestinal epithelial permeability and microparticle uptake in vitro", *International journal of pharmaceutics*, vol. 337, no. 1–2, pp. 133–141.
- Mukherjee, T., Squillantea, E., Gillespieb, M. & Shao, J. 2004, "Transepithelial electrical resistance is not a reliable measurement of the Caco-2 monolayer integrity in transwell", *Drug delivery: journal of delivery and targeting of therapeutic agents,* vol. 11, no. 1, pp. 11–18.
- Murakami, T. 1971, "Application of the scanning electron microscope to the study of the fine distribution of the blood vessels", *Archivum histologicum japonicum*, vol. 32, pp. 445–54.

- Musch, M. W., Walsh-Reitz, M. M., & Chang, E. B. 2006, "Roles of ZO-1, occludin, and actin in oxidant-induced barrier disruption", *American journal of physiology gastrointestinal and liver physiology*, vol. 290, no. 2, pp. G222-G231.
- Muto, M. 1975, "A scanning electron microscopic study on endothelial cells and Kupffer cells in rat liver sinusoids", *Archivum histologicum japonicum*, vol. 37, no. 5, pp. 369–386.
- Mutwiri, G.K., Kosecka, U., Benjamin, M., Rosendal, S., Perdue, M. & Butler, D.G. 2001, "Mycobacterium avium subspecies paratuberculosis triggers intestinal pathophysiologic changes in beige/SCID mice", *Comparative medicine*, vol. 51, no. 6, pp. 538–544.
- Naftalovich, R., Naftalovich, D. & Greenway, F.L. 2016, "Polytetrafluoroethylene ingestion as a way to increase food volume and hence satiety without increasing calorie content", *Journal of diabetes science and technology*, vol. 10, no. 4, pp. 971–976.
- Nagata, T. 2004, "X-ray microanalysis of biological specimens by high voltage electron microscopy", *Progress in histochemistry and cytochemistry*, vol. 39, no. 4, pp. 185–319.
- Nakane, P.K. & Hartman, A.L. 1980, "Immunocytochemical localization of intracellular antigens with SEM", *The histochemical journal*, vol. 12, no. 4, pp. 435–447.
- NCRP, 1989, "Guidance on radiation received in space activities. Report 98", *National council on radiation protection and measurements*, Bethesda, MD.
- Nguyen, D.A. & Neville, M.C. 1998, "Tight junction regulation in the mammary gland", *Journal of mammary gland biology and neoplasia*, vol. 3, no. 3, pp. 233–246.
- Nias, A.H.W. & Paul, J. 1961, "DNA-content of giant cells produced by irradiation", *International journal of radiation biology and related studies in physics, chemistry and medicine*, vol. 3, no. 4, pp. 431–438.
- Nicklin, P.L., Irwin, W.J., Hassan, I.F. & Mackay, M. 1992, "Proline uptake by monolayers of human intestinal absorptive (Caco-2) cells in vitro", *Biochimica et biophysica acta biomembranes*, vol. 1104, no. 2, pp. 283–292.
- Nieuwenhuijzen, G.A.P., Deitch, E.A. & Goris, R.J.A. 1996, "The relationship between gut-derived bacteria and the development of the multiple organ dysfunction syndrome", *Journal of anatomy*, vol. 189, no. 3, pp. 537–548.
- Nikjoo, H. & Goodhead, D.T. 1989, "The relative biological effectiveness (RBE) achievable by high and low LET radiations", in *Low dose radiation: biological bases of risk assessment*, Eds. Baverstock, K. & Stather ,J., Taylor & Francis, London pp. 491-502.
- Norris, D.A., Puri, N. & Sinko, P.J. 1998, "The effect of physical barriers and properties on the oral absorption of particulates", *Advanced drug delivery reviews*, vol. 34, no. 2–3, pp. 135–154.

- Nunn, S., Gilmore, R.S., Dodge, J.A. & Carr, K.E. 1990, "Exudate variation in the rabbit gastrointestinal tract: a scanning electron microscope study", *Journal of anatomy*, vol. 170, pp. 87–98.
- Nusrat, A., Turner, J.R. & Madara, J.L. 2000, "Molecular physiology and pathophysiology of tight junctions. IV. Regulation of tight junctions by extracellular stimuli: nutrients, cytokines, and immune cells", *American journal of physiology – gastrointestinal and liver physiology*, vol. 279, no. 5, pp. G851–G857.
- Nwokolo, C.U., Lewin, J.F., Hudson, M. & Pounder, R.E. 1992, "Transmucosal penetration of bismuth particles in the human stomach", *Gastroenterology*, vol. 102, no. 1, pp. 163–167.
- Oatley, C.W., Nixon, W.C. & Pease, R.F.W. 1965, "Scanning electron microscope", *Advances in electronics and electron phys*ics, vol. 21, pp. 181–247.
- Obert, G., Peiffer, I. & Servin, A. L. 2000, "Rotavirus-induced structural and functional alterations in tight junctions of polarized intestinal caco-2 cell monolayers", *Journal of virology*, vol. 74, no. 10, pp. 4645–4651.
- Odor, D.L. 1954, "Observations of the rat mesothelium with the electron and phase microscopes", *American journal of anatomy*, vol. 95, no. 3, pp. 433-465.
- O'Hagan, D.T. 1990, "Intestinal translocation of particulates implications for drug and antigen delivery", Advanced drug delivery reviews, vol. 5, no. 3, pp. 265–285.
- Ohsugi, T., Kiuchi, Y., Shimoda, K., Oguri, S. & Maejima, K. 1996, "Translocation of bacteria from the gastrointestinal tract in immunodeficient mice", *Laboratory animals*, vol. 30, no. 1, pp. 46–50.
- Oliver, J.M. & Berlin, R.D. 1982, "Mechanisms that regulate the structural and functional architecture of cell surfaces" *International review of cytology*, vol. 74, pp. 55–94.
- Oliver, S.R., Phillips, N.A., Novosad, V.L., Bakos, M.P., Talbert, E.E. & Clanton, T.L. 2012, "Hyperthermia induces injury to the intestinal mucosa in the mouse: evidence for an oxidative stress mechanism", *American journal of physiology - regulatory integrative and comparative physiology*, vol. 302, no. 7, pp. R845–R853.
- Olivera, L., Canul, R.R., Pereira-Pacheco, F., Cockburn, J., Soldani, F., McKenzie, N.H., Duncan, M., Olvera-Novoa, M.A. & Grant, G. 2003, "Nutritional and physiological responses of young growing rats to diets containing raw cowpea seed meal, protein isolate (globulins), or starch", *Journal of agricultural and food chemistry*, vol.51, no. 1, pp.319-325.
- Osborne, D.L., Payne, S.C., Russ, R.D. & Tobin, B. 2000, "Comparison of therapeutic regimens in the amelioration of alterations in rat gastrointestinal mucosal DNA, RNA and protein induced by streptozotocin diabetes mellitus", *Life sciences*, vol. 66, no. 24, pp. 2405–2417.
- Oude Elferink, R.P.J.O., Bakker, C.T.M. & Jansen, P.L.M. 1993, "Glutathione-conjugate transport by human colon adenocarcinoma cells (Caco-2 cells)", *Biochemical journal*, vol. 290, no. 3, pp. 759–764.

- Owen, R.L. & Jones, A.L. 1974, "Epithelial cell specialization within human Peyer's patches: an ultrastructural study of intestinal lymphoid follicles", *Gastroenterology*, vol. 66, no. 2, pp. 189–203.
- Owen, R.L., Pierce, N.F., Apple, R.T., Cray, W.C. Jr. 1986, "M cell transport of Vibrio cholerae from the intestinal lumen into Peyer's patches: a mechanism for antigen sampling and for microbial transepithelial migration", *Journal of infectious diseases*, vol. 153, no. 6, pp. 1108–18.
- Ozeki, K., Kato, M., Sakurai, Y., Ishigai, M., Kudo, T. & Ito, K. 2015, "Evaluation of the appropriate time range for estimating the apparent permeability coefficient (Papp) in a transcellular transport study", *International journal of pharmaceutics*, vol. 495, no. 2, pp. 963–971.
- Özyurt, H., Özden, A.S., Cevik, O., Özgen, Z., Cadirci, S., Elmas, M.A., Ercan, F., Şener, G. & Gören, M.Z. 2014, "Investigation into the role of the cholinergic system in radiation-induced damage in the rat liver and ileum", *Journal of radiation research*, vol. 55, no. 5, pp. 866-875.
- Panozzo, J., Bertoncini, D., Miller, D., Libertin, C.R. & Wotoschak, G.E. 1991, "Modulation of expression of virus-like elements following exposure of mice to high- and low-LET radiations", *Carcinogenesis*, vol. 12, no. 5, pp. 801-804.
- Pappo, J. & Ermak, T.H. 1989, "Uptake and translocation of fluorescent latex particles by rabbit Peyer's patch follicle epithelium: a quantitative model for M cell uptake", *Clinical and experimental immunology*, vol. 76, no. 1, pp. 144–148.
- Pappo, J., Ermak, T.H. & Steger, H.J. 1991, "Monoclonal antibody-directed targeting of fluorescent polystyrene microspheres to Peyer's patch M cells", *Immunology*, vol. 73, no. 3, pp. 277–280.
- Parihar, V.K., Prabhakar, K.R., Veerapur, V.P., Kumar, M.S., Reddy, Y.R., Joshi, R., Unnikrishnan, M.K. & Rao, C.M.
 2006, "Effect of sesamol on radiation-induced cytotoxicity in Swiss albino mice", *Mutation research genetic toxicology and environmental mutagenesis*, vol. 611, no. 1–2, pp. 9–16.
- Parlesak, A., Haller, D., Brinz, S., Baeuerlein, A. & Bode, C. 2004, "Modulation of cytokine release by differentiated Caco-2 cells in a compartmentalized coculture model with mononuclear leucocytes and nonpathogenic bacteria", *Scandinavian journal of immunology*, vol. 60, no. 5, pp. 477–485.
- Parmley, R.T., Barton J.C. & Conrad, M.E. 1984, "Ultrastructural cytochemical identification of the siderophilic enterocyte", *Journal of histochemistry and cytochemistry*, vol. 32, no. 7, pp. 724 730.
- Patel, A.J., Bailey, P. & Balázs, R. 1979, "Effect of reserpine on cell proliferation and energy stores in the developing rat brain". *Neuroscience*, vol. 4, no. 1, pp. 139–143.
- Patel, D.D., Bansal, D.D., Mishra, S., Arora, R., Sharma, A., Jain, S.K. & Kumar, R. 2012, "Radioprotection to small intestine of the mice against ionizing radiation by semiquinone glucoside derivative (SQGD) isolated from Bacillus sp. INM-1", *Molecular and cellular biochemistry*, vol. 370, no. 1–2, pp. 115–125.

- Paterson, C.A., Anvari, M., Tougas, G. & Huizinga, J.D. 2000, "Determinants of occurrence and volume of transpyloric flow during gastric emptying of liquids in dogs ", *Digestive diseases and sciences*, vol. 45, no. 8, pp. 1509–1516.
- Pawley, J.B. & Norton, J.T. 1978, "A chamber attached to the SEM for fracturing and coating frozen biological samples", *Journal of microscopy*, vol. 112, no. 2, pp. 169–82.
- Peuranen, S., Tiihonen, K., Apajalahti, J., Kettunen, A., Saarinen, M. & Rautonen, N. 2004, "Combination of polydextrose and lactitol affects microbial ecosystem and immune responses in rat gastrointestinal tract", *British journal of nutrition*, vol. 91, no. 6, pp.905–914.
- Picard, C., Ksas, B., Griffiths, N.M. & Fioramonti, J. 2002, "Effect of granisetron on radiation-induced alterations of colonic motility and fluid absorption in rats", *Alimentary pharmacology and therapeutics*, vol. 16, no. 3, pp.623-631.
- Pietzonka, P., Walter, E., Duda-Johner, S., Langguth, P. & Merkle, H.P. 2002a, "Compromised integrity of excised porcine intestinal epithelium obtained from the abattoir affects the outcome of in vitro particle uptake studies", *European journal of pharmaceutical sciences*, vol. 15, no. 1, pp. 39–47.
- Pietzonka, P., Rothen-Rutishauser, B., Langguth, P., Wunderli-Allenspach, H., Walter, E. & Merkle, H.P. 2002b,
 "Transfer of lipophilic markers from PLGA and polystyrene nanoparticles to Caco-2 monolayers mimics particle uptake", *Pharmaceutical research*, vol. 19, no. 5, pp. 595–601.
- Pinto, H.C., Portela-Gomes, G.M., Grimelius, L., Kohnert, K.D., de Sousa, J.C. & Albuquerque, M.A.P. 1995, "The distribution of endocrine cell types of the gastrointestinal mucosa in genetically diabetic (db/db) mice", *Gastroenterology*, vol. 108, no. 4, pp. 967–974.
- Playford, R.J., Marchbank, T., Goodlad, R.A., Chinery, R.A., Poulsom, R., Hanby, A.M. & Wright, N.A. 1996, "Transgenic mice that overexpress the human trefoil peptide pS2 have an increased resistance to intestinal damage", *Proceedings of the national academy of sciences of the United States of America*, vol. 93, no. 5, pp. 2137–2142.
- Plummer, A.J., Barrett, W.E. & Rutledge, R. 1955, "The action of reserpine on the motility of the digestive tract", *American journal of digestive diseases*, vol. 22, no. 11, pp. 337–342
- Poley, J.R., 1983, "Chronic nonspecific diarrhea in children: investigation of the surface morphology of small bowel mucosa utilizing the scanning electron microscope", *Journal of pediatric gastroenterology and nutrition*, vol. 2, no.1, pp. 71–94.
- Ponder, B.A.J., Schmidt, G.H. & Wilkinson, M.M. 1986, "Immunohistochemistry in the analysis of mouse aggregation chimaeras", *The histochemical journal*, vol. 18, no. 5, pp. 217–227.
- Porter, E.M., Bevins, C.L., Ghosh, D. & Ganz, T. 2002, "The multifaceted Paneth cell", *Cellular and molecular life sciences*, vol. 59, no. 1, pp. 156–170.

- Porvaznik, M. 1979, "Tight junction disruption and recovery after sublethal γ irradiation", *Radiation research*, vol. 78, no. 2, pp. 233–250.
- Pothier, P. & Hugon, J.S., 1980, "Characterization of isolated villus and crypt cells from the small intestine of the adult mouse", *Cell and tissue research*, vol. 211, no. 3, pp. 405–418.
- Potten, C.S. 1990, "A comprehensive study of the radiobiological response of the murine (BDF1) small intestine", International journal of radiation biology, vol. 58, no. 6, pp. 925–973.
- Potten, C.S., Hendry, J.H., Moore J.W. & Chwalinski, S. 1983, "Cytotoxic effects in gastro-intestinal epithelium (as exemplified by small intestine) ", In *Cytotoxic insult to tissue. Effects on cell lineages*, Eds., Potten, C.S. & Hendry, J.H., Churchill Livingstone Edinburgh, pp. 105–152.
- Powell, J.J., Faria, N., Thomas-McKay, E. & Pele, L.C. 2010, "Origin and fate of dietary nanoparticles and microparticles in the gastrointestinal tract", *Journal of autoimmunity*, vol. 34, no. 3, pp. J226–J233.
- Prabhakar, K.R., Veerapur, V.P., Bansal, P., Parihar, V.K., Reddy Kandadi, M., Bhagath Kumar, P., Priyadarsini, K.I.
 & Unnikrishnan, M.K. 2007, "Antioxidant and radioprotective effect of the active fraction of Pilea microphylla (L.) ethanolic extract", *Chemico-biological interactions*, vol. 165, no. 1, pp. 22–32.
- Prokop, O. 1990, "The Herbst-Volkheimer-effect", *Kitazato archives of experimental medicine,* vol. 63, no. 1, pp. 1-6.
- Quarterman, J. & Morrison, E. 1978, "The effect of age on the absorption and excretion of lead", *Environmental research*, vol. 17, no. 1, pp. 78–83
- Quastler, H. & Hampton, J.C. 1962, "Effects of ionizing radiation on the fine structure and function of the intestinal epithelium of the mouse I, villus epithelium", *Radiation research*, vol. 17, no. 6, pp. 914–931.
- Raftery, A. T., Slater, N. D., & Cope, G. H., 1989, "Clinical anatomy of the peritoneal mesothelium", *Clinical Anatomy*, vol. 2, pp. 69–85.
- Rajaraman, R., Rounds, D.E., Yen, S.P.S. & Rembaum, A. 1974, "A scanning electron microscope study of cell adhesion and spreading in vitro", *Experimental cell research*, vol. 88, no. 2, pp. 327–339.
- Rakieten, N., Rakieten, M.L. & Nadkarni, M.V. 1963, "Studies on the diabetogenic action of streptozotocin (NSC-37917)", *Cancer chemotherapy reports*, vol. 29, pp. 91–98.
- Ramazani, F., Chen, W., Van Nostrum, C.F., Storm, G., Kiessling, F., Lammers, T., Hennink, W.E. & Kok, R.J. 2016,
 "Strategies for encapsulation of small hydrophilic and amphiphilic drugs in PLGA microspheres: state-of-the-art and challenges", *International journal of pharmaceutics*, vol. 499, no. 1–2, pp. 358–367.
- Ranaldi, G., Consalvo, R., Sambuy, Y. & Scarino, M.L. 2003, "Permeability characteristics of parental and clonal human intestinal Caco-2 cell lines differentiated in serum-supplemented and serum-free media", *Toxicology in vitro*, vol. 17, no. 5–6, pp. 761–767.

- Randall, K.J., Turton, J. & Foster, J.R. 2011, "Explant culture of gastrointestinal tissue: a review of methods and applications", *Cell biology and toxicology*, vol. 27, no. 4, pp. 267–284.
- Raymond, C., Anne, V. & Millane, G. 1991, "Development of esophageal epithelium in the fetal and neonatal mouse", *The anatomical record*, vol. 230, no. 2, pp. 225–234.
- Reber, H.A., Roberts, C. & Way, L.W. 1979. "The pancreatic duct mucosal barrier", *American journal of surgery*, vol. 137, pp. 128–134.
- Regoli, M., Bertelli, E., Borghesi, C. & Nicoletti, C. 1995, "Three-dimensional (3D-) reconstruction of M cells in rabbit Peyer's patches: definition of the intraepithelial compartment of the follicle-associated epithelium", *The anatomical record*, vol. 243, no. 1, pp. 19–26.
- Rehnberg, G.L., Hein, J.F., Carter, S.D., Linko, R.S. & Laskey, J.W. 1981, "Chronic ingestion of Mn3O4 by young rats: tissue accumulation, distribution, and depletion", *Journal of toxicology and environmental health, part A, current issues*, vol. 7, no. 2, pp.263–272.
- Reineke, J.J., Cho, D.Y., Dingle, Y., Morello III, A.P., Jacob, J., Thanos, C.G. & Mathiowitz, E. 2013, "Unique insights into the intestinal absorption, transit, and subsequent biodistribution of polymer-derived microspheres", *Proceedings of the national academy of sciences of the United States of America*, vol. 110, no. 34, pp. 13803– 13808.
- Richter, K.K., Fagerhol, M.K., Carr, J.C., Winkler, J.M., Sung, C.C. & Hauer-Jensen, M. 1997, "Association of granulocyte transmigration with structural and cellular parameters of injury in experimental radiation enteropathy", *Radiation oncology investigations*, vol. 5, no. 6, pp. 275–282.
- Robblee, N.M., McLellan, E.A. & Bird, R.P. 1989, "Measurement of the proliferative status of colonic epithelium as a risk marker for colon carcinogenesis: effect of bile acid and dietary fiber", *Nutrition and cancer*, vol. 12, no. 4, pp. 301–310.
- Robert, A. 1979, "Cytoprotection by prostaglandins", *Gastroenterology*, vol. 77, no. 4, pp. 761-767.
- Robertson, A.G., Wilson, P., Wilson, D.J., Carr, K.E. & Hunter, I. 1987, "Microplication patterns on human buccal epithelia following radiotherapy: a scanning electron microscopic analysis", *Journal of submicroscopic cytology and pathology*, vol. 19, no. 3, pp. 515–521.
- Roessler, F., Grossenbacher, R., Stanisic, M. & Walt, H. 1991, "Correlative histological and ultrastructural study of unusual changes in human tracheobronchial epithelium". *The laryngoscope*, vol. 101, no. 5, pp.473–479.
- Rohner, F., Ernst, F.O., Arnold, M., Hilbe, M., Biebinger, R., Ehrensperger, F., Pratsinis, S.E., Langhans, W., Hurrell,
 R.F. & Zimmermann, M.B. 2007, "Synthesis, characterization, and bioavailability in rats of ferric phosphate nanoparticles", *Journal of nutrition*, vol. 137, no. 3, pp. 614–619.
- Ropenga, A., Combes, O., Chapel, A. & Griffiths, N.M. 2004, "Persistence of altered 5-hydroxytryptamine turnover following hemibody X-irradiation in the rat distal colon", *International journal of radiation biology*, vol. 80, no. 6, pp. 399–411.
- Ross, G.A. & Mayhew, T.M. 1984, "Effects of fasting on villi along the small intestine: a stereological approach to the problem of quantifying villus 'shape'", *Experientia*, vol. 40, no. 8, pp. 856–858.
- Rubin, C. E., Brandborg, L. L., Phelps, P. C., & Taylor, H. C., Jr. 1960, "Studies of celiac disease. I. The apparent identical and specific nature of the duodenal and proximal jejunal lesion in celiac disease idiopathic sprue", *Gastroenterology*, vol. 38, pp. 28–49.
- Rubin, P. 1989, "Law and order of radiation sensitivity. Absolute vs relative", In *Radiation tolerance of normal tissues*, Karger Publishers, vol. 23, pp. 7-40.
- Rubio, C.A. & Jalnäs, M. 1996, "Dose-time-dependent histological changes following irradiation of the small intestine of rats", *Digestive diseases and sciences*, vol. 41, no. 2, pp. 392–401.
- Sagher, F.A., Dodge, J.A., Johnston, C.F., Shaw, C., Buchanan, K.D. & Carr, K.E. 1991, "Rat small intestinal morphology and tissue regulatory peptides: effects of high dietary fat", *British journal of nutrition*, vol. 65, no. 1, pp. 21–28.
- Saleh, K.M., Toner, P.G., Carr, K.E. & Hughes, H.E. 1982, "An improved method for sequential light and scanning electron microscopy of the same cell using localising microcoverslips", *Journal of clinical pathology*, vol. 35, no. 5, pp. 576-580.
- Salin, C.A., Samanta, N. & Goel, H.C. 2001, "Protection of mouse jejunum against lethal irradiation by Podophyllum hexandrum", *Phytomedicine*, vol. 8, no. 6, pp. 413–422.
- Sambuy, Y., De Angelis, I., Ranaldi, G., Scarino, M.L., Stammati, A. & Zucco, F. 2005, "The Caco-2 cell line as a model of the intestinal barrier: influence of cell and culture-related factors on Caco-2 cell functional characteristics", *Cell biology and toxicology*, vol. 21, no. 1, pp. 1–26.
- Samonis, G., Falagas, M.E., Lionakis, S., Ntaoukakis, M., Kofteridis, D.P., Ntalas, I. & Maraki, S. 2011, "Saccharomyces boulardii and Candida albicans experimental colonization of the murine gut", *Medical mycology*, vol. 49, no. 4, pp. 395–399.
- Sanders, E.J. & Singal, P.K. 1973, "Visualization of the outer and interblastomeric surface of early embryos of Xenopus laevis by scanning electron microscopy", *Micron*, vol. 4, pp. 156–162.
- Sandoz, P. & Jacquot, M. 2011, "Lensless vision system for in-plane positioning of a patterned plate with subpixel resolution", *Journal of the optical society of America A: optics and image science, and vision*, vol. 28, no. 12, pp. 2494–2500.
- Sass, W., Dreyer, H.P. & Seifert, J. 1990, "Rapid insorption of small particles in the gut", *American journal of gastroenterology*, vol. 85, no. 3, pp. 255–260.

- Satoh, H., Inada, I., Hirata, T. & Maki, Y. 1981, "Indomethacin produces gastric antral ulcers in the refed rat", *Gastroenterology*, vol. 81, no. 4, pp. 719-725.
- Satsu, H., Ishimoto, Y., Nakano, T., Mochizuki, T., Iwanaga, T. & Shimizu, M. 2006, "Induction by activated macrophage-like THP-1 cells of apoptotic and necrotic cell death in intestinal epithelial Caco-2 monolayers via tumor necrosis factor-alpha", *Experimental cell research*, vol. 312, no. 19, pp. 3909-3919.
- Scanff, P., Souidi, M., Grison, S., Griffiths, N.M. & Gourmelon, P. 2004, "Alteration of the enterohepatic recirculation of bile acids in rats after exposure to ionizing radiation", *Canadian journal of physiology and pharmacology,* vol. 82, no. 2, pp. 114–124.
- Schleh, C., Semmler-Behnke, M., Lipka, J., Wenk, A., Hirn, S., Schäffler, M., Schmid, G., Simon, U. & Kreyling, W.G. 2012, "Size and surface charge of gold nanoparticles determine absorption across intestinal barriers and accumulation in secondary target organs after oral administration", *Nanotoxicology*, vol. 6, no. 1, pp. 36-46.
- Schmidt, G.H., Wilkinson, M.M. & Ponder, B.A.J. 1985, "Cell migration pathway in the intestinal epithelium: an in situ marker system using mouse aggregation chimeras", *Cell*, vol. 40, no. 2, pp. 425-429.
- Seifert, J., Haraszti, B. & Sass, W. 1996, "The influence of age and particle number on absorption of polystyrene particles from the rat gut", *Journal of anatomy*, vol. 189, no. 3, pp. 483–486.
- Senesse, P., Janiszewski, C., Flori, N., Vercambre, L., Bories, C. & Assenat, E. 2009, "Oral nutritional supplements for cancer patients: who could benefit?", *Oncologie*, vol. 11, no. 4, pp. 231–235.
- Shanthakumar, J., Karthikeyan, A., Bandugula, V.R. & Rajendra Prasad, N. 2012, "Ferulic acid, a dietary phenolic acid, modulates radiation effects in Swiss albino mice", *European journal of pharmacology*, vol. 691, no. 1–3, pp. 268–274.
- Sharma, R., Schumacher, U., Ronaasen, V. & Coates, M. 1995, "Rat intestinal mucosal responses to a microbial flora and different diets", *Gut*, vol. 36, no. 2, pp. 209–214.
- Shasha'a, S., Dickson, G.R., Gilmore, R.S., Crean, G.C., Butt, M.M. & Carr, K.E., 1993, "Rabbit and human nonkeratinising stratified squamous oesophageal epithelium displays similar microridge structure by scanning electron microscopy", *Scanning microscopy*, vol. 7, no. 3, pp. 953–959.
- Shaw, R. K., Smollett, K., Cleary, J., Garmendia, J., Straatman-Iwanowska, A., Frankel, G., & Knutton, S. 2005,
 "Enteropathogenic Escherichia coli type III effectors EspG and EspG2 disrupt the microtubule network of intestinal epithelial cells", *Infection and immunity*, vol. 73, no. 7, pp. 4385–4390.
- Sherman, P., Fleming, N., Forstner, J., Roomi, N. & Forstner, G. 1987, "Bacteria and the mucus blanket in experimental small bowel bacterial overgrowth". The American journal of pathology, vol. 126, no. 3, pp. 527–534.

- Shi, S.R., Chaiwun, B., Young, L., Cote, R.J. & Taylor, C.R. 1993, "Antigen retrieval technique utilizing citrate buffer or urea solution for immunohistochemical demonstration of androgen receptor in formalin-fixed paraffin sections", *Journal of histochemistry and cytochemistry*, vol. 41, no. 11, pp. 1599–1604.
- Shields, H.M., Zwas, F., Antonioli, D.A., Doos, W.G., Kim, S. & Spechler, S.J. 1993, "Detection by scanning electron microscopy of a distinctive esophageal surface cell at the junction of squamous and Barrett's epithelium", *Digestive diseases and sciences*, vol. 38, no. 1, pp. 97–108.
- Siew, S. & Goldstein, M.L., 1981, "Scanning electron microscopy of mucosal biopsies of the human upper gastrointestinal tract", *Scanning electron microscopy*, vol. 4, pp. 173–181.
- Sikorski, J. 1960, "Studies of fibrous structures," in *Proceedings of the fourth international conference on electron microscopy, Berlin Sept. 1958, physics*, vol.1, Eds. Bargmann, W., Möllenstedt, G., Niehrs, H., Peters, D., Ruska, E. & Wolpers, C., Springer Verlag, Berlin, pp 686–707. In Oatley, C.W., 1982, "The early history of the scanning electron microscope", *Journal of applied physics*, vol. 53, pp. R1-R13.
- Simon, L., Shine, G. & Dayan, A.D. 1994, "Effect of animal age on the uptake of large particulates across the epithelium of the rat small intestine", *International journal of experimental pathology*, vol. 75, no. 5, pp. 369–373.
- Simon, L., Warren, I. & Dayan, A.D. 1997, "Effect of solid and liquid diet on uptake of large particulates across intestinal epithelium in rats", *Digestive diseases and sciences*, vol. 42, no. 7, pp. 1519–1523.
- Simpson, C.J., Toner, P.G., Carr, K.E., Anderson, J.D. & Carter, D.C. 1983, "Effect of bile salt perfusion and intraduct pressure on ionic flux and mucosal ultrastructure in the pancreatic duct of the cat", *Virchows archiv abteilung B cell pathology*, vol. 42, no. 3, pp. 327–342.
- Smith, M.W. & Jarvis, L.G. 1980, "Use of differential interference contrast microscopy to determine cell renewal times in mouse intestine", *Journal of microscopy*, vol. 118, no. 2, pp.153–159.
- Smyth, S.H., Doyle-McCullough, M., Cox, O.T. & Carr, K.E. 2005, "Effect of reproductive status on uptake of latex microparticles in rat small intestine", *Life sciences*, vol. 77, no. 26, pp. 3287–3305.
- Smyth, S.H., Feldhaus, S., Schumacher, U. & Carr, K.E. 2008, "Uptake of inert microparticles in normal and immune deficient mice", *International journal of pharmaceutics*, vol. 346, no. 1–2, pp. 109–118.
- Snoeck, V., Goddeeris, B. & Cox, E. 2005, "The role of enterocytes in the intestinal barrier function and antigen uptake", *Microbes and infection*, vol. 7, no. 7–8, pp. 997–1004.
- Solheim, K.E., Lærum, F., Stordahl, A. & Aase, S. 1991, "Urinary excretion of iohexol after enteral administration in rats with radiation injury of the small intestine", *Scandinavian journal of gastroenterology*, vol. 26, no. 10, pp. 1097–1106.
- Somosy, Z. 2000, Radiation response of cell organelles. *Micron*, vol. 31, 165–181.

- Somosy, Z., Horváth, G., Telbisz, Á., Réz, G. & Pálfia, Z. 2002, "Morphological aspects of ionizing radiation response of small intestine", *Micron*, vol. 33, no. 2, pp. 167–178.
- Somosy, Z., Kovacs, J., Siklos, L., Koteles, G.J. & Seed, T.M. 1993, "Morphological and histochemical changes in intercellular junctional complexes in epithelial cells of mouse small intestine upon X-irradiation: changes of ruthenium red permeability and calcium content", *Scanning microscopy*, vol. 7, no. 3, pp. 961–971.
- Specian, R.D. & Oliver, M.G. 1991, "Functional biology of intestinal goblet cells", *American journal of physiology,* vol. 260, pp. C183–C193.
- Spitzer, T.R. 1995, "Clinical aspects of irradiation induced alimentary tract injury", in *Radiation and the gastrointestinal tract*, Eds. Dubois, A., King, G.L. & Livengood, D.R., CRC Press, Boca Raton, FL, pp. 3-19.
- Sprügel, W., Schubert, E., Mitznegg, P., Heim, F., Hasl, G. & Pauly, H. 1977, "Immediate X-radiation induced contractions of the isolated guinea pig terminal ileum: the localization of the effect by drugs", *Radiation and environmental biophysics*, vol. 14, no. 1, pp. 31–37.
- Srinivasu, B.Y., Mitra, G., Muralidharan, M., Srivastava, D., Pinto, J., Thankachan, P., Suresh, S., Shet, A., Rao, S., Ravikumar, G., Thomas, T.S., Kurpad, A.V. & Mandal, A.K. 2015, "Beneficiary effect of nanosizing ferric pyrophosphate as food fortificant in iron deficiency anemia: evaluation of bioavailability, toxicity and plasma biomarker", *RSC (Royal society of chemistry) advances*, vol. 5, no. 76, pp. 61678–61687.
- Steffens, K.J. 1995, "Persorption—criticism and agreement as based upon in vitro and in vivo studies on mammals" in *Absorption of orally administered enzymes*, Eds. Gardner, M.L.G. & Steffens, K.J., Springer, Berlin, Heidelberg, pp. 9-21.
- Steinberg, GK, Fabrikant, JI., Marks, M.P., Levy, R.P., Frankel, K.A., Phillips, M.H., Shuer, L.M. & Silverberg, G.D. 1990, "Stereotactic heavy-charged-particle Bragg-peak radiation for intracranial arteriovenous malformations", *New England journal of medicine*, vol. 323, no. 2, pp. 96–101.
- Stenson, W.F., Easom, R.A., Riehl, T.E. & Turk, J. 1993, "Regulation of paracellular permeability in Caco-2 cell monolayers by protein kinase C", *American journal of physiology – gastrointestinal and liver physiology,* vol. 265, no. 5, pp. G955–G962.
- Stern, J.R., Olivas, A.D., Valuckaite, V., Zaborina, O., Alverdy, J.C. & An, G. 2013, "Agent-based model of epithelial host-pathogen interactions in anastomotic leak", *Journal of surgical research*, vol. 184, no. 2, pp.730-738.
- Stevanović, M., Savanović, I., Uskoković, V., Škapin, S.D., Bračko, I., Jovanović, U. & Uskoković, D. 2012, "A new, simple, green, and one-pot four-component synthesis of bare and poly(α,γ,l-glutamic acid)-capped silver nanoparticles", *Colloid and polymer science*, vol. 290, no. 3, pp. 221-231.
- Stitt, F.W., Clayton, D.G., Crawford, M. D. & Morris, J.N. 1973, "Clinical and biochemical indicators of cardiovascular disease among men living in hard and soft water areas", *The lancet*, vol. 301, no. 7795, pp. 122–126.

- Stone, H.H., Kolb, L.D., Currie, C.A., Geheber, C.E. & Cuzzell, J.Z. 1974, "Candida sepsis: pathogenesis and principles of treatments", *Annals of surgery*, vol. 179, no. 5, pp. 697–711.
- Strup-Perrot, C., Vozenin, M.C., Monceau, V., Pouzoulet, F., Petit, B., Holler, V., Perrot, S., Desquibert, L., Fouquet, S., Souquere, S., Pierron, G., Rousset, M., Thenet, S., Cardot, P., Benderitter, M., Deutsch, E. & Aigueperse, J. 2016, "PrP(c) deficiency and dasatinib protect mouse intestines against radiation injury by inhibiting of c-Src," *Radiotherapy and oncology*, vol. 120, no. 1, pp. 175–183.
- Suciu, D. 1985, "Morphometric study of the interphase nucleus in some radiosensitive and radioresistant mammalian cells", *Journal of theoretical biology*, vol. 113, no. 4, pp. 599–609.
- Sugano, K. 2010, "Possible reduction of effective thickness of intestinal unstirred water layer by particle drifting effect", *International journal of pharmaceutics*, vol. 387, no. 1–2, pp. 103–109.
- Sugano, K. & Terada, K. 2015, "Rate- and extent-limiting factors of oral drug absorption: theory and applications", *Journal of pharmaceutical sciences*, vol. 104, no. 9, pp. 2777–2788.
- Summers, R.W., Glenn, C.E., Flatt, A.J. & Elahmady, A. 1991, "Radiation and indomethacin effects on morphology, prostaglandins, and motility in dog jejunum", *American journal of physiology – gastrointestinal and liver physiology*, vol. 261, no. 124–1, pp. G145–G151.
- Summers, R.W., Maves, B.V., Reeves, R.D., Arjes, L.J. & Oberley, L.W. 1989, "Irradiation increases superoxide dismutase in rat intestinal smooth muscle", *Free radical biology and medicine*, vol. 6, no. 3, pp. 261–270.
- Swanson, R.S., Shipley, W.U., Tepper, J.E. & Warshaw, A.L. 1988, "Intraoperative radiotherapy in recent advances in surgery", in *Recent advances in surgery*, Ed. Russell, R.C.G., Churchill Livingstone, Edinburgh, pp. 83–102
- Swift, J.A. & Marsh, M.N. 1968, "Scanning electron microscopy of rat intestinal microvilli", *The lancet*, vol. 2, no. 7574, pp. 915.
- Syljuåsen, R.G., Belldegrun, A., Tso, C., Withers, H.R. & McBride, W.H. 1997, "Sensitization of renal carcinoma to radiation using alpha interferon (IFNA) gene transfection", *Radiation research*, vol. 148, no. 5, pp. 443-448.
- Tabata, Y., Inoué, Y. & Ikada, Y. 1996, "Size effect on systemic and mucosal immune responses induced by oral administration of biodegradable microspheres", *Vaccine*, vol. 14, no. 17–18, pp. 1677–1685.
- Takeuchi, A. & Zeller, J.A. 1972, "Scanning electron microscopic observations on the surface of the normal and spirochete-infested colonic mucosa of the rhesus monkey", *Journal of ultrastructure research*, vol. 40, no. 3-4, pp. 313–324.
- Tamura, M. & Suzuki, H. 1997, "Effects of pectin on jejunal and ileal morphology and ultrastructure in adult mice", Annals of nutrition and metabolism, vol. 41, no. 4, pp. 255–259.
- Taylor, A.B. & Anderson, J.H. 1972, "Scanning electron microscope observations of mammalian intestinal villi, intervillus floor and crypt tubules", *Micron*, vol. 3, pp. 430–453.

- Thomas, N.W., Jenkins, P.G., Howard, K.A., Smith, M.W., Lavelle, E.C., Holland, J. & Davis, S.S. 1996, "Particle uptake and translocation across epithelial membranes", *Journal of anatomy*, vol. 189, no. 3, pp. 487–490.
- Thomas, A.G., Phillips, A.D. & Walker-Smith, J.A. 1992, "The value of proximal small intestinal biopsy in the differential diagnosis of chronic diarrhoea", *Archives of disease in childhood*, vol. 67, no. 6, pp. 741-744.
- Thornton, P. 1965, "The scanning electron microscope", *Science journal*, vol. 1, pp. 2-7.
- Thornton, P. 1968, "Scanning electron microscopy", Chapman and Hall, London.
- Tolmach, L.J. & Marcus, P.I. 1960, "Development of X-ray induced giant HeLa cells", *Experimental cell research*, vol. 20, no. 2, pp. 350–360.
- Toner, P.G. 1965, "The fine structure of the globule leucocyte in the fowl intestine", *Acta anatomica (Basel)*, vol. 61, pp. 321–330.
- Toner, P.G. & Carr, K.E. 1968, "Cell structure: an introduction to biological electron microscopy", E & S Livingstone, Edinburgh.
- Toner, P.G. & Carr, K.E., 1971, "Cell structure: an introduction to biological electron microscopy" (2nd edition), E & S Livingstone, Edinburgh.
- Toner, P.G., Carr, K.E. & Wyburn, G.M. 1971. "The digestive system an ultrastructural atlas and review", (excerpt), Butterworths, London.
- Toothaker, R.D. & Elmer, G.W. 1984, "Prevention of clindamycin-induced mortality in hamsters by Saccharomyces boulardii", *Antimicrobial agents and chemotherapy*, vol. 26, no. 4, pp. 552–556.
- Tria, S.A., Jimison, L.H., Hama, A., Bongo, M. & Owens, R.M. 2013a, "Validation of the organic electrochemical transistor for in vitro toxicology", *Biochimica et biophysica acta – general subjects*, vol. 1830, no. 9, pp. 4381– 4390.
- Tria, S.A., Jimison, L.H., Hama, A., Bongo, M. & Owens, R.M. 2013b, "Sensing of EGTA mediated barrier tissue disruption with an organic transistor", *Biosensors*, vol. 3, no. 1, pp. 44-57.
- Trott, K.R., Tamou, S., Sassy, T. & Kiszel, Z. 1995, "The effect of irradiated volume on the chronic radiation damage of the rat large bowel", *Strahlentherapie und onkologie*, vol. 171, no. 6, pp. 326–331.
- Tsilibary, E.C. & Wissig, S.L. 1977, "Absorption from the peritoneal cavity: SEM study of the mesothelium covering the peritoneal surface of the muscular portion of the diaphragm", *American journal of anatomy*, vol. 149, no. 1, pp. 127–133.
- Tsubouchi, S., Kano, E., Nishimoto, Y. & Nakamura, W. 1984, "The effect of microwave-induced whole-body hyperthermia on the mucosa of mouse small intestine", *Journal of radiation research*, vol. 25, no. 2, pp. 131–139.

- Tsuji, T., Lavker, R.M. & Kligman, A.M. 1979, "A new method for scanning electron microscopic visualization of dermal elastic fibres", *Journal of microscopy*, vol. 115, no. 2, pp. 165–173.
- Tyrer, P., Ruth Foxwell, A., Kyd, J., Harvey, M., Sizer, P. & Cripps, A. 2002, "Validation and quantitation of an in vitro M-cell model", *Biochemical and biophysical research communications*, vol. 299, no. 3, pp. 377-383.
- Uddin, A.N., Bejugam, N.K., Gayakwad, S.G., Akther, P. & D'Souza, M.J. 2009, "Oral delivery of gastro-resistant microencapsulated typhoid vaccine", *Journal of drug targeting*, vol. 17, no. 7, pp. 553–560.
- Uematsu, M., Mohamed, Y.H., Onizuka, N., Ueki, R., Inoue, D., Fujikawa, A., Sasaki, H. & Kitaoka, T. 2016, "Less invasive corneal transepithelial electrical resistance measurement method", *Ocular surface*, vol. 14, no. 1, pp. 37–42.
- Umemiya, R., Fukuda, M., Fujisaki, K. & Matsui, T. 2005, "Electron microscopic observation of the invasion process of Cryptosporidium parvum in severe combined immunodeficiency mice", *Journal of parasitology*, vol. 91, no. 5, pp. 1034–1039.
- Unmack, M.A., Hansen, M.B., Grondahl, M.L., Olsen, J.E., Christensen, P. & Skadhauge, E. 2001, "Effects of indomethacin on Salmonella typhimurium- and Cholera toxin-induced fluid accumulation in the porcine small intestine", *Journal of veterinary medicine series A: physiology pathology clinical medicine*, vol. 48, no. 3, pp. 153–163.
- Uskoković, V., Lee, K., Lee, P.P., Fischer, K.E. & Desai, T.A. 2012b, "Shape effect in the design of nanowire-coated microparticles as transepithelial drug delivery devices", *ACS (American chemical society) Nano*, vol. 6, no. 9, pp. 7832–7841.
- Uskoković, V., Lee, P.P., Walsh, L.A., Fischer, K.E. & Desai, T.A. 2012a, "PEGylated silicon nanowire coated silica microparticles for drug delivery across intestinal epithelium", *Biomaterials*, vol. 33, no. 5, pp. 1663–1672.
- van der Lubben, I.M., van Opdorp, F.A.C., Hengeveld, M.R., Onderwater, J.J.M., Koerten, H.K., Verhoef, J.C., Borchard, G. & Junginger, H.E. 2002, "Transport of chitosan microparticles for mucosal vaccine delivery in a human intestinal M-cell model", *Journal of drug targeting*, vol. 10, no. 6, pp. 449-456.
- van't Hof, W. & van Meer, G. 1990, "Generation of lipid polarity in intestinal epithelial (Caco-2) cells: sphingolipid synthesis in the Golgi complex and sorting before vesicular traffic to the plasma membrane", *Journal of cell biology*, vol. 111, no. 3, pp. 977–986.
- Velikov, K.P. & Pelan, E. 2008, "Colloidal delivery systems for micronutrients and nutraceuticals", *Soft matter,* vol. 4, no. 10, pp. 1964–1980.
- Venkatesh, M., Mukherjee, S., Wang, H., Li, H., Sun, K., Benechet, A.P., Qiu, Z., Maher, L., Redinbo, M.R., Phillips, R.S., Fleet, J.C., Kortagere, S., Mukherjee, P., Fasano, A., Le Ven, J., Nicholson, J.K., Dumas, M.E., Khanna, K.M. & Mani, S. 2014, "Symbiotic bacterial metabolites regulate gastrointestinal barrier function via the xenobiotic sensor PXR and toll-like receptor 4", *Immunity*, vol. 41, no. 2, pp. 296–310.

- Verginadis, I.I., Kanade, R., Bell, B., Koduri, S., Ben-Josef, E. & Koumenis, C. 2017, "A novel mouse model to study image-guided, radiation-induced intestinal injury and preclinical screening of radioprotectors", *Cancer research*, vol. 77, no. 4, pp. 908–917.
- Vidon, N., Huchet, B. & Rambaud, J.C. 1986, "Effect of S. boulardii on water and sodium secretion induced by cholera toxin", *Gastroenterologie clinique et biologique*, vol. 10, no. 1, pp. 13–16.
- Vigneulle, R.M., Herrera, J., Gage, T., MacVittie, T.J., Taylor, P., Zeman, G., Nold, J.B. & Dubois, A. 1990, "Nonuniform irradiation of the canine intestine. I. Effects", *Radiation research*, vol. 121, no. 1, pp. 46–53.
- Vogel, P., Janke, L., Gravano, D.M., Lu M., Sawant, D.V., Bush, D., Shuyu, E., Vignali, D.A., Pillai, A. & Rehg, J.E. 2018,
 "Globule leukocytes and other mast cells in the mouse intestine", *Veterinary pathology*, vol. 55, no.1, pp. 76–97.
- Volkheimer, G. 1977, "Persorption of particles: physiology and pharmacology", *Advances in pharmacology and chemotherapy*, vol.14, pp. 163–187.
- Volkheimer, G. 2001, "The phenomenon of persorption: persorption, dissemination, and elimination of microparticles", In *Old Herborn University seminar monograph 14: intestinal translocation*, Eds. Heidt, P. J., Nieuwenhuis, P., Rusch, V. D. & van der Waaij, D., Herborn Litterae, Herborn-Dill, pp. 7 – 18.
- Volkheimer, G. & Schulz, F.H. 1968, "The phenomenon of persorption", Digestion, vol. 1, no. 4, pp. 213-218.
- Volkheimer, G., John, H., Wendland, H., Olejniczak, H., Werner, G., Schmidt, W., Wagner, G., Seebo, M. & Hackbarth, H. 1965, "Quantitative studies on the Herbst effect", *Deutsche gesundheitswesen,* vol. 20, no. 37, pp. 1685-1687.
- Volkheimer, G., Schulz, F.H., Aurich, I., Strauch, S., Beuthin, K. & Wendlandt, H. 1968a, "Persorption of particles", *Digestion*, vol. 1, no. 2, pp. 78–80.
- Volkheimer, G., Schulz, F.H., Lehmann, H., Aurich, I., Hübner, R., Hübner, M., Hallmayer, A., Münch, H., Oppermann,
 H. & Strauch, S. 1968b, "Primary portal transport of persorbed starch granules from the intestinal wall",
 Medicina experimentalis international journal of experimental medicine, vol. 18, no. 2, pp. 103–108.
- Volkheimer, G., Schulz, F., Hofmann, I., Pieser, J., Rack, O., Reichelt, G., Rothenbaecher, W., Schmelich, G., Schurig,
 B., Teicher, G. & Weiss, B. 1968c, "The effect of drugs on the rate of persorption.", *Pharmacology*, vol. 1, no. 1, pp. 8–14.
- Volkheimer, G., Schulz, F.H., Lindenau, A. & Beitz, U. 1969, "Persorption of metallic iron particles", *Gut*, vol. 10, no. 1, pp. 32–33.
- Volpe, D.A. 2008, "Variability in caco-2 and MDCK cell-based intestinal permeability assays", *Journal of pharmaceutical sciences*, vol. 97, no. 2, pp. 712–725.

- Wakeford, R. 2007, "The Windscale reactor accident 50 Years on", *Journal of radiological protection*, vol. 27, no. 3, pp. 211–215.
- Wakeford, R. 2011a, "The silver anniversary of the Chernobyl accident. Where are we now?", *Journal of radiological protection*, vol. 31, no. 1, pp. 1–7.
- Wakeford, R. 2011b, "And now, Fukushima", Journal of radiological protection, vol. 31, no. 2, pp. 167-176.
- Walsh, D. 1897, "Deep tissue traumatism from roentgen rat exposure", *British medical journal*, vol. 2, no. 1909, pp. 272–273.
- Walter, H., Krob, E.J., Ascher, G.S. & Seaman, G.V.F. 1973, "Partition of rat liver cells in aqueous dextranpolyethylene glycol phase systems", *Experimental cell research*, vol. 82, no. 1, pp. 15–26.
- Wang, J. & Hauer-Jensen, M. 2003, "Radiation toxicity and proteinase-activated receptors", *Drug development research*, vol. 60, no. 1, pp. 1–8.
- Wang, J., Richter, K.K., Sung, C.C. & Hauer-Jensen, M. 1999, "Upregulation and spatial shift in the localization of the mannose 6- phosphate/insulin-like growth factor II receptor during radiation enteropathy development in the rat", *Radiotherapy and oncology*, vol. 50, no. 2, pp. 205–213.
- Wang, J., Zheng, H., Sung, C.C., Richter, K.K. & Hauer-Jensen, M. 1998, "Cellular sources of transforming growth factor-β isoforms in early and chronic radiation enteropathy", *American journal of pathology*, vol. 153, no. 5, pp. 1531–1540.
- Wang, Q., Dickson, G.R., Abram, W.P. & Carr, K.E. 1994, "Electron irradiation slows down wound repair in rat skin: a morphological investigation", *British journal of dermatology*, vol. 130, no. 5, pp. 551–560.
- Wang, Q., Dickson, G.R. & Carr, K.E. 1996, "The effect of graft-bed irradiation on the healing of rat skin grafts", *Journal of investigative dermatology*, vol. 106, no. 5, pp. 1053–1057.
- Wartiovaara, J. & Tarpila, S. 1977, "Cell contacts and polysomes in irradiated human jejunal mucosa at onset of epithelial repair", *Laboratory investigation*, vol. 36, no. 6, pp. 660–665.
- Wassilev, W., Wedel, T., Michailova, K. & Kühnel, W. 1998, "A scanning electron microscopy study of peritoneal stomata in different peritoneal regions", *Annals of anatomy*, vol. 180, no. 2, pp. 137–143.
- Wei, W.I., Lam, K.H. & Lau, W.F. 1989, "Fate of skin element of pectoralis major flap in intraoral reconstruction", *Archives of otolaryngology-head and neck surgery*, vol. 115, no. 3, pp. 360–363.
- Wells, C.L., Maddaus, M.A., Erlandsen, S.L. & Simmons, R.L. 1988, "Evidence for the phagocytic transport of intestinal particles in dogs and rats", *Infection and immunity*, vol. 56, no. 1, pp. 278–282.
- Wells, O.C. 1960, "Correction of errors in electron stereomicroscopy", *British journal of applied physics*, vol. 11, no. 5, pp. 199–201.

- Wellwood, J.M. & Jackson, B.T. 1973, "The intestinal complications of radiotherapy", *British journal of surgery*, vol. 60, pp. 814–818.
- Whur, P. & Johnston, H.S. 1967, "Ultrastructure of globule leucocytes in immune rats infected with Nippostrongylus brasiliensis and their possible relationship to the Russell body cell", *Journal of pathology and bacteriology*, Vol. 93, no.1, pp. 81–85.
- Wiernik, G. 1966a, "Changes in the villous pattern of the human jejunum associated with heavy radiation damage", *Gut*, vol. 7, no. 2, pp. 149–153.
- Wiernik, G. 1966b, "Radiation damage and repair in the human jejunal mucosa", *Journal of pathology and bacteriology*, vol. 91, no. 2, pp. 389-393.
- Wiernik, G. & Plant, M. 1971, "The origin and kinetics of goblet cells in the human jejunum during irradiation", *The British journal of radiology*, vol. 44, no. 521, pp. 348-356.
- Wiesner, S.M., Bendel, C.M., Hess, D.J., Erlandsen, S.L. & Wells, C.L. 2002, "Adherence of yeast and filamentous forms of Candida albicans to cultured enterocytes", *Critical care medicine*, vol. 30, no. 3, pp. 677–683.
- Williams, J.P., Brown, S.L., Georges, G.E., Hauer-Jensen, M., Hill, R.P., Huser, A.K., Kirsch, D.G., MacVittie, T.J., Mason, K.A., Medhora, M.M., Moulder, J.E., Okunieff, P., Otterson, M.F., Robbins, M.E., Smathers, J.B. & McBride, W.H. 2010, "Animal models for medical countermeasures to radiation exposure", *Radiation research*, vol. 173, no. 4, pp. 557–578.
- Willmann, S., Edginton, A.N., Kleine-Besten, M., Jantratid, E., Thelen, K. & Dressman, J.B. 2009, "Whole-body physiologically based pharmacokinetic population modelling of oral drug administration: inter-individual variability of cimetidine absorption", *Journal of pharmacy and pharmacology*, vol. 61, no. 7, pp. 891–899.
- Wilson, T.J.G., Ponder, B.A.J. & Wright, N.A. 1985, "Use of a mouse chimaeric model to study cell migration patterns in the small intestinal epithelium", *Cell proliferation*, vol. 18, no. 3, pp. 333–343.
- Wirtz, P., Loermans, H. & Rutten, E. 1982, "Effects of irradiation on regeneration in dystrophic mouse leg muscles" *British journal of experimental pathology*, vol. 63, no. 6, pp. 671 – 679.
- Withers, H.R. & Elkind, M.M. 1970, "Microcolony survival assay for cells of mouse intestinal mucosa exposed to radiation", *International journal of radiation biology*, vol. 17, no. 3, pp. 261–267.
- Withers, H.R., Brennan, J.T. & Elkind, M.M. 1970, "The response of stem cells of intestinal mucosa to irradiation with 14 MeV neutrons", *The British journal of radiology*, vol. 43, no. 515, pp.796-801.
- Withers, H.R., Chu, A.M., Reid, B.O. & Hussey, D.H. 1975, "Response of mouse jejunum to multifraction radiation", International journal of radiation oncology, biology, physics, vol. 1, no. 1–2, pp.41–52.
- Witkowski, J.A. & Brighton, W.D. 1971, "Stages of spreading of human diploid cells on glass surfaces", *Experimental cell research*, vol. 68, no. 2, pp. 372–380.

- Woitiski, C.B., Carvalho, R.A., Ribeiro, A.J., Neufeld, R.J. & Veiga, F. 2008, "Strategies toward the improved oral delivery of insulin nanoparticles via gastrointestinal uptake and translocation", *BioDrugs*, vol. 22, no. 4, pp. 223–237.
- Worthington, B.S. & Syrotuck, J. 1976, "Intestinal permeability to large particles in normal and protein deficient adult rats", *Journal of nutrition*, vol. 106, no. 1, pp. 20–32.
- Wouters, C.H. 1987, "Techniques for combining light microscopy and scanning electron microscopy: a survey of the literature", *Journal of microscopy*, vol. 147, no. 1, pp. 5–14.
- Wright, N.A., Carter, J. & Irwin, M. 1989, "The measurement of villus cell population size in the mouse small intestine in normal and abnormal states: a comparison of absolute measurements with morphometric estimators in sectioned immersion-fixed material", *Cell proliferation*, vol. 22, no. 6, pp. 425-450.
- Wu, T., Zheng, W.L., Zhang, S.Z., Sun, J.H. & Yuan, H. 2011, "Bimodal visualization of colorectal uptake of nanoparticles in dimethylhydrazine-treated mice", *World journal of gastroenterology*, vol. 17, no. 31, pp. 3614-3622.
- Wyatt, M.G., Hume, S.P., Carr, K.E. & Marigold, J.C.L. 1987, "A preliminary study of the role of gastrointestinal endocrine cells in the maintenance of villous structure following X-irradiation", *Scanning microscopy*, vol. 1, no. 1, pp. 291–300.
- Wynder, E.L. & Shigematsu, T. 1967, "Environmental factors of cancer of the colon and the rectum" *Cancer*, vol. 20, no. 9, pp., 1520–1561.
- Xiao, Y., Li, P., Cheng, Y., Zhang, X., Sheng, J., Wang, D., Li, J., Zhang, Q., Zhong, C., Cao, R. & Wang, F. 2014, "Enhancing the intestinal absorption of low molecular weight chondroitin sulfate by conjugation with α-linolenic acid and the transport mechanism of the conjugates", *International journal of pharmaceutics*, vol. 465, no. 1–2, pp. 143–158.
- Xue, B., Hausmann, M., Müller, M.H., Pesch, T., Karpitschka, M., Kasparek, M.S., Hu, W.C., Sibaev, A., Rogler, G. & Kreis, M.E. 2009, "Afferent nerve sensitivity is decreased by an iNOS-dependent mechanism during indomethacin-induced inflammation in the murine jejunum in vitro", *Neurogastroenterology and motility*, vol. 21, no. 3, pp. 322–334.
- Yan, L., Cai, C., Li, J., Xu, S., Chang, Q., Li, Y. & Wu, B. 2009, "Present status and perspectives of stem cell-based therapies for gastrointestinal diseases", *Stem cell reviews and reports*, vol. 5, no. 3, pp. 278–282.
- Yang, S., Zhu, J., Lu, Y., Liang, B. & Yang, C. 1999, "Body distribution of camptothecin solid lipid nanoparticles after oral administration", *Pharmaceutical research*, vol. 16, no. 5, pp. 751–757.
- Yuan, H., Lu, L.J, Du, Y.Z. & Hu, F.Q. 2011, "Stearic acid-g-chitosan polymeric micelle for oral drug delivery: in vitro transport and in vivo absorption", *Molecular pharmaceutics*, vol. 8, no. 1, pp. 225–238.

- Zhang, S.T., Wong, W.M., Hu, W.H.C., Trendell-Smith, N.J. & Wong, B.C.Y. 2003, "Esophageal injury as a result of ingestion of iron tablets [4]", *Journal of gastroenterology and hepatology (Australia)*, vol. 18, no. 4, pp. 466– 467.
- Zhang, W., Thornton Jr., W.H. & MacDonald, R.S. 1998, "Insulin-like growth factor-I and II receptor expression in rat colon mucosa are affected by dietary lipid intake", *Journal of nutrition*, vol. 128, no. 2, pp. 158–165.
- Zook, B.C., Bradley, E.W., Casarett, G.W. & Rogers, C.C. 1983, "Pathologic effects of fractionated fast neutrons or photons on the pancreas, pylorus and duodenum of dogs", *International journal of radiation oncology, biology, physics,* vol. 9, no. 10, pp. 1493–1504.
- Zoubi, S.A., Mayhew, T.M. & Sparrow, R.A. 1995a, "The small intestine in experimental diabetes: cellular adaptation in crypts and villi at different longitudinal sites", *Virchows archiv*, vol. 426, no. 5, pp. 501–507.
- Zoubi, S.A., Williams, M.D., Mayhew, T.M. & Sparrow, R.A. 1995b, "Number and ultrastructure of epithelial cells in crypts and villi along the streptozotocin-diabetic small intestine: a quantitative study on the effects of insulin and aldose reductase inhibition", *Virchows archiv*, vol. 427, no. 2, pp. 187–193.
- Zucco, F., Batto, A.F., Bises, G., Chambaz, J., Chiusolo, A., Consalvo, R., Cross, H., Dal Negro, G., De Angelis, I., Fabre, G., Guillou, F., Hoffman, S., Laplanche, L., Morel, E., Pinçon-Raymond, M., Prieto, P., Turco, L., Ranaldi, G., Rousset, M., Sambuy, Y., Scarino, M.L., Torreilles, F. & Stammati, A. 2005, "An inter-laboratory study to evaluate the effects of medium composition on the differentiation and barrier function of Caco-2 cell lines", *ATLA Alternatives to laboratory animals*, vol. 33, no. 6, pp. 603–618.

Probing tissue surfaces

Katharine Elizabeth Carr BSc PhD FRSB FAS FHEA

Submitted in fulfilment of the requirements for the degree of DSc in Medicine

School of Medicine, Dentistry and Nursing College of Medicine, Veterinary and Life Sciences University of Glasgow

Volume 2 of 3

© K.E.Carr, October, 2018

Accompanying material

Published work parts of Volume 2 are not included for Copyright reasons.

Code for publications divider pages

The references in the divider pages have not been set out in full Harvard format, apart from being in the order 'author(s), date, reference': the abbreviated lay-out was used for space reasons to allow tabulation. The abbreviations list given below on this page applies to the Tables in this Volume.

Abbreviation	In full
Ch	Chapter
Scn	Section
No	Number of publication in sequence in Section, as presented in the text
a/b	Designation of publication when more than one/year: in some cases there are additional publications in the same year, not included in the thesis, but cited in the text and given in the full list of References.
PS	Number of publication in Pre-submission list, also in chronological order. In this earlier list, the term 'Section' was used for what in the thesis has been called 'Chapter'.

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
1	1	1	Carr KE, Arbuthnott JP, Toner PG, Gemmell CG	1967		The fine structure of rat liver cells in suspension	Z Zellforsch. 79: 265-71.	1
		2	Carr KE	1967		Fine structure of crystalline inclusions in the globule leucocyte of the mouse intestine	J Anat. 101: 793- 803.	2
		3	Carr KE, Whur P	1968		Ultrastructure of globule leucocyte inclusions in the rat and mouse	Z Zellforsch. 86: 153-62.	3
		4	Moore MR, Meredith PA, Goldberg A, Carr KE, Toner PG, Lawrie TD	1975		Cardiac effects of lead in drinking water of rats	Clin Sci Mol Med. 49: 337-41.	13
1	2	1	Ettarh RR, Carr KE	1996	a	Morphometric analysis of the small intestinal epithelium in the indomethacin-treated mouse	J Anat. 189: 51-6.	75
		2	Abbas B, Hayes TL, Wilson DJ, Carr KE	1989		Internal structure of the intestinal villus: morphological and morphometric observations at different levels of the mouse villus	J Anat. 162: 263- 73.	52
		3	Sagher FA, Dodge JA, Johnston CF, Shaw C, Buchanan KD, Carr KE	1991		Rat small intestinal morphology and tissue regulatory peptides: effects of high dietary fat	Br J Nutr. 65: 21-8.	59
		4	McCullough JS, Ratcliffe B, Mandir N, Carr KE, Goodlad RA	1998		Dietary fibre and intestinal microflora: effects on intestinal morphometry and crypt branching	Gut. 42: 799-806.	88

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
2	1	1	Carr KE, Toner PG	1968		Scanning electron microscopy of rat intestinal villi	Lancet. 2: 570-571.	4
		2	Ferguson A, Maxwell JD, Carr KE	1969		Progressive changes in the small-intestinal villous pattern with increasing length of gestation	J Path. 99: 87-91.	5
		3	Carr KE, Joffe SN, Toner PG, Watt C	1979	a	A preliminary study of scanning electron microscopic changes of the duodenum during healing of duodenal ulcers	Scand J Gastroenterol Suppl. 54: 78-83.	16
		4	Carr KE	1970	с	Scanning electron microscope studies of human skin	Br J Plast Surg. 23: 66-72.	6
		5	Carr KE, Carr I	1970		How cells settle on glass: A study by light and scanning electron microscopy of some properties of normal and stimulated macrophages	Z. Zellforsch. 105: 234-241.	7
		6	Cook GT, Carr KE, Duncan HJ	1979		The influence of morphological differences in bracken pinnules on the foliar uptake of aminotriazole	Ann Appl Biol. 93: 311-317.	17
		7	Al-Jaff DMA, Cook, GT, Carr KE, Duncan HJ	1982		Further studies on bracken morphology in relation to herbicide uptake	In 'The Plant Cuticle' Eds Cutler, DE, Alvin KL, Price CE Academic Press, London. 293- 301.	26
2	2	1	Carr KE	1971		Applications of scanning electron microscopy in biology	Int Rev Cytol. 30: 183-255.	8
		2	Toner PG, Carr KE, Wyburn GM	1971		The Digestive System - an Ultrastructural Atlas and Review Excerpt	Butterworths.	9
		3	Carr KE, Shaw Dunn J, Toner PG	1974		Scanning electron microscopy of the alimentary tract	Scot Med J. 19: 211-220.	11

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
2	3	1	Carr KE, Hamlet R, Nias AH, Watt C	1981	a	Multinucleate giant enterocytes in small intestinal villi after irradiation.	J Microsc. 123: 169-76.	20
		2	Carr KE, Wong AL, Young DG, Toner PG, Watt C	1981	b	Scanning electron microscopy and resin histology of large bowel biopsies	Scot Med J. 26: 103-14.	21
		3	Saleh KM, Toner PG, Carr KE, Hughes HE	1982		An improved method for sequential light and scanning electron microscopy of the same cell using localising microcoverslips	J Clin Path. 35: 576-80.	27
		4	Carr KE, Kamel HM, Toner PG, McGadey J, Wong A	1984	a	Correlative scanning electron microscopy in gastrointestinal pathology	Scan Electron Microsc. 2: 761-72.	35

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
2	4	1	Carr KE, McGadey J	1974		Staining of biological material for the scanning electron microscope	J Microsc.100: 323- 30.	12
		2	Hodges GM, Carr KE, Hume SP, Marigold JC, Southgate J, Marshall JF	1985		Changes in surface structure and concanavalin A-binding capacity of urothelium in the mouse bladder after whole-body neutron irradiation	Scan Electron Microsc. 4: 1603- 14.	39
		3	Carr KE, Hayes TL, McKoon M, Sprague M, Bastacky SJ	1983	a	Low temperature scanning electron microscope studies of mouse small intestine	J Microsc.132: 209- 17.	30
		4	Carr KE, Hayes TL, Hume SP, Kamel HMH	1983	b	Qualitative and quantitative SEM of gastrointestinal tract	Proc 41st Ann Mtg EMSA. 496-499.	31
		5	Carr KE, Hayes TL, McKoon M, Bastacky SJ, Kamel HM	1984	b	Etched surfaces of plastic embedded and frozen hydrated gastrointestinal tissue	J Submicrosc Cytol. 16: 219-26.	36
		6	Carr KE, Hayes TL, Watt A, Bastacky SJ, Klein S, Fife MG	1986	a	Specimen handling and data interpretation in X- ray microanalysis of frozen hydrated etched gastrointestinal tract	J Electron Micr Techn. 4: 371-379.	42
		7	Carr KE, Hayes TL, Watt A, Klein S, McKoon M, Bastacky SJ, Ellis S	1987	a	Biological variation in cryo-microanalytical data from mouse small intestinal villi	J Electron Micr Techn. 5: 65-74.	44
		8	Dickson GR, McKenna S, McColl K, Carr KE	1989		Examination of mucosal surfaces by scanning electron microscopy before and after removal of debris	J Microsc. 153: 75- 9.	53
		9	Carr KE, Ellis S, Hayes TL Ainsworth EJ	1989		Multiview technique for SEM study of intestinal villi	Scanning 11: 109- 110.	54
		10	Abbas B, Boyle FC, Wilson DJ, Nelson AC, Carr KE.	1990	a	Radiation induced changes in the blood capillaries of rat duodenal villi: a corrosion cast, light and transmission electron microscopical study	J Submicrosc Cytol Pathol. 22: 63-70.	55

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
2	5	1	Nunn S, Gilmore RS, Dodge JA, Carr KE	1990		Exudate variation in the rabbit gastrointestinal tract: a scanning electron microscope study	J Anat. 170: 87-98.	56
		2	Laferla GA, Shashaa S, Crean GC, Gilmore RS, Carr KE	1988		Topographical variation in the mucosal surface of oesophageal biopsies	J Submicrosc Cytol Pathol. 20: 731-9.	49
		3	Shasha'a S, Dickson GR, Gilmore RS, Crean GC, Butt MM, Carr KE	1993		Rabbit and human non- keratinising stratified squamous oesophageal epithelium displays similar microridge structure by scanning electron microscopy	Scanning Microsc. 7: 953-959.	66
		4	Johnston BT, Nunn S, Sloan JM, Collins JS, McFarland RJ, Parkin S, Carr KE, Collins BJ	1996		The application of microridge analysis in the diagnosis of gastro- oesophageal reflux disease	Scand J Gastroenterol. 31: 97-102.	78

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
2	5	5	Ferguson A, Carr KE, MacDonald TT, Watt C	1978		Hypersensitivity reactions in the small intestine. IV. Influence of allograft rejection on small intestinal mucosal architecture: a scanning and transmission electron microscope study	Digestion.18: 56- 63.	15
		6	Ettarh, RR, Carr KE	1993		Structural and morphometric analysis of murine small intestine after indomethacin administration	Scand J Gastroent 28: 795-802.	65
		7	Ettarh RR, Carr KE	1997		A morphological study of the enteric mucosal epithelium in the streptozotocin-diabetic mouse	Life Sci. 61: 1851- 8.	86
		8	McClean P, Dodge JA, Nunn S, Carr KE, Sloan JM	1996		Surface features of small- intestinal mucosa in childhood diarrheal disorders	J Pediatr Gastroenterol Nutr. 23: 538-46.	77
		9	Galloway DJ, Jarrett, F, Boyle, P, Indran, M, Carr, K, Owen, RW, George, WD	1987		Morphological and cell kinetic effects of dietary manipulation during colorectal carcinogenesis	Gut. 28: 754-763.	45
		10	Ettarh RR, Carr KE.	1996	b	Ultrastructural observations on the peritoneum in the mouse	J Anat. 188: 211-5.	76
		11	Simpson CJ, Toner PG, Carr KE, Anderson JD, Carter DC	1983		Effect of bile salt perfusion and intraduct pressure on ionic flux and mucosal ultrastructure in the pancreatic duct of the cat	Virchows Arch (Cell Pathol) 42: 327-342.	32

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
2	6	1	Carr KE	1980		Scanning electron microscopy of macrophages	In 'The reticuloendothelial System' Vol I. Eds I Carr, WT Daems. 259-295.	19
		2	Carr KE, Toner PG, Saleh KM	1982	a	Scanning electron microscopy	Histopathology. 6: 3-24.	28
		3	Carr KE, Toner PG	1982		Cell Structure An Introduction to Biomedical Electron Microscopy. 3rd edition. Excerpt	Longman Group.	29
		4	Carr KE, Toner PG	1984		Morphology of the intestinal mucosa	Handbook Exptl Pharmacol. 70/I: 1- 50.	37

Probing tissue surfaces

Katharine Elizabeth Carr BSc PhD FRSB FAS FHEA

Submitted in fulfilment of the requirements for the degree of DSc in Medicine

School of Medicine, Dentistry and Nursing College of Medicine, Veterinary and Life Sciences University of Glasgow

Volume 3 of 3

© K.E.Carr, October, 2018

Accompanying material

Published work parts of Volume 3 are not included for Copyright reasons.

Code for publications divider pages

The references in the divider pages have not been set out in full Harvard format, apart from being in the order 'author(s), date, reference': the abbreviated lay-out was used for space reasons to allow tabulation. The abbreviations list given below on this page applies to the Tables in this Volume.

Abbreviation	In full
Ch	Chapter
Scn	Section
No	Number of publication in sequence in Section, as presented in the text
a/b	Designation of publication when more than one/year: in some cases there are additional publications in the same year, not included in the thesis, but cited in the text and given in the full list of References.
PS	Number of publication in Pre-submission list, also in chronological order. In this earlier list, the term 'Section' was used for what in the thesis has been called 'Chapter'.

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
3	1	1	Carr KE, Toner PG	1972		Surface studies of acute radiation injury in the mouse intestine	Virchows Arch B Cell Pathol. 11: 201-10.	10
		2	Hamlet R, Carr KE, Toner PG, Nias AH	1976		Scanning electron microscopy of mouse intestinal mucosa after cobalt 60 and D-T neutron irradiation	Br J Radiol. 49: 624-9.	14
		3	Hamlet R, Carr KE, Nias AH, Watt C	1981		Surface damage in the small intestine of the mouse after X - or neutron irradiation	Scan Electron Microsc. 4: 73-8.	22
		4	Carr KE, Hamlet R, Nias AH, Watt C	1979	b	Lack of correlation between villus and crypt damage in irradiated mouse intestine	Br J Radiol.52: 485-93.	18
3	2	1	Carr KE	1981		Scanning electron microscopy of tissue response to irradiation	Scan Electron Microsc. 4: 35-46.	23
		2	Carr KE, Seed TM	1983		Ultrastructural effects of radiation on tissues and cells Excerpt	Soft-cover book, 17 papers, reprinted from Scan Electron Microsc 1981, 1982.	33
3	3	1	Robertson AG, Wilson P, Wilson DJ, Carr KE, Hunter I	1987		Microplication patterns on human buccal epithelia following radiotherapy: a scanning electron microscopic analysis	J Submicrosc Cytol. 19: 515-21.	46
		2	Wang Q, Dickson GR, Abram WP, Carr KE	1994		Electron irradiation slows down wound repair in rat skin: a morphological investigation	Br J Dermatol. 130: 551-60.	68
		3	Wang Q, Dickson GR, Carr KE	1996		The effect of graft-bed irradiation on the healing of rat skin grafts	J Invest Dermatol. 106: 1053-7.	79
3	4	1	Indran M, Carr KE, Boyle FC	1988		Radiation-induced changes in mouse duodenal papilla	Br J Radiol. 61: 1039-42.	50
		2	Carr KE, Hayes TL, Abbas B, Ainsworth EJ	1990		Collared crypts in irradiated small intestine	J Submicrosc Cytol Pathol. 22:265-71.	57

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
3	5	1	Carr KE, Hamlet R, Nias AH, Watt C	1983	с	Damage to the surface of the small intestinal villus: an objective scale of assessment of the effects of single and fractionated radiation doses	Br J Radiol.56: 467-75.	34
		2	Carr KE, Ellis S, Michalowski A	1986	b	Surface studies of duodenal lesions induced by thoracic irradiation	Scan Electron Microsc. 1: 209- 19.	43
	6	1	Carr KE, Hamlet R, Watt C	1981	с	Scanning electron microscopy, autolysis, and irradiation as techniques for studying small intestinal morphology	J Microsc.123: 161-8.	24
		2	Carr KE, Toner PG, McLay AL, Hamlet R	1981	d	The ultrastructure of some gastrointestinal lesions in experimental animals and man	Scand J Gastroenterol Suppl. 70: 107-28.	25
		3	Carr KE, Hamlet R, Nias AH, Boyle FC, Fife MG	1985		Stromal damage in the mouse small intestine after Co60 gamma or D-T neutron irradiation	Scan Electron Microsc. 4: 1615- 21.	41
		4	Carr KE, Hayes TL, Indran M, Bastacky SJ, McAlinden G, Ainsworth EJ, Ellis S	1987	b	Morphological criteria for comparing effects of X- rays and neon ions on mouse small intestine.	Scanning Microsc. 1: 799-809.	48
		5	Abbas, B, Hume SP, McCullough JS, Wilson DJ, Stewart PC, Carr KE	1990	b	Early morphological changes in blood capillaries of mouse duodenal villi induced by X-irradiation	J Submicrosc Cytol Pathol 22: 609- 614.	58
		6	Carr KE, Hamlet R, Nias AH, Watt C	1984	с	Morphological differences in the response of mouse small intestine to radiobiologically equivalent doses of X and neutron irradiation	Scan Electron Microsc. 1: 445- 54.	38
		7	Wyatt MG, Hume SP, Carr KE, Marigold JC	1987		A preliminary study of the role of gastrointestinal endocrine cells in the maintenance of villous structure following X- irradiation	Scanning Microsc. 1: 291-300.	47

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
3	7	1	Kamel HM, Hume SP, Carr KE, Marigold JC, Michalowski A	1988		Development of villous damage in mouse small intestine after local hyperthermia or irradiation	J Submicrosc Cytol Pathol. 20: 185-93.	51
		2	Indran M, Boyle FC, Carr KE	1985		Simulation with pharmacological agents of radiation damage to small intestinal villi	Scan Electron Microsc. 3: 1165- 75.	40
		3	Indran M, Carr KE, Gilmore RS, Boyle FC	1991		Mucosal changes in mouse duodenum after gamma- irradiation or reserpine treatment	J Submicrosc Cytol Pathol. 23: 267-78.	60
		4	Carr KE, Bullock C, Ryan SS, McAlinden MG, Boyle FC	1991	a	Radioprotectant effects of atropine on small intestinal villous shape	J Submicrosc Cytol Pathol. 23: 569-77.	61
3	8	1	Carr KE, McCullough JS, Nunn S, Hume SP, Nelson AC	1991	с	Neutron and X-ray effects on small intestine summarized by using a mathematical model or paradigm	Proc Roy Soc London B. 243: 187-94.	62
		2	Carr KE, Nelson AC, Hume SP, McCullough JS	1992	a	Characterization through a data display of the different cellular responses in X-irradiated small intestine	J Radiat Res. 33: 163-77.	63
		3	Carr KE, Hume SP, Nelson AC, O'Shea O, Hazzard RA, McCullough JS	1996	a	Morphological profiles of neutron and X-irradiated small intestine	J Radiat Res. 37: 38-48.	80
		4	Carr KE, McCullough JS, Nelson AC, Hume SP, Nunn S, Kamel HM	1992	b	Relationship between villous shape and mural structure in neutron irradiated small intestine	Scanning Microsc. 6: 561-72.	64
		5	Carr KE, McCullough JS, Gilmore R S, Abbas, B, Hume SP, Nelson AC, Hayes TL, Ainsworth EJ	1993		Effects of neon ions, neutrons and X rays on small intestine	Biological effects and physics of solar and galactic cosmic radiation Part A Eds Swenberg CE et al Plenum Press, New York. 203-216.	67
		6	Carr KE, McCullough JS, Brennan P, Hayes TL, Ainsworth EJ, Nelson AC	1994		Heavy ion induced changes in small intestinal parameters	Adv Space Res. 14: 521-30.	69

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
3	9	1	Brennan PC, Carr KE, Seed T, McCullough JS	1998		Acute and protracted radiation effects on small intestinal morphological parameters	Int J Radiat Biol. 73: 691-8.	89
		2	Ettarh RR, Hodges GM, Carr KE	2000		Radiation effects in the small bowel of the diabetic mouse	Int J Radiat Biol. 76: 241-8.	91
		3	Brennan PC, McCullough JS, Carr KE	1999		Variations in cell and structure populations along the length of murine small intestine	Cells Tissues Organs. 164: 221- 6.	90
	10	1	Carr KE, Hume SP, Ettarh R, Carr EA, McCullough JS	1995		Radiation-induced changes to epithelial and non- epithelial tissue	In Radiation and the Gastrointestinal Tract Eds A. Dubois. D.R. King; D.R. Livengood. Taylor and Francis: 113-128.	70
		2	Carr KE	2001		Effects of radiation damage on intestinal morphology	Int Rev Cytol. 208: 1-119.	92

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
4	1	1	Carr KE, McCullough SJ, Hazzard RA, Brennan P, McGuiness C, Seed, T and Hodges GM	1997		Structural and functional profiles of mammalian small intestine	Recent Advances in Microscopy of Cells, Tissues and Organs. Ed Motta PM. 381-386.	87
		2	Carr KE, Dayan AD, Florence AT, Goodlad RA, Hodges GM, Rowlands BJ	1996	b	Symposium on particle translocation across epithelial membranes	J Anat. 189: 475.	81
		3	Hodges GM, Carr EA, Hazzard RA, O'Reilly C, Carr KE	1995	a	A commentary on morphological and quantitative aspects of microparticle translocation across the gastrointestinal mucosa	J Drug Target. 3: 57-60.	71

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
4	2	1	Hodges GM, Carr EA, Hazzard RA, Carr KE	1995	b	Uptake and translocation of microparticles in small intestine. Morphology and quantification of particle distribution.	Dig Dis Sci. 40: 967-75.	74
4	3	1	McCullough JS, Hodges GM, Dickson GR, Yarwood A, Carr KE	1995		A morphological and microanalytical investigation into the uptake of particulate iron across the gastrointestinal tract of rats	J Submicrosc Cytol Pathol. 27: 119-24.	72
		2	Cartwright- Shamoon J, Dickson GR, Dodge J, Carr KE	1995		Morphological aspects of particle translocation in vivo following ingestion of the yeast Saccharomyces boulardii	J Drug Target. 3: 61-3.	73
		3	Carr KE, Youngman MJ, Barr M, Colhoun L, Harrison JD	2001		Visualisation of fused aluminosilicate microparticles taken up by intestinal tissue	J Anat 199: 225–6.	93

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
4	4	1	Hazzard RA, Hodges GM, Scott JD, McGuinness CB, Carr KE	1996		Early intestinal microparticle uptake in the rat	J Anat. 189: 265- 71.	82
		2	McMinn LH, Hodges GM, Carr KE	1996		Gastrointestinal uptake and translocation of microparticles in the streptozotocin-diabetic rat	J Anat. 189: 553-9.	85
		3	Carr KE, Hazzard RA, Reid S, Hodges GM	1996	с	The effect of size on uptake of orally administered latex microparticles in the small intestine and transport to mesenteric lymph nodes	Pharm Res. 13: 1205-9.	83
		4	Cartwright- Shamoon J, Dickson GR, Dodge J, Carr KE	1996		Uptake of yeast (Saccharomyces boulardii) in normal and rotavirus treated intestine	Gut. 39: 204-9.	84
		5	Smyth SH, Doyle- McCullough M, Cox OT, Carr KE	2005		Effect of reproductive status on uptake of latex microparticles in rat small intestine	Life Sci. 77: 3287- 305.	94
		6	Doyle- McCullough M, Smyth SH, Moyes SM, Carr KE	2007		Factors influencing intestinal microparticle uptake in vivo	Int J Pharm. 335: 79-89.	96
		7	Smyth SH, Feldhaus S, Schumacher U, Carr KE	2008		Uptake of inert microparticles in normal and immune deficient mice	Int J Pharm. 346: 109-18.	97

Ch	Scn	No	Authors	Date	a/b	Title	Reference	PS
4	5	1	Moyes SM, Smyth SH, Shipman A, Long S, Morris JF, Carr KE	2007		Parameters influencing intestinal epithelial permeability and microparticle uptake in vitro	Int J Pharm 337: 133-141.	95
4	6	1	Moyes SM, Morris JF, Carr KE	2010	a	Culture conditions and treatments affect Caco-2 characteristics and particle uptake	Int J Pharm. 387: 7-18.	102
		2	Moyes SM, Morris JF, Carr KE	2011		Roles of pre-treatment time and junctional proteins in Caco-2 cell microparticle uptake	Int J Pharm. 407: 21-30.	99
		3	Moyes SM, Killick EM, Morris JF, Kadhim MA, Hill MA, Carr KE	2008		Changes produced by external radiation in parameters influencing intestinal permeability and microparticle uptake in vitro	Int J Radiat Biol. 84: 467-86.	98
		4	Moyes SM, Morris JF, Carr KE	2010	b	Macrophages increase microparticle uptake by enterocyte-like Caco-2 cell monolayers	J Anat. 217: 740- 54.	103
		5	Carr KE	2018			Personal communication.	101
4	7	1	Carr KE, Smyth SH, McCullough MT, Morris JF, Moyes SM	2012		Morphological aspects of interactions between microparticles and mammalian cells: intestinal uptake and onward movement	Prog Histochem Cytochem. 46: 185-252.	100